



EMMA-RIIKKA MYLLYMÄKI

Essays on Internal Control and External Auditing in the Context of Financial Reporting Quality



ACTA WASAENSIA 322

BUSINESS ADMINISTRATION 132 ACCOUNTING AND FINANCE

Reviewers

Professor Marleen Willekens KU Leuven Faculty of Economics and Business Naamsestraat 69 BE-3000 LEUVEN BELGIUM

Professor Mervi Niskanen University of Eastern Finland Department of Business P.O. Box 1627 FI-70211 Kuopio FINLAND

Julkaisija	Julkaisupäi	vämäärä	
Vaasan yliopisto	Huhtikuu 20	15	
Tekijä(t)	Julkaisun tyyppi		
Emma-Riikka Myllymäki	Artikkelikokoelma		
	Julkaisusarjan nimi, osan numero		
	Acta Wasaer	isia, 322	
Yhteystiedot	ISBN		
Vaasan yliopisto	978-952-476-604-3 (print)		
Kauppatieteellinen tiedekunta	978-952-476-605-0 (online)		
Laskentatoimi ja rahoitus	ISSN		
PL 700	0355–2667 (Acta Wasaensia 322, print)		
65101 Vaasa	2323–9123 (Acta Wasaensia 322, online)		
00101 / 4404	1235–7871 (Acta Wasaensia.		
	Business Administration 132, print)		
2323–9735 (Acta Was		ta Wasaensia.	
	Business Administration 132, online)		
	Sivumäärä	Kieli	
	172	englanti	

Julkaisun nimike

Esseitä sisäisestä valvonnasta ja tilintarkastuksesta taloudellisen raportoinnin laadun kontekstissa

Tiivistelmä

Tämä väitöskirja käsittelee yrityksen sisäisen valvonnan ja tilintarkastuksen roolia taloudellisen raportoinnin laadun kontekstissa. Yritysten julkaiseman taloudellisen informaation oikeellisuus ja luotettavuus on tärkeää sidosryhmien päätöksenteolle ja siten talouden toimivuudelle. Väitöskirjan kaksi ensimmäistä esseetä keskittyvät sisäisen valvonnan tehokkuuden aihepiiriin, kun taas kahden jälkimmäisen esseen tarkastelukohteena on tilintarkastuksen laatu. Tutkimusaineisto koostuu kolmessa ensimmäisessä esseessä yhdysvaltalaisista listatuista yhtiöistä ja neljännessä esseessä suomalaisista listatuista yhtiöistä.

Ensimmäisen esseen tutkimustuloksena havaitaan, että heikko taloudellisen raportoinnin laatu on todennäköisempää sisäisen valvonnan heikkouksista raportoivissa yrityksissä vielä kahtena vuotena viimeisimmän heikkouksista tiedottavan raportin jälkeen. Toinen essee jatkaa sisäisen valvonnan aihepiirissä ja sen tulokset osoittavat, että operatiivinen tuloksenjärjestely on yleisempää yrityksissä, joilla on heikkouksia sisäisessä valvonnassa. Kolmannen esseen tutkimustulokset antavat viitteitä heikommasta tilintarkastuksen laadusta silloin, kun tilintarkastusyhteisö tarjoaa myös verotukseen liittyviä palveluita samalle asiakkaalle. Neljäs essee tarkastelee tilintarkastajan erikoistumista listattujen yhtiöiden tarkastamiseen ja tuloksena havaitaan, että tällaisella erikoistumisella on positiivinen vaikutus listatun asiakasyrityksen taloudellisen raportoinnin laatuun.

Kokonaisuutena tämän väitöskirjan tutkimustulokset tuovat uutta tietoa sekä sisäisen valvonnan heikkouksista juontuvien ongelmien laajuudesta että tilintarkastuksen laadusta riippumattomuuden vaarantumisen ja tilintarkastajan erikoistumisen näkökulmista.

Asiasanat

taloudellisen raportoinnin laatu, sisäinen valvonta, tilintarkastusyhtiön tarjoamat oheispalvelut, tilintarkastajan erikoistuminen

Publisher	Date of publication		
Vaasan yliopisto	April 2015		
Author(s)	Type of publication		
Emma-Riikka Myllymäki	Selection of articles		
	Name and number of series		
	Acta Wasaensia, 322		
Contact information	ISBN		
University of Vaasa	978-952-476-604-3 (print)		
Faculty of Business Studies	978-952-476-605-0 (online)		
Accounting and Finance	ISSN		
P.O. Box 700	0355–2667 (Acta Wasaensia 322, print)		
FI-65101 Vaasa, Finland	2323–9123 (Acta Wasaensia 322, online)		
<i>,</i>	1235–7871 (Acta Wasaensia.		
	Business Administration 132, print)		
	2323–9735 (Acta Wasaensia.		
	Business Administration 132, online)		
	Number of pages Language		
	172 English		

Title of publication

Essays on Internal Control and External Auditing in the Context of Financial Reporting Quality

Abstract

This thesis examines the role of internal control and external auditing in the context of financial reporting quality. The truthfulness and reliability of companies' financial reporting is essential for the decision-making of stakeholders and hence for economic functionality. The first two essays of the dissertation focus on internal control over financial reporting, while the last two essays study audit quality. Three essay uses data on US listed companies, and the fourth essay uses data on Finnish listed companies.

The findings of the first essay indicate that companies disclosing internal control weaknesses have greater likelihood of low financial reporting quality still two years after the last disclosure. The second essay continues in the field of internal controls and finds that companies with internal control weaknesses are manipulating real operational activities to manage earnings. The findings of the third essay imply lower audit quality, when the incumbent audit firm provides also tax services to the client. The fourth essay examines auditors' specialization in auditing publicly listed companies, and the results suggest that this type of specialization has a positive impact on the financial reporting quality of public clients.

All in all, the findings of this thesis provide new evidence both on the pervasive nature of internal control weaknesses, and on external audit quality from the perspective of auditor independence and auditor specialization.

Keywords

financial reporting quality, internal control, non-audit services, auditor specialization

ACKNOWLEDGEMENTS

I could write a book of just acknowledgements. In the following – which is merely a short list that does not cover everything – I try to express my deep appreciation to all of those who have contributed to my dissertation project.

First of all, I wish to thank my supervisor, Professor Teija Laitinen, for the guidance and encouragement she has provided over the years. Thank you for believing in my abilities to do academic research. I am very grateful to Professor Marleen Willekens from the Katholieke Universiteit Leuven and Professor Mervi Niskanen from the University of Eastern Finland for serving as the pre-examiners of this dissertation thesis.

I have had the privilege to work in such a great research environment where assistance, advice, and support are always near. I would like to thank all my coauthors for their valuable contribution in the essays of this dissertation. I am grateful to Dr. Tuukka Järvinen for co-authoring the second essay and for all the helpful discussions, advice and support over the years. I wish to thank Professor Erkki K. Laitinen for his expert advice and suggestions on my research, and for co-authoring the third essay. I would like to thank Professor H. Gin Chong for his contribution in the third essay. I am grateful to Professor Karla Johnstone for sharing her expertise and for co-authoring the fourth essay. I wish to thank Dr. Kim Ittonen for his advice during my studies, for co-authoring the fourth essay, and for introducing me to the Department of Accounting and Finance in the first place.

I am very grateful to Dr. Mikko Zerni, not only for all his excellent comments and insightful views on my research, but also for the encouragements. I wish to thank Professor Stefan Sundgren for the many valuable comments and suggestions. I express my warm thanks to Dr. Annukka Jokipii for all the conversations, generous support, and guidance over the years. I would also like to thank Professor Sami Vähämaa and Dr. Jukka Sihvonen for their valuable advice. A big thank you goes to Dr. Nina Sormunen, Elina Haapamäki, and Eija Kärkinen, with whom I began this process at about the same time, and with whom I have shared many conference trips and PhD courses, not to mention experienced the terrifying moments of first conference presentations. Overall, I wish to express my heartfelt gratitude to all my colleagues – friends. I truly value your exceptional support, kindness, good conversations and laughs, and all the special times shared.

During my studies, I was fortunate to make research visits abroad. I wish to thank the people at the Department of Accounting and Auditing at the Copenhagen Business School, especially Dr. Kim Pettersson, and the people at the Business Department at the Universitat Autònoma de Barcelona, especially Professor Diego Prior. I am also grateful to Laura Baselga Pascual from the Universidad Pablo de Olavide, and to Dr. Tobias Svanström from the Umeå School of Business and Economics for the opportunity to present my research in their seminars. Over the years, I have presented the essays of this dissertation in conferences and workshops, and I wish to thank the discussants/participants in the 37th Annual Congress of European Accounting Association, the 7th European Auditing Research Network Symposium and Ph.D. Workshop, the 6th International Workshop on Accounting and Regulation, the 6th Workshop on Auditing and Financial Accounting Research, and Doctoral Tutorial in Accounting 2014.

I thank the University of Vaasa for employing me during my PhD studies, and for providing an excellent working environment. Also, I gratefully acknowledge the financial support for my dissertation project from the Finnish Foundation for Economic Education, Jenny and Antti Wihuri Foundation, Nordea Bank Foundation, Oskar Öflund's Foundation, and Evald and Hilda Nissi Foundation.

Finally, a special thanks goes to my family and friends for supporting me, listening to my worries, encouraging me, and for helping me to try to keep things in perspective. I am grateful to all my friends for giving me the chance to escape from this project from time to time. Especially Johanna, Matti, and Kirsi deserve a big thank you for their encouragements. Above all, I owe my deepest gratitude to my parents Helena and Hannu, sister Henna, and brother Tatu for their support. You are my heroes.

Vaasa, April 2015

Emma-Riikka Myllymäki

Sisällys

AC	CKNOV	WLEDGEMENTS	VII
1	INTR	ODUCTION	1
2	FINA	NCIAL REPORTING	4
	2.1	Demand for financial reporting	4
	2.2	Financial reporting quality	5
3	INTE	RNAL CONTROL OVER FINANCIAL REPORTING	8
	3.1	Section 404 of the Sarbanes-Oxley Act	9
	3.2	Determinants and consequences of internal control weaknesses	10
	3.3	Pervasiveness of internal control weaknesses	12
4	AUD	T QUALITY	14
	4.1	Non-audit services and audit quality	15
	4.2	Individual auditor specialization and audit quality	16
5	SUM	MARY OF THE ESSAYS	18
	5.1	The persistence in the association between Section 404 material weaknesses and financial reporting quality	18
	5.2	Real earnings management before and after reporting SOX 404 material weaknesses	19
	5.3	Incumbent audit firm-provided tax services and clients with low financial reporting quality	21
	54	Audit partner public-client specialization and client abnormal	21
	2	accruals	22
RE	FERE	NCES	24

This dissertation thesis consists of an introductory chapter and the following four essays:

- Myllymäki, Emma-Riikka (2014). The persistence in the association between Section 404 material weaknesses and financial reporting quality. *Auditing: A Journal of Practice & Theory* 33:1, 93–116.¹
- Järvinen, Tuukka & Emma-Riikka Myllymäki (2014). Real earnings management before and after reporting SOX 404 material weaknesses.
- Chong, H. Gin, Erkki K. Laitinen & Emma-Riikka Myllymäki (2014). Incumbent audit firm-provided tax services and clients with low financial reporting quality. *Universitat Autònoma de Barcelona Working paper series* No. 14/4. http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2539018
- Ittonen, Kim, Karla M. Johnstone & Emma-Riikka Myllymäki (2014). Audit partner public-client specialisation and client abnormal accruals. *European Accounting Review*, Forthcoming, DOI: 10.1080/09638180.2014.906315.²

¹ Printed with kind permission of American Accounting Association.

² Printed with kind permission of Taylor & Francis.

1 INTRODUCTION

Due to the recent financial crisis and the widely publicized accounting scandals at the beginning of the century (e.g., Enron, Parmalat, WorldCom), there has been a growing demand to enhance the transparency of companies' operations and the integrity of financial reporting. Shareholders, creditors, and other company stakeholders base their decisions on publicly disclosed financial information, and hence, the truthfulness and reliability of that information is essential for economic functionality. This doctoral dissertation focuses on two corporate governance mechanisms for financial reporting quality: internal control over financial reporting and external auditing. In response to the aforementioned scandals, a highprofile change to the legislation was made in 2002 in the USA with the passing the Sarbanes-Oxley Act (SOX), which aims at improving companies' governance, internal controls, and external audit quality.³ Consequently, corporate governance codes and audit regulation have been developed worldwide by further defining and including new requirements especially for publicly listed companies, audit firms, and auditors.⁴ Thus, internal controls and external auditing have been at the center of the discussion among academics, practitioners, and regulators. As a whole, the purpose of the four essays that comprise this dissertation is to provide new evidence on the role of internal controls and external auditing in the context of financial reporting quality.

The first essay uses data on internal control reports mandated by Section 404 of the SOX, which requires the management of listed companies to annually assess and report on the effectiveness of internal control over financial reporting, and to disclose any material weaknesses. The study examines whether Section 404 material weakness (MW404) disclosures are predictive of future financial reporting quality, and concentrates on the fiscal years following the last MW404 disclosure, i.e., the post-MW404 period. The findings suggest that in the first two years after the last MW404 disclosure, internal controls are still not as effective at preventing or detecting misstatements in a timely manner as they are in companies without a history of MW404s. The findings further imply that the reason for the misstate-

³ Along with the SOX, Public Company Accounting Oversight Board (PCAOB) was established. PCAOB is responsible for regulating the auditing profession and monitoring public accounting firms and the compliance with SOX.

⁴ For example, the new Statutory Audit Directive of the EU was enacted in 2006, and amended again in 2014. The main new issues in the 2006 directive were related to strengthening the oversight of auditors and auditor independence. The new requirements in the 2014 directive further aim at improving auditor independence, audit oversight and audit report informative-ness, for example.

2 Acta Wasaensia

ment incidences in the post-MW404 period is the unacknowledged pervasiveness of control problems.

The second essay examines the association between MW404s and manipulation of real operational activities to manage earnings (e.g., inventory overproduction). The empirical findings indicate that real earnings management is greater in companies with existing material weaknesses, and in companies disclosing previous year's material weaknesses. It appears that the poor commitment by management to provide effective internal control systems and high quality financial information relates to a tendency to use real earnings management methods and also impairs management's real operational decisions. Moreover, the public disclosure of material weaknesses might induce management to strive to mitigate the expected negative reactions of stakeholders to the disclosure by engaging in real earnings management. Overall, this study provides further insights into pervasive control problems that may exist in companies with material weaknesses by documenting how operational activities fall under the sphere of influence of internal control effectiveness.

The third essay investigates audit quality in the context of tax services provided by incumbent audit firms. Although having the same audit firm to provide both auditing and non-auditing services could improve audit quality due to possible knowledge spillover, the greater economic dependence may jeopardize auditor independence leading to lower audit quality. The SOX legislation prohibits most non-audit services being provided by incumbent audit firms, but allows tax services. The empirical findings of the third essay suggest that there is a greater likelihood of low financial reporting quality remaining unacknowledged when taxrelated fees are higher, which supports the economic dependence view. However, the findings also imply that the mere act of providing both audit and tax services does not in itself have an impact on audit quality, but rather it is the magnitude of the tax-related fees that counts.

The fourth essay focuses on individual auditor specialization in auditing public clients. Especially nowadays with increasingly complex client companies and greater demand for high quality auditing by regulators and stakeholders, auditors need to possess specialized in-depth knowledge of their clients to build expertise in a domain and perform high quality audits. Using Finnish data on listed companies, the fourth essay examines the association between an audit partner's public-client specialization and the client company's abnormal accruals, which is used as a proxy for audit/financial reporting quality. The identity of the individual auditor in charge is publicly available in Finland because, unlike in the USA or the UK for example, the responsible auditors are required to personally sign the audit re-

port. The findings suggest that greater public-client specialization is associated with higher audit/financial reporting quality. Moreover, it appears that this association is attributable to partners with a moderate level of public-client specialization, while the higher number of public-clients may reflect busyness, mitigating the benefits related to the public-client specialization.

Collectively, the findings of this dissertation contribute to the literature on internal control and external auditing in the context of financial reporting quality. The inferences from the first two essays underline the pervasive nature of internal control weaknesses and the role of management in developing effective internal control systems. The last two essays focus on audit quality, and discuss how auditor judgments can be disrupted by the economic dependence of the audit firm, but benefit from individual auditor specialization.

The remainder of the introductory chapter is structured as follows. Section 2 briefly describes the theoretical background of financial reporting quality in order to illustrate the role of internal controls and external auditing. Section 3 presents the concept and relevant prior research on internal control over financial reporting, and introduces the research questions in the first two essays. Section 4 describes the perspective on audit quality used in this dissertation, and discusses the relevant background of the provision of non-audit services by incumbent audit firms, and of auditor specialization. Section 5 summarizes the four essays.

2 FINANCIAL REPORTING

2.1 Demand for financial reporting

The role of financial reporting is most commonly explained by agency theory. In an agency relationship, a principal engages an agent to work on the principal's behalf through a contract. The fundamental problem in the principal-agent relationship stems from self-interest, where both parties attempt to maximize their own utility, but their interests are not necessarily aligned. Information asymmetry between the two parties creates an opportunity for the agent to gain private benefits. Therefore, the principal faces the risk that the agent will try to maximize his/her private benefits at the expense of the principal. The agent's behavior is unobservable to the principal, who faces the risk that the agent is not doing what he/she is supposed to do (moral hazard), and/or cannot verify the skills and abilities of the agent (adverse selection). (E.g., Jensen and Meckling 1976; Eisenhardt 1989.)

The agency relationship between shareholders (principals) and management (agents) is considered to predominantly create the demand for financial reporting especially in large companies (such as public (listed) companies) where, in principle, ownership and control are separated. Financial reporting alleviates agency problems by aligning the interests of management with those of the shareholders (bonding), and by monitoring. The bonding role can be observed from incentive contracts, which are usually based on the financial statement numbers. The monitoring role suggests that financial statements are used to monitor managerial actions (performance and contract terms). In addition to the shareholder-manager relationship, financial reporting also reduces information asymmetry between blockholders and minority shareholders, or creditors and shareholders, for instance.⁵ Moreover, publicly disclosed financial statements provide information for the decision-making by a number of different stakeholders, including shareholders and creditors, but also potential shareholders, suppliers, and employees,

⁵ Small and medium-sized companies, which are usually private (i.e., unlisted) companies, are mostly run by owner-managers. Thus, the problem of information asymmetry in private companies centers on the relationship of manager-owners and creditors, for instance. Ball and Shivakumar (2005), however, argue that private companies are likely to distribute information via unofficial channels, and thus, information asymmetry is not as profound as in listed companies. They suggest that the demand and supply of financial reporting in private companies can rather be explained by tax, dividend, and compensation payment policies than information asymmetry issues.

etc. (Fama 1980; Fama and Jensen 1983; Watts and Zimmermann 1983; Bushman and Smith 2001.)

2.2 Financial reporting quality

In order for financial reporting to serve its bonding and/or monitoring purpose of reducing agency problems, the disclosed information needs to be truthful and reliable. The conceptual frameworks for financial reporting produced by both the Financial Accounting Standards Board (FASB) and International Accounting Standards Board (IASB) state that relevance and faithful representation (complete, neutral and free from error) are the important qualitative characteristics of financial information (FASB 2010; IASB 2010). Quality levels however vary across companies. Previous literature has extensively investigated the determinants and consequences of financial reporting quality (or earnings quality), using various proxies to capture different dimensions of quality (see Dechow et al. 2010 for a review).⁶

Although financial reporting quality can have many slightly different dimensions, this dissertation focuses more on the aspect of faithful representation, and not, for example, on perceived quality by investors. Bias and erroneous information can occur both intentionally and unintentionally. The financial reporting process involves decision-making, and discretion is used in accounting choices (e.g., Watts and Zimmerman 1983). Due to its decision-making authority, management has both opportunities and incentives (bonuses and reputation building, for example) to manipulate accounting numbers to reach earnings targets (e.g., Watts and Zimmerman 1983). However, not all the deterioration in financial reporting quality stems from intentional malpractice, unintentional errors may also contribute. Nonetheless, in these cases too, the accountability of management is important since it is responsible for establishing a properly functioning accounting system.

The measures of financial reporting quality used in the essays of this dissertation are introduced briefly here.

⁶ Dechow et al. (2010) discuss the different proxies for earnings quality examined in previous studies by categorizing them to earnings properties (earnings persistence, abnormal accruals, earnings smoothness, asymmetric timeliness and timely loss recognition, and target beating), investor responsiveness to earnings (the research on earnings response coefficient as a proxy for perceived earnings quality), and external indicators of earnings misstatements (for example, restatements and internal control weaknesses).

6 Acta Wasaensia

- In the USA the generally accepted accounting principles (GAAP) requires companies to restate previous financial statement(s), if it includes (either intentional or unintentional) material misstatement (e.g., DeFond and Jiambalvo 1991). Restatements explicitly indicate problems in the accounting system (violations of accounting principles), and hence, are used as a proxy for financial reporting quality (e.g., DeFond and Francis 2005).
- A vast number of previous studies have focused on the magnitude of abnormal accruals as a proxy for earnings quality. The basic idea is to distinguish abnormal accruals from the normal by modeling the expected accruals. The magnitude of abnormal accruals is considered to capture the problems in the accounting measurement system, and distortions that stem from discretion used in accrual choices or earnings management. Previous literature uses several different models to estimate abnormal accruals (e.g., Jones 1991; Dechow et al. 1995; DeFond and Park 2001; Dechow and Dichev 2002; McNichols 2002; Kothari et al. 2005; Ball and Shivakumar 2006).
- Real earnings management is not related to accounting issues per se, but refers to the manipulation of real operational activities such as inventory overproduction or reduction of discretionary expenses. These actions are departures from normal operating practices conducted to achieve financial targets, but which might have a negative effect on long-term company value (Roychow-dhury 2006). Prior research has stated that real earnings management is extensively employed, because it is not easily detected or constrained by outsiders (e.g., Graham et al. 2005; Cohen et al. 2008). The most commonly examined real earnings management methods are inventory overproduction (proxied by abnormal levels of discretionary expenses), and sales manipulation (proxied by abnormal levels of cash flows from operations).

Because of the risk of misstatements, users of financial statement information need assurance of the integrity of the accounting system and financial reporting. *Internal control over financial reporting* aims to prevent and/or detect errors or malpractice that could result in a misstatement in a financial statement (PCAOB 2007). However, if not properly established, maintained and developed, internal control may not serve its purpose. Consequently, the existence of internal control weaknesses can also be considered an indicator of low financial reporting quality, while effective internal control should manifest in high quality financial information. Internal control over financial reporting is further elaborated upon in section 3.

The task of *external auditing* is to provide reasonable assurance to the users of financial information that a client company's financial statements are fairly stated and free of material misstatement (e.g., Watts and Zimmerman 1986; Eilifsen and Messier 2000). However, the quality of an audit is not just about meeting legal and professional requirements, but it is rather a continuum that ranges between low and high quality (e.g., Francis 2004; Francis 2011). Audit quality as it relates to the provision of non-audit services by incumbent audit firm and auditor specialization is discussed in more detail in section 4.

3 INTERNAL CONTROL OVER FINANCIAL REPORTING

According to the Committee of Sponsoring Organizations of the Treadway Commission's (COSO) framework from 1992, the three objectives of internal control are 1) the effectiveness and efficiency of operations, 2) reliability of financial reporting, and 3) compliance with applicable laws and regulations. Naturally, an internal control process involves the risk that the objectives are not achieved (Kinney 2000). This dissertation concentrates on the risk of not achieving the second objective, and examines the effectiveness of internal control over financial reporting.

> "Internal control over financial reporting is a process designed... to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes" (PCAOB 2007).

Management is responsible for establishing and maintaining adequate internal controls, and ought to adopt an internal control framework to assist with both establishing internal controls and with evaluating the effectiveness of the control system (SOX 2002; SEC 2003a).⁷ Because the management is accountable for the quality of financial information, internal controls can be considered a tool for management to alleviate the risk of not achieving the objective of reliable financial reporting due to errors or malpractice by the personnel, for instance. However, management itself has the incentives and opportunities to neglect its responsibility to establish a properly functioning internal control system. In accordance with the definition of internal control over financial reporting, ineffective internal controls may not be able to prevent or detect misstatements in financial information (e.g., DeFond and Jiambalvo 1991; Eilifsen and Messier 2000; PCAOB 2007; Doyle et al. 2007a; Ashbaugh-Skaife et al. 2008).

⁷ The most widely known internal control framework is the COSO (1992) framework, which comprises five components of internal controls: control environment sets the foundation for the entire internal control system, 'the tone at the top', and involves management's philosophy, human resources, policies and practices, among others; risk assessment involves the identification and assessment of the relevant risks in achieving objectives; control activities involves policies and practices to ensure that the risks are mitigated or eliminated and objectives are achieved; information and communication relates to the identification and communication of relevant information throughout the organization; monitoring involves the follow-up of the internal controls. COSO-ERM is a refined integrated framework that focuses on enterprise risk management. In addition to the components of the traditional COSO framework, the COSO-ERM framework includes also three other components to help with risk management: objective setting, event identification, risk response.

3.1 Section 404 of the Sarbanes-Oxley Act

Section 404 of the SOX focuses on improving companies' internal controls, and requires the management of companies under the authority of the Securities and Exchange Commission (SEC) to provide annual reports containing an internal control report. The internal control report must include a statement about management's responsibility for establishing and maintaining adequate internal control over financial reporting, and the management's assessment of the effectiveness of the internal controls. Any material weaknesses must be disclosed. A material weakness is defined as a significant deficiency, or combination of significant deficiencies, that results in more than a remote likelihood that a material misstatement of the interim or annual financial statements will not be prevented or detected (PCAOB 2007). Section 404 also requires that the company's external auditor attests to and reports on the assessment made by the management. In order to be able to render an opinion on the effectiveness of the client's internal controls, the auditor must plan and perform a comprehensive evaluation of them (PCAOB 2007).⁸

Section 404 became effective for the fiscal years ending after November 15, 2004 for accelerated filers (market capitalization of at least 75 million dollars). The SOX also includes Section 302, which became effective for fiscal years ending after August 29, 2002, and requires management to assess the effectiveness of disclosure controls and procedures on a quarterly basis. The quarterly certification should indicate that the management has evaluated the effectiveness of internal controls, as well as any significant changes in internal controls. The provisions of Section 302 are, however, somewhat less stringent than those of Section 404, as Ashbaugh-Skaife et al. (2007) note "...under the provisions of Section 302, the review of internal control is subject to less scrutiny by both management and the auditor and the disclosure rules are less specific than subsequently exist under Section 404."

Although there are some prior studies suggesting that Section 404 has achieved its objective of improving the quality of companies' financial reporting (e.g., Nagy 2010), it has also attracted considerable criticism, especially due to the high costs it has brought to companies and their auditors. Audit fees have risen substantially with the adoption of Section 404, since the scope of financial statement audits has widened, and auditor responsibility, and litigation risk due to investor expecta-

⁸ External auditors follow the standards by the PCAOB in their internal control assessment, while management follows the guidance issued by the SEC (Schneider et al. 2009).

tions have increased (e.g., Raghunandan and Rama 2006; Krishnan et al. 2008; Hoag and Hollingsworth 2011). In 2007, Auditing Standard No. 2 was replaced by Auditing Standard No. 5 (AS5), which attempts to increase the efficiency of internal control evaluation. In particular, AS5 includes guidance on the "top-down risk-based" approach to internal control testing, which means that audit firms should focus on areas that include the most risk and they should scale audits based on the client company size and complexity (PCAOB 2007). Moreover, since 2007 smaller public companies have also started to report on their internal control effectiveness, but because of the concerns of high costs compared to bene-fits, external auditors' internal control reporting is not required (SEC 2010).⁹

3.2 Determinants and consequences of internal control weaknesses

The public internal control disclosures have enabled researchers to use large datasets to examine different aspects of internal control effectiveness. Companies disclosing internal control weaknesses are found to be smaller, riskier, more complex, and poorly performing in comparison to companies that have reported effective internal controls (e.g., Ge and McVay 2005; Ashbaugh-Skaife et al. 2007; Doyle et al. 2007b). That is, weaknesses appear to occur in companies that may have difficulties in investing in internal controls due to limited resources, for example. Moreover, companies with weak boards, audit committees, and financial management are more likely to have internal control weaknesses (Krishnan and Visvanathan 2007; Zhang et al. 2007; Hoitash et al. 2009; Li et al. 2010). Strong governance has the expertise and the will to invest in internal controls. The previous research also suggests that companies are more likely to remediate their internal control deficiencies if they have stronger governance and better financial performance (e.g., Goh 2009; Li et al. 2010; Johnstone et al. 2011). Bedard et al. (2012), however, point out that the likelihood of such remediation depends on the type of weakness, that is, some weaknesses are remediated more quickly.

Several studies have examined the consequences of having/disclosing weaknesses. As the aim of internal controls over financial reporting is to secure the reliability of financial information, prior literature has examined the association between the effectiveness of internal controls and financial reporting quality (Doyle et al.

⁹ The first essay of this dissertation uses data on companies with auditor internal control reports, and the second and the third essay uses data that comprises both auditors' internal control reports and management-only internal control reports.

2007a; Chan et al. 2008; Ashbaugh-Skaife et al. 2008; Bedard et al. 2012). In general, the findings of these studies suggest that internal control weaknesses cause deterioration in financial reporting quality. Doyle et al. (2007a) find that the association is primarily driven by weaknesses in less auditable entity-level controls, while the association is not especially evident with account-specific internal control weaknesses. Chan et al. (2008) provide marginally significant evidence that companies with material weaknesses have greater discretionary accruals. Ashbaugh-Skaife et al. (2008) report that accruals quality is better in companies remediating weaknesses compared to companies continuing to disclose internal control weaknesses, and Bedard et al. (2012) suggest that the significant impact of the material weakness.

According to the agency theory view on internal controls, a company's public disclosures should matter to users of financial statement information. Prior research has examined the impact of internal control disclosures on equity markets, and found that these disclosures do affect investors' risk assessment, stock returns, and companies' cost of equity (Beneish et al. 2009; Ashbaugh-Skaife et al. 2009; Lopez et al. 2009; Rezee et al. 2012). Previous studies investigating debt holders' reactions to internal control reports have documented that internal control weaknesses affect loan officers' risk assessment (Schneider and Church 2008) and cost of debt (Kim et al. 2011). In addition to the research investigating equity and debt market reactions, Su et al. (2014) found that customer demand decreases after internal control weaknesses are disclosed.

Considering the negative consequences of internal control weaknesses on financial reporting quality, and equity and debt markets, the impact of internal control effectiveness on auditors' risk assessment and audit fees has been discussed in prior studies. When designing the audit process, auditors need to carefully evaluate risks involved to that specific audit engagement in order to plan the required audit effort, and to reduce overall risk to an acceptable level (e.g., O'Keefe et al. 1994; Johnstone and Bedard 2001).¹⁰ Thus, high levels of perceived risk increase

¹⁰ The risks that auditors need to consider can be categorized into a client's business risk, an auditor's business risk, and an audit risk (e.g., Johnstone 2000). The client company business risk is the risk that the client's economic condition will deteriorate in the short or long term; the auditor business risk is the risk that the audit firm will suffer a loss resulting from the engagement (either through a lack of engagement profitability, loss of reputation, or via future litigation); the audit risk is the likelihood of undetected material misstatements in a client's financial statements. Furthermore, audit risk comprises inherent risk (the risk of material misstatements), control risk (the risk that material misstatements will not be prevented or detected or detec

audit effort (and possibly induce a fee premium due to heightened litigation risk), and results in higher audit fees (Simunic 1980; Simunic and Stein 1996). Raghunandan and Rama (2006), Hoitash et al. (2008) and Hogan and Wilkins (2008), for example, found strong evidence on the positive association between internal control weaknesses and audit fees.

3.3 Pervasiveness of internal control weaknesses

Since management is responsible for establishing and maintaining adequate internal controls, it could be argued that the effectiveness or ineffectiveness of an internal control system stems from (and reflects) the competence and overall attitude of management toward internal controls and financial reporting quality (i.e., 'the tone at the top'). Consequently, problems in the control environment could be expected to be pervasively reflected, for example, in internal control assessments and in the credibility of those assessments, and in the operational decisions by the management.

Rice and Weber (2012) show that some existing material weaknesses remain unacknowledged (undiscovered and undisclosed). They suggest that whether material weaknesses are actually acknowledged depends on the incentives for detection and disclosure. Moreover, Bedard and Graham (2011) point out that judging whether the internal control weakness should be designated material is difficult, because of the ambiguity in the definition of materiality. Thus, there are challenges for reliable assessment of internal control effectiveness. The pervasiveness of internal control weaknesses and the challenges in evaluating internal controls are discussed in the first essay, which examines whether the low financial reporting quality of companies disclosing Section 404 material weaknesses persists into the period after the last material weakness disclosure. In particular, the essay investigates whether companies will continue to have a higher likelihood of misstatement in financial information in the post-MW404 period compared to companies without a history of disclosed material weaknesses.

Recent research on internal controls suggests that, in addition to the direct link to accounting quality, material weaknesses may have spillover effects to operations too (Cheng et al. 2013; Bauer et al. 2014; Feng et al. 2015). Considering the governance problems in companies with internal control weaknesses (e.g., Zhang et

by internal controls), and detection risk (the risk that auditor fails to detect material misstatements) (e.g., Eilifsen and Messier 2000).

al. 2007; Hoitash et al. 2009; Skaife et al. 2013) and the importance of the 'tone at the top' in developing internal control process, material weaknesses could be expected to reflect such a business environment, which is permissive to manipulation of real activities to manage earnings. Therefore, not only do internal control weaknesses increase the probability of accounting misstatements, but they also might be associated with real earnings management. The second essay investigates whether the *existence* of material weaknesses in internal controls manifests in real earnings management behavior and whether the subsequent year's *disclosure* of the weaknesses induces company management to employ real earnings management methods.

4 AUDIT QUALITY

The stakeholders of the financial reporting process have different expectations of external auditing, and thus, there is no unanimous agreement on the definition of audit quality (Watkins et al. 2004; Francis 2011; Knechel et al. 2013). Similar to many prior studies, this dissertation adopts the perspective of the users of financial statements, and explores audit quality in the context of client companies' financial reporting quality. Audit outcome, and hence, audit quality itself is unobservable (Knechel et al. 2013). Consequently, the observable audit outputs, audited financial statements and audit reports, have been utilized in prior archival research to examine quality and to test hypotheses (e.g., Francis 2011). The underlying assumption in the fourth essay is that better quality audits should be manifested in better quality financial reporting (for example, smaller abnormal accruals). The third essay takes another approach to audit quality in the context of financial reporting quality. That is, when solely examining client companies with low financial reporting quality, high quality auditing should lead to a greater likelihood of acknowledging existing problems/bias/errors, for example, discovering misstatements or material weaknesses.

Before the audit report and financial statement are issued, the preceding audit process involves a number of phases, which all require judgment and decisionmaking: risk assessment, internal control evaluation, analytical procedures, obtaining and assessing audit evidence, testing, and reviewing (Knechel et al. 2013). The quality of the audit process is dependent on the quality of judgments during each phase. Risk assessment, for example, is vital to the entire audit process, affecting internal control evaluation, the nature and extent of the audit procedures, and testing. However, the judgments are affected by the various circumstances that auditors face (e.g., an audit firm's economic dependence on a client) and by the auditors' individual characteristics (e.g., auditor domain-specific knowledge) (Knechel et al. 2013). The circumstances and personal characteristics influence the probability that an auditor will both discover a breach in a client's accounting system (implying expertise and audit effort) and disclose that breach (implying objectivity and independence), which is the most commonly used definition of audit quality (DeAngelo 1981). The following sections discuss audit quality in the context of non-audit services provided by an incumbent audit firm, and auditor specialization in building expertise.

4.1 Non-audit services and audit quality

Over the last few decades, audit firms have expanded their businesses to include consultancy services, such as those addressing taxation, mergers and acquisitions, and risk management. Audit quality research has investigated two conflicting hypotheses on the association between non-audit services provided by incumbent audit firms and audit quality.

- 1. The knowledge spillover view holds that information acquired in consulting flows to the audit partner, improving audit quality.
- 2. The economic dependence view holds that non-audit fees increase an auditor's economic dependence on the clients, thereby impairing audit quality.

Overall, the previous studies investigating the association between non-audit fees and audit quality have provided mixed results (Schneider et al. 2006). The findings of Frankel et al. (2002), Kanagaretnam et al. (2011), and Rice and Weber (2012) suggest that non-audit fees jeopardize auditor independence and result in a lower audit quality. However, DeFond et al. (2002), Ashbaugh et al. (2003), Chung and Kallapur (2003), and Reynolds et al. (2004) do not find a statistically significant association between non-audit fees and audit quality. Prior research has also investigated whether investors perceive the quality-enhancing or qualitydeteriorating effects of incumbent audit firm-provided non-audit services. Studies examining earnings response coefficients (e.g., Krishnan et al. 2005), market valuation of earnings surprise (Francis and Ke 2006), and cost of equity capital (Khurana and Raman 2006) suggest that investors do perceive non-audit fees to be a threat to auditor independence. However, Gosh et al. (2009) do not find a significant association between non-audit fees and perceived auditor independence.

Particularly after the accounting scandal of Enron and its audit firm Arthur Andersen, investors and regulators became concerned over the magnitude of the fees paid to incumbent audit firms for their non-audit services. Consequently, the SOX (2002) prohibits audit firms from offering audit and certain non-audit services to the same client on a concurrent basis. Because of the potential benefits from knowledge spillover, the SOX permits the provision of tax services. There are, however, certain specific requirements for incumbent audit firms providing tax services, such as an audit committee's pre-approval of the tax services, a separate disclosure of the amount of non-audit fees paid by type of the service (auditrelated fees, tax fees, other fees) and limitations to the scope of the consulting (SEC 2003b; PCAOB 2005).

Since the restrictions on the non-audit services provided by incumbent audit firms were established, the research has focused on examining tax services. The findings of these prior studies indicate that tax services are associated with a reduced likelihood of restatements (Kinney et al. 2004), and of tax-related restatements (Seetharaman et al. 2011), with a greater likelihood of a going concern opinion prior to bankruptcy filing (Robinson 2008), with reduced discretionary accruals (Choi and Lee 2009), a reduced likelihood of loss avoidance (Krishnan and Visvanathan 2011), improved estimates for tax reserves (Gleason and Mills 2011), the value-relevance of earnings (Krishnan et al. 2013), and a reduced likelihood of non-tax internal control weaknesses (Harris and Zhou 2013). In summary, these studies support the knowledge spillover hypothesis. Moreover, Huang et al. (2007) found mostly insignificant associations between tax fees and proxies for financial reporting quality (some weak evidence of lower abnormal accruals and insignificant association with meeting or beating earnings benchmarks). However, a few prior studies have found indications suggesting that tax fees may negatively affect audit quality. Using an experiment, Favere-Marchesi (2006) found that the joint provision of audit and tax services led to significantly lower fraud-risk assessments. Paterson and Valencia (2011) found that the recurring tax services provided by audit firm create knowledge spillover, but nonrecurring tax services seem to have a detrimental impact on auditor independence.

The issue of tax-related fees and financial reporting quality is examined in the third essay. The study specifically focuses on a sample of companies that all have poor financial reporting quality (misstatements in financial information), and investigates whether tax services being provided by incumbent audit firms enhance or impair the likelihood that the client company acknowledges the low financial reporting quality.

4.2 Individual auditor specialization and audit quality

Knechel et al. (2013) refer to an audit as a knowledge-based professional service. Thus, audit performance is affected by the quality of judgments, and hence, the expertise of individual auditors. Expertise is determined by a person's innate abilities (problem-solving abilities) and knowledge (e.g., Bonner and Lewis 1990; Libby 1995). Although education and training develop knowledge, experience and extensive practice are required to acquire expertise in auditing (e.g., Bonner and Lewis 1990; Bédard and Chi 1993; Libby 1995).

An expert can be characterized as an individual with specialized knowledge of the domain (Bédard and Chi 1993). That is, in order to gain expertise, a person must

acquire specialized knowledge in addition to more general knowledge (e.g., Bonner and Lewis 1990; Bedard and Biggs 1991; Bedard and Chi 1993). Specialization acquired by auditing similar types of client companies can foster domainspecific knowledge and in-depth expertise, and result in high quality auditing (e.g., Bonner and Lewis 1990; Libby 1995; Bedard and Chi 1993; Bedard and Biggs 1991; Zerni 2012). Prior studies examining auditors' domain-specific knowledge have primarily examined the specialization in auditing specific industries. The findings of these studies generally indicate higher audit fees charged (Craswell et al. 1995; Zerni 2012) and higher quality auditing (Owhoso et al. 2002; Hammersley 2006; Chin and Chi 2009; Reichelt and Wang 2010; Chi and Chin 2011; Gul et al. 2009; Lim and Tan 2008; Lim and Tan 2010) by industryspecialist auditors.

The majority of the previous archival auditing research has investigated audit quality at the firm-level (e.g., Simunic 1980; Becker et al. 1998; Francis and Krishnan 1999; Balsam et al. 2003). The underlying assumption in these studies is that quality relates to the audit firm's brand name and that knowledge can be distributed across audit offices (e.g., Becker et al. 1998; Francis and Krishnan 1999; Balsam et al. 2003). However, as audit performance depends on the expertise of individual auditors (e.g., Bonner and Lewis 1990), more recent archival research has studied audit quality at the office-level (e.g., Ferguson et al. 2003; Francis et al. 2005; Reichelt and Wang 2010) and at the individual partner-level (e.g., Chin and Chi 2009; Chi and Chin 2011; Zerni 2012). Accordingly, recent research suggests that audit partners' characteristics affects audit quality (Carey and Simnett 2006; Gul et al. 2009; Chin and Chi 2009; Chi and Chin 2011; Zerni 2012; Knechel et al. 2015). In terms of industry specialization, the findings of Chin and Chi (2009), for example, indicate that an individual audit partner's industry specialization is associated with higher quality financial reporting of client companies, but the audit-firm level industry specialization does not of itself lead to better quality auditing.

Zerni (2012) points out that audit firms organize their business lines, not only based on industry sectors, but also according to criteria like client size and ownership structure. However, there is a lack of research on other dimensions of specialization in determining domain-specific knowledge. An exception is the study by Zerni (2012) that finds that both auditor industry specialization and specialization in auditing public companies are associated with higher audit fees. Extending the research on auditor specialization, the fourth essay examines the association between an individual audit partner's public-client specialization and audit/financial reporting quality.

5 SUMMARY OF THE ESSAYS

5.1 The persistence in the association between Section 404 material weaknesses and financial reporting quality

The first essay investigates whether MW404 disclosures are predictive of future financial reporting quality. In particular, the paper examines whether the low financial reporting quality of MW404 companies persists into the post-MW404 period. Given that changing organizational policies takes time (Kotter 1995) it is intuitively appealing to assume that this kind of persistence occurs. Because material weaknesses in internal controls carry a threat that material misstatements are not detected in a timely manner, the current study relies on the view that an incidence of a misstatement indicates a failure in a company's internal controls (e.g., Eilifsen and Messier 2000; Leone 2007; Rice and Weber 2012). If companies in the post-MW404 period have not yet reached as high level of internal control effectiveness as companies without a history of MW404s, the likelihood of incidences of misstatement would be higher in the post-MW404 period too.

The empirical findings indicate that there is a greater likelihood of undiscovered material misstatements in financial information among MW404 companies and companies in the post-MW404 period compared to companies without a history of MW404s (referred to as EIC companies, i.e., companies with effective internal controls). On average, the greater likelihood of misstatements is estimated to persist for two years. That is, in the two years immediately following the last MW404 disclosure, internal control over financial reporting is still not as effective at preventing or detecting misstatements in a timely manner as it is in EIC companies. The magnitude of the effect, however, decreases non-linearly (i.e., first rapidly then slowly).

When exploring the possible explanations for the empirical findings, the additional descriptive analysis provides some evidence that companies with undiscovered misstatements in the post-MW404 period have previously disclosed more entitylevel internal control problems (multiple account-specific MW404s, and entitylevel MW404s related to accounting personnel training and competence, year-end adjustment, and untimely or inadequate account reconciliations) compared to those companies without undiscovered misstatements. More interestingly, the exploration reveals that the majority of the misstatements in the post-MW404 period are unrelated to the previously disclosed account-specific MWs. It appears that many companies with misstatements in the post-MW404 period have even more pervasive internal control problems than reported in the last MW404 disclosure.

Overall, the findings of this study indicate that in the post-MW404 period, there is a greater likelihood of existing control problems remaining unacknowledged. It might be that management's assertion of the effectiveness of internal controls is too easily accepted by some auditors. Moreover, due to insufficient expertise and/or lack of resources, some auditors might concentrate their effort on the previously discovered MW problem and not adequately examine other aspects of financial reporting. These inferences bring additional insights to the problem of undiscovered and undisclosed control weaknesses, suggesting the need to develop auditors' competence in evaluating the effectiveness of internal control over financial reporting.

5.2 Real earnings management before and after reporting SOX 404 material weaknesses

The second essay investigates whether the existence of Section 404 material weaknesses manifests in real earnings management behavior and/or whether the disclosure of material weaknesses induces company management to employ real earnings management. Firstly, it is examined whether real earnings management is employed in company years with ineffective internal controls based on a subsequent SOX 404 internal control report - that is, material weaknesses exist, but have not yet been disclosed. Because of management responsibility for establishing and maintaining adequate internal controls (SOX 2002), weaknesses in internal controls implicitly create doubts about management's competence and its attitude toward financial reporting and/or in extreme cases, even its integrity. Material weaknesses particularly reflect pervasive problems in the control environment (e.g., Zhang et al. 2007; Hoitash et al. 2009), and thus these weaknesses might stem from the 'tone at the top'. More recent studies on internal control effectiveness suggest that weaknesses in internal control over financial reporting have spillover effects to operations (Cheng et al. 2013; Bauer et al. 2014). The existence of material weaknesses could hence be expected to reflect such a business environment, which is permissive to real earnings management. Moreover, given that material weaknesses per se frequently reduce the quality of internal information (Feng et al. 2015), it is possible that management is making (unintentionally) poor operational decisions based on this information, which manifest in greater real earnings management.

The essay goes on to examine whether companies disclosing material weaknesses relating to the prior period have higher levels of real earnings management. Higher levels of real earnings management could be expected in companies that have recently attracted bad publicity due to material weakness disclosures. Prior research has shown that the disclosures of internal control deficiencies are perceived negatively in debt and equity markets (e.g., Ashbaugh-Skaife et al. 2009; Kim et al. 2011), and lead to a decline in customer demand (Su et al. 2014). The negative consequences anticipated from material weakness disclosure (e.g., negative investor reactions, increased cost of debt, or impact on personal reputation) mean management could be expected to strive to mitigate these concerns and therefore have an incentive for earnings management by manipulating real operational activities, which is not something easily detected or constrained by outsiders.

The sample used in this study comprises fiscal year observations of US listed companies from 2004 to 2012. The real earnings management methods investigated are inventory overproduction (proxied by abnormal levels of production costs), reduction of discretionary expenses (proxied by abnormal levels of discretionary expenses), and sales manipulation (proxied by abnormal levels of cash flow from operations) (Roychowdhury 2006). The main analyses are additionally conducted using a propensity-score matched sample.

The empirical findings indicate that companies with material weaknesses in their internal controls have higher levels of real activities manipulation (particularly inventory overproduction but also the reduction of discretionary expenses) compared to companies with effective internal controls. This implies that the weak commitment by management to provide effective internal control systems and high quality financial information relates to a tendency to use real earnings management methods and also impairs management's real operational decisions. Furthermore, the empirical results indicate that companies employ real earnings management (overproduction and reduction of discretionary expenses) after disclosing a previous year's material weaknesses. It appears that the public disclosure of material weaknesses induces management to strive to manage the expected negative reactions of stakeholders to the disclosure by engaging in real earnings management, which is not easily detected or constrained by outsiders. Overall, this study suggests that material weaknesses in internal controls signal an environment where management is more inclined to employ real earnings management.

5.3 Incumbent audit firm-provided tax services and clients with low financial reporting quality

The third essay investigates whether tax services provided by incumbent audit firms enhance or impair the likelihood of acknowledging client companies' low financial reporting quality. As discussed in Section 4.1, SOX (2002) prohibits audit firms from providing most non-audit services to their audit clients, but permits tax services due to the potential benefits from knowledge spillover. This study approaches the issue of incumbent audit firm-provided tax services by using a sample of companies with poor financial reporting quality, that is, companies with misstatements. The misstatements are determined from the restated periods indicated by restatement data. First, the essay investigates whether tax fees are associated with restatement lags, in other words, those misstatements in financial information that remain undiscovered in a particular fiscal year. That is, the companies with a restatement lag are compared to those companies with misstatements more quickly restated. Second, the essay investigates whether tax fees are associated with the likelihood of Section 404 internal control weakness disclosures among companies with misstatements. Based on the view that an incidence of a misstatement indicates underlying internal control weaknesses (e.g., Eilifsen and Messier 2000; Rice and Weber 2012), material weakness disclosures would suggest greater scrutiny by auditors. The research setting featured a sample of similar companies in terms of poor accounting quality, allows examining the auditors' professional skepticism in particular. Enhanced knowledge of the client acquired via the provision of tax services could make restatements more timely and material weakness disclosures more likely for companies with poor accounting quality. However, economic dependence may disrupt an auditor's professional skepticism, resulting in restatement lags and unacknowledged control problems.

The inferences of the findings in prior studies investigating the association between tax services being provided by an incumbent audit firm and financial reporting quality largely support the knowledge spillover view. For example, Kinney et al. (2004) suggest that tax fees reduce the likelihood of restatements, implying there are benefits from knowledge spillover. Seetharaman et al. (2011), however, report an insignificant association between tax fees and restated periods, but a significant negative association with tax-related restatements. These studies examine whether or not restatements/misstatements occur. Harris and Zhou (2013) suggest that tax consulting leads to a reduced likelihood of non-tax-related internal control weaknesses but does not have an effect on tax-related weaknesses. Lower likelihood of internal control weakness disclosures could, however, also indicate a reluctance to disclose weaknesses. Rice and Weber (2012) examine a sample of companies with existing internal control weaknesses and conclude that larger non-audit fees makes it less likely that material weaknesses are disclosed, supporting the economic dependence view. The current study extends the findings of these prior studies and investigates the role of tax services being provided by audit firm among companies with poor financial quality, using a sample of fiscal-year observations of US companies from 2005–2012.

The empirical findings indicate that higher tax-related fees are associated with a reduced likelihood of SOX 404 internal control weakness disclosures for companies with misstatements, implying that underlying control problems are unacknowledged. However, the findings suggest that just providing both audit and tax services does not itself have an impact on audit quality, but rather it is the magnitude of the tax-related fees in particular that counts. The results also provide modest evidence suggesting that lower levels of tax-related fees are associated with a lower likelihood of restatement lags, that is, misstatements are restated in a more timely manner. Overall, the findings of this study provide some evidence suggesting that, among companies with poor accounting quality, greater economic bond with the clients might impair auditors' professional skepticism. In other words, auditors' scrutiny of their client is weaker when the magnitude of fees generated from tax services provided to that client is higher.

5.4 Audit partner public-client specialization and client abnormal accruals

The fourth essay examines whether the extent an audit partner specializes in public-clients is associated with abnormal accruals, a proxy for client companies' audit quality/financial reporting quality. Prior research has provided extensive evidence of the effects of audit firm or local audit office-level characteristics on both audit quality (e.g., Reynolds and Francis 2001; Balsam et al. 2003; Krishnan 2005; Reichelt and Wang 2010) and audit fees (e.g., Craswell et al. 1995; Ferguson et al. 2003). However, the empirical evidence on how individual audit partner specialization affects client financial reporting outcomes is limited due to the wide-spread absence of audit partner signature on audit reports, which would enable the identification of individual partners with specific client engagements.

Prior research has mainly investigated auditors' specialization in different industries, and audit partner industry specialization has been found to be positively associated with audit quality (Chin and Chi 2009; Chi and Chin 2011). Specialization in public companies is another means to gain domain-specific knowledge (Zerni 2012). Auditing public-clients requires specialist knowledge of the relevant financial reporting and auditing requirements. Zerni (2012) finds that both industry specialization and specialization in public companies are associated with higher audit fees. However, prior research has not investigated whether public-client specialization is associated with actual audit outcomes that are indicative of higher quality. In addition to the enhanced expertise achieved through public-client specialization, the willingness to resist client pressure is likely to increase with the number of public clients in the partner's portfolio as the partner's dependence on any one client diminishes, which should help to ensure audit quality (Johnstone et al. 2001).

This study uses a sample of 420 company-year observations from the NASDAQ OMX Exchange in Finland. The identities of audit partners assigned to publicclient engagements are linked with client financial statement data. The findings reveal a negative association between greater public-client specialization and absolute abnormal working capital accruals. Moreover, the findings indicate that the negative association between public-client specialization and abnormal accruals only occurs for partners with three to six public clients; a moderate level of public-client specialization appears optimal, and the busyness that goes with having a high level of such specialization appears to mitigate the knowledge and independence benefits associated with this type of specialization. Finally, the results reveal that companies audited by partners with a higher level of public-client specialization have significantly smaller income-decreasing abnormal accruals. In the setting with high-tax and high-alignment between financial reporting and tax accounting, auditors with a greater public-client specialization might be more likely to recognize the negative reputational implications of their clients' aggressive tax minimization strategies and discourage income-decreasing earnings management. Overall, this study suggests that audit partner public-client specialization is associated with improved audit quality, reveals the extent of public-client specialization that appears to be optimal, and the nature of earnings management that is mitigated by partner public-client specialization.

References

Ashbaugh, H., R. LaFond & B. Mayhew (2003). Do nonaudit services compromise auditor independence? Further evidence. *The Accounting Review* 78:3, 611–639.

Ashbaugh-Skaife, H., D. W. Collins & W. R. Kinney, Jr. (2007). The discovery and reporting of internal control deficiencies prior to SOX-mandated audits. *Journal of Accounting and Economics* 44, 166–192.

Ashbaugh-Skaife, H., D. W. Collins, W. R. Kinney, Jr. & R. LaFond (2008). The effect of SOX internal control deficiencies and their remediation on accrual quality. *The Accounting Review* 83:1, 217–250.

Ashbaugh-Skaife, H., D. W. Collins, W. R. Kinney, Jr. & R. LaFond (2009). The effect of SOX internal control deficiencies on firm risk and cost of equity. *Journal of Accounting Research* 47:1, 1–43.

Ball, R. & L. Shivakumar (2005). Earnings quality in UK private firms: comparative loss recognition timeliness. *Journal of Accounting and Economics* 39: 83– 128.

Ball, R. & L. Shivakumar (2006). The role of accruals in asymmetrically timely gain and loss recognition. *Journal of Accounting Research* 44:2, 207–242.

Balsam, S., J. Krishnan & J. Yang (2003). Auditor industry specialization and earnings quality. *Auditing: A Journal of Practice & Theory* 22:2, 71–97.

Bauer, A. (2014). Tax avoidance and the implications of weak internal controls. *Working paper*. University of Illinois at Urbana-Champaign.

Becker, C., M. DeFond, J. Jiambalvo & K. R. Subramanyam (1998). The effect of audit quality on earnings management. *Contemporary Accounting Research* 15:1, 1–24.

Bédard, J. & M. T. H. Chi (1993). Expertise in auditing. *Auditing: A Journal of Practice & Theory* 12:Supplement, 21–45.

Bedard, J. C. & S. Biggs (1991). The effect of domain-specific experience on evaluation of management representations in analytical procedures. *Auditing: A Journal of Practice & Theory* 10:Supplement, 77–90.

Bedard, J. C. & L. Graham (2011). Detection and severity classifications of Sarbanes-Oxley Section 404 internal control deficiencies. *The Accounting Review* 86:3, 825–855.
Bedard, J. C., R. Hoitash, U. Hoitash & K. Westermann (2012). Material weakness remediation and earnings quality: A detailed examination by type of control deficiency. *Auditing: A Journal of Practice & Theory* 31:1, 57–78.

Beneish, M. D., M. B. Billings & L. D. Hodder (2009). Internal control weaknesses and information uncertainty. *The Accounting Review* 83:3, 665–703.

Bonner, S. E. & B. L. Lewis (1990). Determinants of auditor expertise. *Journal of Accounting Research* 28:Supplement, 1–20.

Bushman, R. M. & A. J. Smith (2001). Financial accounting information and corporate governance. *Journal of Accounting and Economics* 32, 237–333.

Carey, P. & R. Simnett (2006). Audit partner tenure and audit quality. *The Accounting Review* 81:3, 653–676.

Chan, K.C., B. Farrell & P. Lee (2008). Earnings management of firms reporting material internal control weaknesses under section 404 of the Sarbanes-Oxley act. *Auditing: A Journal of Practice & Theory* 27:2, 161–179.

Cheng, M., D. Dhaliwal & Y. Zhang (2013). Does investment efficiency improve after the disclosure of material weaknesses in internal control over financial reporting? *Journal of Accounting and Economics* 56:1, 1–18.

Chi, H. & C. Chin (2011). Firm versus partner measures of auditor industry expertise and effects on auditor quality. *Auditing: A Journal of Practice & Theory* 30:2, 201–229.

Chin, C-L. & H. Y. Chi (2009). Reducing restatements with increased industry expertise. *Contemporary Accounting Research* 26:3, 729–65.

Choi, W.-W. & H.-Y. Lee (2009). The provision of tax services by incumbent auditors and earnings management: evidence from Korea. *Journal of International Financial Management and Accounting* 20:1, 79–103.

Chung, H. & S. Kallapur (2003). Client importance, nonaudit services, and abnormal accruals. *The Accounting Review* 78:4, 931–955.

Cohen, D. A., A. Dey & T. Z. Lys (2008). Real and accrual-based earnings management in the pre- and post-Sarbanes-Oxley periods. *The Accounting Review* 83:3, 757–787.

COSO (1992). Committee of Sponsoring Organization of Treadway Commission. *Internal Control – Integrated Framework*. New York: AICPA.

Council of the European Union (2006). Directive 2006/43/EC on statutory audits of annual accounts and consolidated accounts, amending Council Directives 78/60/EEC and 83/349/EEC and repealing Council Directive 84/253/EEC. *Official Journal of the European Union* L157/87 9 June 2006.

Craswell, A. T., J. R. Francis & S. Taylor (1995). Auditor brand name reputations and industry specializations. *Journal of Accounting and Economics* 20:3, 297–322.

DeAngelo, L. E. (1981). Auditor size and audit quality. *Journal of Accounting and Economics* 3, 183–199.

Dechow, P. M. & I. D. Dichev (2002). The quality of accruals and earnings: the role of accrual estimation errors. *The Accounting Review* 77:4, 35–59.

Dechow, P., W. Ge & C. Schrand (2010). Understanding earnings quality: A review of the proxies, their determinants and their consequences. *Journal of Accounting and Economics* 50, 344–401.

Dechow, P. M., R. G. Sloan & A. P. Sweeney (1995). Detecting earnings management. *The Accounting Review* 70:2, 193–225.

DeFond, M. L. & J. R. Francis (2005). Audit research after Sarbanes-Oxley. *Auditing: A Journal of Practice & Theory* 24:Supplement, 5–30.

DeFond, M. L. & J. Jiambalvo (1991). Incidence and circumstances of accounting errors. *The Accounting Review* 66:3, 643–655.

DeFond, M. L. & C. W. Park (2001). The reversal of abnormal accruals and the market valuation of earnings surprises. *The Accounting Review* 76:3, 5–30.

DeFond, M. L., K. Raghunandan & K. Subramanyam (2002). Do nonaudit service fees impair auditor independence? Evidence from going-concern audit opinions. *Journal of Accounting Research* 40:4, 1247–1274.

Doyle, J., W. Ge & S. McVay (2007a). Accruals quality and internal control over financial reporting. *The Accounting Review* 82:5, 1141–1170.

Doyle, J., W. Ge & S. McVay (2007b). Determinants of weaknesses in internal control over financial reporting. *Journal of Accounting and Economics* 44, 193–223.

Eilifsen, A. & W. F. Messier, Jr. (2000). The incidence and detection of misstatements: A review and integration of archival research. *Journal of Accounting Literature* 19, 1–43.

Eisenhardt, K. M. (1989). Agency theory: An assessment and review. Academy of Management Review 14:1, 57–74.

Fama, E. F. (1980). Agency problems and the theory of the firm. *Journal of Political Economy* 88:2, 288–307.

Fama, E. F. & M. C. Jensen (1983). Separation of ownership and control. *Journal of Law & Economics* 26:2, 301–325.

FASB (2010). Financial Accounting Standards Board. *Conceptual framework for financial reporting*. Financial Accounting Foundation.

Favere-Marchesi, M. (2006). The impact of tax services on auditors' fraud-risk assessment. *Advances in Accounting* 22, 149–165.

Feng, M., C. Li, S. E. McVay & H. Skaife (2015). Does ineffective internal control over financial reporting affect a firm's operations? Evidence from firms' inventory management. *The Accounting Review* 90:2, 529–557.

Ferguson, A., J. R. Francis & D. J. Stokes (2003). The effects of firm-wide and office-level industry expertise on audit pricing. *The Accounting Review* 78:2), 429–448.

Francis, J. R. (2004). What do we know about audit quality? *The British Accounting Review* 36, 345–368.

Francis, J. R. (2011). A framework for understanding and researching audit quality. *Auditing: A Journal of Practice and Theory* 30, 125–152.

Francis, J. R. & B. Ke (2006). Disclosure of fees paid to auditors and the market valuation of earnings surprises. *Review of Accounting Studies* 11:4, 495–523.

Francis, J. R. & Krishnan, J. (1999). Accounting accruals and auditor reporting conservatism. *Contemporary Accounting Research* 16, 135–165.

Francis, J. R., K. Reichelt, and D. Wang (2005). The pricing of national and cityspecific reputations for industry expertise in the U.S. audit market. *The Accounting Review* 80, 113–136.

Frankel, R., M. Johnson & K. Nelson (2002). The relation between auditors' fees for nonaudit services and earnings management. *The Accounting Review* 77:4, 71–105.

Ge, W. & S. McVay (2005). The disclosure of material weaknesses in internal control after the Sarbanes-Oxley act. *Accounting Horizons* 19:3, 137–158.

Gleason, C. A. & L. F. Mills (2011). Do auditor-provided tax services improve the estimate of tax reserves? *Contemporary Accounting Research* 28:5, 1484–1509.

Goh, B. W. (2009). Audit committees, board of directors, and remediation of material weaknesses in internal control. *Contemporary Accounting Research* 26:2, 549–579.

Gosh, A., S. Kallapur & D. Moon (2009). Audit and non-audit fees and capital market perceptions of auditor independence. *Journal of Accounting and Public Policy* 28, 369–385.

Graham, J. R., C. R. Harvey & S. Rajgopal (2005). The economic implications of corporate financial reporting. *Journal of Accounting and Economics* 40:1-3, 3–73.

Gul, F. A., Fung, S. Y. K. & Jaggi, B. (2009). Earnings quality: Some evidence on the role of auditor tenure and auditors' industry expertise. *Journal of Accounting and Economics* 47:3, 265–287.

Hammersley, J. S. (2006). Pattern identification and industry-specialist auditors. *The Accounting Review* 81:2, 309–336.

Harris, D., & J. Zhou (2013). Auditor-provided tax consulting, knowledge spillovers, and reported weaknesses in internal control. *Working paper*. Syracuse University, University of Hawaii at Manoa.

Hoag, M. L. & C. W. Hollingsworth (2011). An intertemporal analysis of audit fees and section 404 material weaknesses. *Auditing: A Journal of Practice & Theory* 30:2, 173–200.

Hogan, C. & M. Wilkins (2008). Evidence on the audit risk model: Do auditors increase audit fees in the presence of internal control deficiencies? *Contemporary Accounting Research* 25:1, 219–242.

Hoitash, R., U. Hoitash & J. C. Bedard (2008). Internal control quality and audit pricing under the Sarbanes-Oxley Act. *Auditing: A Journal of Practice & Theory* 27:1, 105–126.

Hoitash, R., U. Hoitash & J. C. Bedard (2009). Corporate governance and internal control over financial reporting: a comparison of regulatory regimes. *The Accounting Review* 84:3, 839–867.

Huang, H-W., S. Mishra & K. Raghunandan (2007). Types of nonaudit fees and financial reporting quality. *Auditing: A Journal of Practice & Theory* 26:1, 133–145.

IASB (2010). International Accounting Standards Board. *The conceptual framework for financial reporting*. IFRS Foundation.

Jensen, M. C. & W. H. Meckling (1976). Theory of the firm: Managerial behavior, agency costs and ownership structure. *Journal of Financial Economics* 3:4, 305–360.

Johnstone, K. M. (2000). Client-acceptance decisions: Simultaneous effects of client business risk, audit risk, auditor business risk, and risk adaptation. *Auditing: A Journal of Practice & Theory* 19:1, 1–25.

Johnstone, K. M. & J. C. Bedard (2001). Engagement planning, bid pricing, and client response in the market for initial attest engagements. *The Accounting Review* 76:2, 199–220.

Johnstone, K. M., C. Li & K. H. Rupley (2011). Changes in corporate governance associated with the revelation of internal control material weaknesses and their subsequent remediation. *Contemporary Accounting Research* 28:1, 331–383.

Johnstone, K. M., M. Sutton & T. Warfield (2001). Antecedents and consequences of independence risk: Framework for analysis. *Accounting Horizons* 15:1, 1–18.

Jones, J. (1991). Earnings management during import relief investigations. *Journal of Accounting Research* 29:2, 193–228.

Kanagaretnam, K., G. Krishnan & G. Lobo (2011). An empirical analysis of auditor independence in the banking industry. *The Accounting Review* 85:6, 2011– 2046.

Khurana, I. & K. K. Raman (2006). Do investors care about the auditor's economic dependence on the client? *Contemporary Accounting Research* 23:4, 977– 1016.

Kim, J., B. Y. Song, and L. Zhang (2011). Internal control weakness and bank loan contracting: Evidence from SOX Section 404 disclosures. *The Accounting Review* 86:4, 1157–1188.

Kinney Jr., W. R. (2000). Research opportunities in internal control quality and quality assurance. *Auditing: A Journal of Practice & Theory* 19, 83–90.

Kinney Jr., W. R., Z.-V. Palmrose & S. Scholz (2004). Auditor independence, non-audit services, and restatements: was the U.S. government right? *Journal of Accounting Research* 42:3, 561–588.

Knechel, W. R., G. V. Krishnan, M. Pevzner, L. B. Shefchik & U. K. Velury (2013). Audit quality: Insights from the academic literature. *Auditing: A Journal of Practice & Theory* 32:1, 385–421.

Knechel, W. R., A. Vanstraelen, and M. Zerni (2015). Does the identity of engagement partners matter? An analysis of the persistence and economic consequences of low audit partner quality. *Contemporary Accounting Research*, Forthcoming.

Kothari, S., A. Leone & C. Wasley (2005). Performance matched discretionary accruals measures. *Journal of Accounting and Economics* 39, 163–197.

Kotter, J. P. (1995). Leading change: Why transformation efforts fail. *Harvard Business Review* 73:2, 59–67.

Krishnan, G. V. (2005). The association between Big 6 auditor industry expertise and the asymmetric timeliness of earnings. *Journal of Accounting Auditing & Finance* 20:3, 209–228.

Krishnan, G. V. & G. Visvanathan (2007). Reporting internal control deficiencies in the post-Sarbanes-Oxley Era: The role of auditors and corporate governance. *International Journal of Auditing* 11, 73–90.

Krishnan, G. V. & G. Visvanathan (2011). Is there an association between earnings management and auditor-provided tax services? *Journal of the American Taxation Association* 33:2, 111–135.

Krishnan, G. V., G. Visvanathan & W. Yu (2013). Do auditor-provided tax services enhance or impair the value relevance of earnings? *Journal of the American Taxation Association* 35:1, 1–19.

Krishnan, J., D. Rama & Y. Zhang (2008). Costs to comply with SOX Section 404. *Auditing: A Journal of Practice & Theory* 27:1, 169–186.

Krishnan, J., H. Sami & Y. Zhang (2005). Does the provision of nonaudit services affect investor perceptions of auditor independence? *Auditing: A Journal of Practice & Theory* 24:2, 111–135.

Leone, A. (2007). Factors related to internal control disclosure: a discussion of Ashbaugh, Collins, and Kinney (2007) and Doyle, Ge, and McVay (2007). *Journal of Accounting and Economics* 44, 224–237.

Li, C., L. Sun & M. Ettredge (2010). Financial executive qualifications, financial executive turnover, and adverse SOX 404 opinions. *Journal of Accounting and Economics* 50:1, 93–110.

Libby, R. (1995). The role of knowledge and memory in audit judgment. In: *Judgment and Decision-Making Research in Accounting and Auditing*, 176–206. Ed. A. H. Ashton and R. H. Ashton. New York: Cambridge University Press.

Lim, C.-Y. & H.-T. Tan (2008). Non-audit service fees and audit quality: The impact of auditor specialization. *Journal of Accounting Research* 46:1, 199–246.

Lim, C.-Y. & H.-T. Tan (2010). Does auditor tenure improve audit quality? Moderating effects of industry specialization and fee dependence. *Contemporary Accounting Research* 27:3, 923–957.

Lopez, . J., S. D. Vandervelde & Y. Wu (2009). Investor perceptions of an auditor's adverse internal control opinion. *Journal of Accounting and Public Policy* 28:3, 231–250.

McNichols, M. (2002). Discussion of the quality of accruals and earnings: The role of accrual estimation errors. *The Accounting Review* 77:Supplement, 61–69.

Nagy, A. L. (2010). Section 404 Compliance and Financial Reporting Quality. *Accounting Horizons* 24:3, 441–454.

O'Keefe, T. B., D. Simunic & M. Stein (1994). The production of audit services: Evidence from a major public accounting firm. *Journal of Accounting Research* 32:2, 241–261.

Owhoso, V. E., W. F. Messier Jr. & J. G. Lynch (2002). Error detection by industry-specialized teams during sequential audit review. *Journal of Accounting Research* 40:3, 883–900.

Paterson, J. S. & A. Valencia (2011). The effects of recurring and nonrecurring tax, audit-related, and other nonaudit services on auditor independence. *Contemporary Accounting Research* 28:5, 1510–1536.

PCAOB (2005). Public Company Accounting Oversight Board. *Ethics and independence rules concerning independence, tax services, and contingent fees.* PCAOB Rulemaking Docket Matter No. 017, Release No. 2005-014. Available from: http://www.pcaobus.org/.

PCAOB (2007). Public Company Accounting Oversight Board. Auditing Standard No. 5 – An audit of internal control over financial reporting that is integrated with an audit of financial statements.

Raghunandan, K. & D. Rama (2006). SOX Section 404 material weakness disclosures and audit fees. *Auditing: A Journal of Practice & Theory* 25:1, 99–114.

Reichelt, K. & D. Wang (2010). National and office-specific measures of auditor industry expertise and effects on audit quality. *Journal of Accounting Research* 48:3, 647–686.

Reynolds, J. K., D. R. Deis, Jr. & J. R. Francis (2004). Professional service fees and auditor objectivity. *Auditing: A Journal of Practice & Theory* 23:1, 29–52.

Reynolds, J. K. & J. R. Francis (2001). Does size matter? The influence of large clients on office-level auditor reporting decisions. *Journal of Accounting and Economics* 30:3, 375–400.

Rezee, Z., R. Espahbodi, P. Espahbodi & H. Espahbodi (2012). Firm characteristics and stock price reaction to SOX 404 compliance. *Abacus* 48:4, 473–501.

Rice, S. C. & D. P. Weber (2012). How effective is internal control reporting under SOX 404? Determinants of the (non-)disclosure of existing material weaknesses. *Journal of Accounting Research* 50:3, 811–843.

Robinson, D. (2008). Auditor independence and auditor-provided tax service: Evidence from going-concern audit opinions prior to bankruptcy filings. *Auditing: A Journal of Practice & Theory* 27:2, 31–54.

Roychowdhury, S. (2006). Earnings management through real activities manipulation. *Journal of Accounting and Economics* 42:3, 335–370.

Schneider, A. & B. K. Church (2008). The effect of auditors' internal control opinions on loan decisions. *Journal of Accounting and Public Policy* 27:1, 1–18.

Schneider, A., B. K. Church & K. M. Ely (2006). Non-audit services and auditor independence: a review of the literature. *Journal of Accounting Literature* 25: 169–211.

Schneider, A., A. A. Gramling, D. R. Hermanson & Z. Ye (2009). A review of academic literature on internal control reporting under SOX. *Journal of Accounting Literature* 28, 1–46.

SEC (2003a). Securities and Exchange Commission. *Final rule: Management's report on internal control over financial reporting and certification of disclosure in exchange act periodic reports.* Washington, DC. Available from: http://www.sec.gov/.

SEC (2003b). Securities and Exchange Commission. *Strengthening the commission's requirements regarding auditor independence*. Washington, DC. Available from: http://www.sec.gov/.

SEC (2010). Securities and Exchange Commission. *Internal control over financial reporting in exchange act periodic reports of non-accelerated filers*. Washington, DC. Available from: http://www.sec.gov/.

Seetharaman, A., Y. Sun & W. Wang (2011). Tax-related financial statement restatements and auditor-provided tax services. *Journal of Accounting, Auditing & Finance* 26:4, 677–698.

Simunic, D. (1980). The pricing of financial audit services: Theory and evidence. *Journal of Accounting Research* 18:1, 161–190.

Simunic, D. & M. Stein (1996). The impact of litigation risk on audit pricing: A review of the economics and the evidence. *Auditing: A Journal of Practice & Theory* 15:2, 119–134.

Skaife, H. A., D. Veenman & D. Wangerin (2013). Internal control over financial reporting and managerial rent extraction: Evidence from the profitability of insider trading. *Journal of Accounting and Economics* 55:1, 91–110.

SOX (2002). *The Sarbanes-Oxley act of 2002*. United States House of Representatives, Public Law 107–204 [H.R. 3763]. Washington D.C.

Su, L., X. Zhao & G. Zhou (2014). Do customers respond to the disclosure of internal control weaknesses? *Journal of Business Research* 67:7, 1508–1518.

Watkins, A. L., W. Hillison & S. E. Morecroft (2004). Audit quality: a synthesis of theory and empirical evidence. *Journal of Accounting Literature* 23, 153-193.

Watts, R. & J. Zimmerman (1983). Agency problems, auditing, and the theory of the firm: Some evidence. *Journal of Law & Economics* 26:3, 613–633.

Watts, R. & J. Zimmerman (1986). Positive Accounting Theory. Prentice-Hall.

Zerni, M. (2012). Audit partner specialization and audit fees: Some evidence from Sweden. *Contemporary Accounting Research* 29:1, 312–340.

Zhang, Y., J. Zhou & N. Zhou (2007). Audit committee quality, auditor independence, and internal control weaknesses. *Journal of Accounting and Public Policy* 26:3, 300–327.

The Persistence in the Association between Section 404 Material Weaknesses and Financial Reporting Quality,

published in Auditing: A Journal of Practice & Theory Vol. 33, No. 1, Feb 2014.

Due to copyright restrictions, this article is only available in the printed doctoral thesis pages 34–57 or in the journal *Auditing: A Journal of Practice & Theory* 33:1, 93-116.

REAL EARNINGS MANAGEMENT BEFORE AND AFTER REPORTING SOX 404 MATERIAL WEAKNESSES⁺

The co-author of this essay is Tuukka Järvinen

ABSTRACT

The purpose of this study is to investigate whether the *existence* of SOX section 404 material weaknesses manifests in real earnings management behavior and/or whether the *disclosure* of material weaknesses induces company management to employ real earnings management. First, the empirical findings indicate that companies with material weaknesses in their internal controls engage in more manipulation of real activities (particularly inventory overproduction, but also reduction of discretionary expenses) compared to companies with effective internal controls. This implies that the weak commitment by management to provide effective internal control system and high quality financial information relates to a tendency to use real earnings management methods and also impairs management's real operational decisions. Second, the empirical results indicate that companies employ real earnings management (overproduction and reduction of discretionary expenses) after disclosing previous year's material weaknesses. It appears that the public disclosure of material weaknesses induces management to strive to mitigate the expected negative reactions of stakeholders to the disclosure by engaging in real earnings management, which is not easily detected or constrained by outsiders. Overall, this study suggests that material weaknesses in internal controls signal an environment where management is more inclined to employ real earnings management.

Keywords: internal control, material weakness; real earnings management

Acknowledgements: The authors gratefully acknowledge the comments received from Sami Vähämaa, Juha Kinnunen, Stefan Sundgren, Erkki K. Laitinen, Mikko Zerni, Jukka Sihvonen, and Jouni Laitinen. Moreover, the authors appreciate the thoughtful comments by the discussant Jere Francis and the participants at the PhD Workshop of the EARNet Symposium in Trier (Germany), the comments by the participants at the 6th International Workshop on Accounting & Regulation in Siena (Italy), and the comments by the seminar participants at the Department of Financial Economics and Accounting of Pablo de Olavide University in Seville (Spain). Myllymäki gratefully acknowledges the financial support received from the Finnish Foundation for Economic Education, Jenny and Antti Wihuri Foundation, Nordea Bank Foundation, Oskar Öflund's Foundation, and Evald and Hilda Nissi Foundation. Any remaining errors are our own.

1 Introduction

The past two decades have seen a series of well-publicized failures of corporate governance, some being related to accounting scandals, some to the financial crisis. Such failures make it essential for both the financial markets and accounting research to improve the understanding of internal control effectiveness; of the pervasiveness of control issues; and of how they relate to earnings management behavior.¹ Utilizing internal control reports mandated by Section 404 of the Sarbanes-Oxley Act (SOX) of 2002, this study investigates whether the existence of material weaknesses in internal controls manifests in real earnings management behavior, and whether the subsequent disclosure of weaknesses induces management to employ real earnings management methods.² Real earnings management refers to the manipulation of real operational activities to achieve certain financial goals, which do not necessarily enhance long-term company value but are instead likely to decrease future cash flows (e.g., Roychowdhury 2006). Real earnings management, as opposed to accrual-based earnings management, is not related to accounting issues *per se*. Consequently, real earnings management can be fully executed within the generally accepted accounting principles (GAAP), and (after the proper disclosure of these transactions in the financial statements) does not necessarily fall within the purview of auditors or regulators (Cohen et al. 2008; Kim et al. 2010). Considering the wide-ranging use of real activities manipulation to manage earnings instead of or in addition to accounting manipulation (e.g., Graham et al. 2005; Cohen et al. 2008) and the pervasive problems in the control environment and governance in companies with internal control weaknesses (e.g., Zhang et al. 2007; Hoitash et al. 2009; Skaife et al. 2013), it is interesting to examine whether real earnings management is greater in companies with material internal control weaknesses.³

¹ Related to internal controls in general, it is stated in a report of global survey on risk management and internal control that "Strong risk management and internal control systems are crucial to strengthening the governance of organizations, and therefore the global financial infrastructure, and helping to prevent future crises." (PAIB 2011)

² Section 404 requires Securities and Exchange Commission (SEC) registrants' managers to assess and report on the effectiveness of internal control over financial reporting, and to disclose the existence of any material weaknesses. Moreover, companies' auditors must also evaluate and report on the effectiveness of internal controls. Section 404 became effective for fiscal years ending after November 15, 2004 for the largest companies (market capitalization of at least 75 million dollars). Since 2007, SEC has required also smaller public companies to report on internal control effectiveness, but do not require external auditors internal control reporting. Thus, the data used in this study comprises both auditors' internal control reports and management-only internal control reports.

³ Although Chan et al. (2008), for example, have found some evidence of accrual-based earnings management in companies with material weaknesses, the current study takes a broader per-

First, we examine whether real earnings management is employed in company years with ineffective internal controls based on subsequent SOX 404 internal control report. That is, material weaknesses exists, but have not yet been disclosed. We particularly scrutinize situations where material weaknesses were disclosed for the first time, so as to explore the impact of the existence of weaknesses without the possibly confounding effect arising from previous material weakness disclosures. Internal control systems are designed and implemented by the management (COSO 1992; SOX 2002), and consequently, weaknesses in internal controls implicitly create doubts about management's competence and its attitude toward financial reporting and/or in extreme cases, even its integrity. Material weaknesses particularly reflect pervasive problems in the control environment, and thus these weaknesses might stem from the 'tone at the top'. As prior research suggests, low quality governance characteristics are associated with the presence of material weaknesses (e.g., Zhang et al. 2007; Hoitash et al. 2009). Recent studies also suggest that weaknesses in internal control over financial reporting have spillover effects to operations (Cheng et al. 2013; Bauer et al. 2014). The existence of material weaknesses could hence be expected to reflect such a business environment, which is permissive to real earnings management. Moreover, given that material weaknesses per se frequently give rise to lower quality internal information (Feng et al. 2015), it is possible that management is (unintentionally) making poor operational decisions based on this information, which manifest in greater real earnings management.

Second, we examine whether companies disclosing previous year's material weaknesses have higher levels of real earnings management. Higher levels of real earnings management could be expected in companies that have recently attracted bad publicity due to material weakness disclosures. Prior research has shown that disclosures of internal control deficiencies are perceived negatively in debt and equity markets (e.g., Ashbaugh-Skaife et al. 2009; Kim et al. 2011). Moreover, Su et al. (2014) report a decline in customer demand following the disclosure of internal control weaknesses. Because of the negative consequences anticipated from the disclosure of material weakness (e.g., negative investor reactions, increased cost of debt, or impact on personal reputation), management could be expected to strive to mitigate these effects and therefore have an incentive for earnings management executed by the manipulation of real operational activities, which is not easily detected or constrained by outsiders. Therefore, inducing a short-term in-

spective on the influence of material weaknesses by examining whether also operational activities fall under the sphere of influence of internal control effectiveness, not only accounting issues.

crease in earnings by real earnings management in an attempt to mitigate the negative reactions to material weakness disclosures might be perceived to outweigh the risk posed by real earnings management being detected.

Using a sample of fiscal year observations of US listed companies from 2004 to 2012, we examine whether real earnings management is greater in companies 1) having material weaknesses for the first time (existence), 2) disclosing material weaknesses for the first time and subsequently disclosing a clean 404 report (disclosure), and 3) disclosing material weaknesses for the first time and continuing to have material weaknesses based on a subsequent internal control report⁴. Our control group consists of companies without a history of material weakness disclosures (i.e., companies that have always had effective internal controls based on 404 reports). The main analyses are also conducted using a propensity-score matched sample. The real earnings management methods investigated are inventory overproduction (proxied by abnormal levels of production costs), reduction of discretionary expenses (proxied by abnormal levels of discretionary expenses), and sales manipulation (proxied by abnormal levels of cash flows from operations) (Roychowdhury 2006). For each method, we use three measures: the basic real earnings management measures introduced by Roychowdhury (2006) and two other measures where the basic measures have been adjusted by performance.⁵

The empirical findings of this study suggest that the existence of material weaknesses and the disclosure of those weaknesses are related to more extensive real earnings management than in companies that have always had effective internal controls. First, companies with existing material weaknesses appear to be using overproduction as a real earnings management method. This implies that, although the association may stem from the behavior of an opportunistic management, the poor 'tone at the top' may also flow through to inventory-related operational decisions via low quality internal information that results from material weaknesses (see Feng et al. 2015). The existence of material weaknesses is also related to the reduction of discretionary expenses, but the significance of that relationship appears to be somewhat smaller. Moreover, we find only modest evi-

⁴ The third group captures both effects: the existence of material weaknesses and the disclosure of material weaknesses. Therefore, the indicator variable for this group is merely used to provide supplemental information.

⁵ The first performance-adjusted measures are calculated by adding lagged return on assets to the estimation models of real earnings management measures. The second performance-adjusted measures are calculated by subtracting the mean values of real earnings management measures of matched control companies based on year, industry, and return on assets (e.g., Kim and Park 2014).

dence on the association between the existence of material weaknesses and sales manipulation.

Second, companies disclosing material weaknesses and subsequently disclosing a clean internal control report also use inventory overproduction and reduction of discretionary expenses as real earnings management methods. In particular, cutting discretionary expenses seems likely to be the most feasible real earnings management method after disclosure. The results do not, however, indicate that material weakness disclosure is associated with sales manipulation. This insignificance might be related to the reduction in customer demand following the disclosure of an internal control weakness (Su et al. 2014). Thus, sales manipulation is not a workable real earnings management method for companies with material weaknesses. Finally, companies disclosing material weaknesses and subsequently continuing to disclose material weaknesses employ inventory overproduction to manage earnings, but there is also some evidence on reduction of discretionary expenses among these companies.

To provide additional evidence on the incentives for real earnings management among material weakness companies, we examine whether the associations between material weakness/material weakness disclosure companies and real earnings management differ based on previous-year performance. These findings primarily indicate that when the sample consists of previously poorly performing companies, the *existence* (but not the disclosure) of material weaknesses is associated with inventory overproduction and reduction of discretionary expenses. That is, although the motivation is similar (previous poor performance), the management of a material weakness company employs real earnings management methods. When the sample consists of previously better performing companies (i.e., the sample companies do not have a clear performance-related incentive to manage earnings), the companies *disclosing* material weaknesses are utilizing overproduction and reduction of discretionary expenses, suggesting that the negative publicity of material weakness disclosures induces management to manipulate real operational activities to manage earnings. There is also, however, some modest evidence of sales manipulation among poor performing companies that disclose material weaknesses, and among better performing companies with existing material weaknesses.

This study contributes to the research on internal control over financial reporting and the research on earnings management. First, while previous studies have found evidence for a relationship between internal control deficiencies and earnings management through accrual-manipulation/accrual quality (e.g., Chan et al. 2008; Doyle et al. 2007a; Ashbaugh-Skaife et al. 2008), the current study investigates another possible method, namely real earnings management. In particular, the findings on the impact of material weaknesses on real earnings management behavior imply the pervasiveness of control issues in companies with material weaknesses. That might reflect a direct tendency to use real earnings management methods or decisions made based on biased internal accounting information (nevertheless caused by management's failure to establish an adequate control system). These findings therefore offer insights into the issue of the pervasive nature of control problems and spillover effects to operations that have been discussed in prior studies (e.g., Skaife et al. 2013; Cheng et al. 2013; Feng et al. 2015; Bauer et al. 2014; Myllymäki 2014).

Second, this study extends the literature on the effects of internal control disclosures (e.g., Ashbaugh-Skaife et al. 2009; Beneish et al. 2009; Kim et al. 2011; Rezee et al. 2012; Clinton et al. 2014; Su et al. 2014) and provides new information about greater real earnings management in companies after a material weakness disclosure, suggesting that incentives for real earnings management exist due to the expected negative effects of such disclosures. Third, this study contributes to the real earnings management literature (e.g., Roychowdhury 2006; Cohen et al. 2008; Cohen and Zarowin 2010; Gunny 2010; Badertscher 2011; Chi et al. 2011; Zang 2012; Kim and Park 2014) by indicating that material weaknesses in internal controls signal an environment where management is more susceptible to real earnings management behavior, which may subsequently incur costs for stockholders by decreasing company value in the longer term. Overall, our study provides further insights on pervasive governance problems that may exist in companies with material weaknesses by documenting how operational activities fall under the sphere of influence of internal control effectiveness.

The remainder of the paper is organized as follows. Section two discusses the relevant prior studies, and reasoning behind the hypotheses. Section three describes the data and methodology. The empirical findings are reported in section four. Section five concludes the paper.

2 Background and development of the hypotheses

Healy and Wahlen (1999) define earnings management in the following way:

"Earnings management occurs when managers use judgment in financial reporting and in structuring transactions to alter financial reports to either mislead some stakeholders about the underlying economic performance of the company or to influence contractual outcomes that depend on reported accounting numbers."

According to the survey results of Graham et al. (2005), managers consider meeting or exceeding financial objectives to be very important, although the ways in which they do so might have a negative effect on the company's future value. The primary incentives for managers to meet short-term objectives are stock prices and career and reputation concerns (Graham et al. 2005). Prior research on earnings management has primarily concentrated on investigating accrual-based earnings management, which relates to the discretion that management employs in applying an accounting method (Zang 2012). In addition to accrual-based earnings management, more recent research has considered whether management might apply earnings management through operational decisions, that is, real activities manipulation (e.g., Graham 2005; Roychowdhury 2006; Badertscher 2011; Zang 2012).

Real activities manipulation relates to departures from normal operating practices (Roychowdhury 2006). According to Roychowdhury (2006), these departures do not necessarily enhance the long-term company value, although they enable the company to meet certain financial reporting targets. That is, the stakeholders are misled into believing that normal business operations led to the achievement of the goals. However, the manipulation of current earnings might have a negative effect on cash flows in the future periods (Roychowdhury 2006). Real earnings management methods available to managers are, for example, acceleration of sales through aggressive price discounts or more lenient credit terms, overproduction, and cutting discretionary expenditure (Roychowdhury 2006). These actions increase earnings in the current period, but most likely result reducing cash flows in the longer term. The recent research on real activities manipulation is primarily motivated by the survey results in Graham et al. (2005), which suggest that the bulk of earnings management results from manipulating real operating activities. Supporting the survey results in Graham et al. (2005), the findings of Roychowdhury (2006), Cohen et al. (2008), and Zang (2012) suggest that managers do engage in real activities manipulation. For example, investigating earnings management in the pre- and post-SOX environment, Cohen et al. (2008) find that the level of real earnings management activities increased significantly after the passage of SOX in 2002, whereas accrual-based earnings management declined. The use of real earnings management is related to the assumption that manipulating real operating activities is harder for company outsiders to detect or constrain (see the discussion in, for example, Graham et al. 2005; Roychowdhury 2006; Cohen et al. 2008).

2.1 The effect of existence: real earnings management before disclosure of material weakness

SOX section 404 requires management to report annually on its assessment of the effectiveness of internal controls over financial reporting, and to disclose any material weaknesses.^{6,7} The internal controls over financial reporting aim to provide a reasonable assurance that financial information is reliable (PCAOB 2007). Accordingly, prior research has indicated that an effective internal control system is associated with higher quality earnings (Doyle et al. 2007a; Chan et al. 2008; Ashbaugh-Skaife et al. 2008).⁸ Chan et al. (2008) concentrate on the association between internal control weaknesses and accrual-based earnings management (proxied by discretionary accruals), and document some evidence of accrual-based earnings management among companies reporting material weaknesses.

Although internal controls over financial reporting are not specifically designed to constrain real activities manipulation, in as far as internal control effectiveness reflects management's attitude toward financial reporting quality, by implication, it could be linked with real earnings management behavior. The effectiveness of internal controls is an outcome of management decision making, since management is responsible for designing, implementing, monitoring, and improving internal control systems (SOX 2002). Thus, weaknesses in internal controls result from a lack of commitment to establish or maintain appropriate internal controls. Such weaknesses then implicitly cast doubt on the management's competence, and/or attitude toward financial reporting and internal control systems. In the extreme cases, material weaknesses may reflect an absence of integrity among the management. Prior research has demonstrated that effective internal control systems are associated with different governance characteristics: audit committee characteristics (e.g., Zhang et al. 2007; Hoitash et al. 2009), CFO characteristics (Li et al. 2010), and board strength (e.g., Hoitash et al. 2009). Thus, companies with internal control weaknesses are exposed to greater governance risk.

⁶ Material weakness is defined as "a significant deficiency, or combination of significant deficiencies, that results in more than a remote likelihood that a material misstatement of the annual or interim financial statements will not be prevented or detected" (PCAOB 2007).

⁷ SOX contains also Section 302, which relates to quarterly reporting and requires management to assess the effectiveness of disclosure controls and procedures. The quarterly certification should indicate that the management has evaluated the effectiveness of internal controls as well as any significant changes in internal controls.

⁸ Ineffective internal controls have the potential to allow both unintentional errors in accounting information and intentional earnings management (Doyle et al. 2007a; Ashbaugh-Skaife et al. 2008).

If material weaknesses signal an environment that promotes actions and policies supporting management's self-interest (see the discussion in Skaife et al. 2013), management of material weakness companies might intentionally apply real earnings management to boost earnings and to gain private benefits. Recent literature suggests that, in addition to the direct link to accounting quality, material weaknesses have spillover effects on operations as well (Cheng et al. 2013; Bauer et al. 2014). Moreover, if internal control weaknesses *per se* reduce the quality of *internal* information (Feng et al. 2015), it is possible that management is unintentionally making poor operational decisions based on inaccurate information, which manifest in real earnings management. Feng et al. (2015), for example, find that inventory-related weaknesses are associated with poor inventory management.

Based on the discussion above, we propose the following hypothesis:

H1: Companies with first-time internal control material weaknesses have higher levels of real earnings management than companies that have always had effective internal controls.

In line with the hypothesis above, we concentrate on *first-time* material weaknesses, because we wish to capture the effect of the *existence* of internal control material weaknesses in particular. That is, we use the company years before the actual disclosure of 404 material weaknesses, in order to investigate whether real earnings management decisions are related to the existence of material weaknesses rather than the disclosure of those weaknesses.

2.2 The effect of disclosure: real earnings management after disclosure of material weakness

Company stakeholders may perceive material weaknesses in internal control over financial reporting to be an indicator of unreliable financial reporting, and therefore as increasing information risk. Previous studies generally suggest that equity markets react negatively to disclosures of internal control deficiencies (Schneider et al. 2009). For example, Ashbaugh-Skaife et al. (2009) document adverse reactions to the disclosures of internal control deficiencies, and although Beneish et al. (2009) suggested that investors react negatively only to section 302 disclosure reports (see footnote 7), Rezee et al. (2012) found negative stock market reactions to section 404 disclosures. Moreover, previous research has found evidence of the high cost of debt for companies with internal control deficiencies (e.g., Kim et al. 2011), and less financial analysts' coverage after internal control weaknesses had been disclosed (Clinton et al. 2014). Experimental studies have also found, for example, that adverse internal control opinions have an impact on loan officers' assessments of risk (Schneider and Church 2008), analysts' risk assessments (Shelton and Whittington 2008), and investors' decision making (Lopez et al. 2009). Finally, Su et al. (2014) illustrate a decline in customer demand after the disclosure of internal control weakness. In summary, managers probably expect negative reactions from various stakeholders and negative economic consequences to arise from material weakness disclosures. Considering that corporate executives pay a great deal of attention to stock prices, and their personal and company reputation (Graham et al. 2005), managers of companies disclosing material weaknesses could be expected to have an incentive for upward earnings management to mitigate the negative consequences involved (e.g., lowered stock-based compensation, career concerns, increased risk of being the target of a hostile takeover).

The management of companies with recent disclosures of material weaknesses might expect to attract the rigorous attention of auditors and regulators, making accruals-based earnings management seem risky. Real earnings management does not violate GAAP and therefore is not something that external auditors, for example, would necessarily constrain or could even detect.⁹ Moreover, the opaque nature of real earnings management is likely to mislead investors (Kothari et al. 2012). Therefore, managers might consider enhancing financial performance by real earnings management in cases when the companies have received bad publicity from material weakness disclosure outweighs the risk of that real earnings management being detected. Based on the above indications of management incentives to manipulate earnings, this study examines whether companies disclosing material weaknesses have higher levels of real earnings management compared to companies without a history of material weaknesses. Accordingly, we propose the following hypothesis:

H2: Companies disclosing previous year internal control material weaknesses for the first time have higher levels of real earnings management than companies that have always had effective internal controls.

⁹ Kim and Park (2014) do suggest that auditors drop clients with aggressive real activities manipulation. Thus, in order to avoid excessive risk, auditors appear to be concerned about real earnings management. Whether auditors would actually constrain real earnings management is, however, another issue. Chi et al. (2011) find that companies that have higher quality external auditors employ real earnings management, since these auditors are constraining accrual earnings management.

3 Data and methodology

3.1 Sample description

The data used in this study consist of company-year observations of listed companies located in the U.S., covering the years 2004–2012. The data on internal control over financial reporting and external auditors are obtained from the Audit Analytics database, whereas the financial statement data are obtained from the Thomson Financial database.

The initial sample of internal control opinions from Audit Analytics consists of 54,798 company-year observations (11,491 companies). Then, in order to examine the hypotheses, we identify the following groups:

- MWEXIST to test Hypothesis 1: company-year observations of existing material weaknesses for the first time based on subsequent Section 404 internal control reports.
- MWDISCL_{CLEAN} to test Hypothesis 2: company-year observations of firsttime material weakness disclosures, where companies subsequently disclosed a clean report. Because our second hypothesis relates to the impact of the disclosure in particular, we exclude those MWDISCL_{CLEAN} observations where the previous year's material weaknesses were disclosed only in the restated internal control report.
- MWDISCL_{REPEATED} to provide supplemental information: company-year observations of first-time material weakness disclosures, where companies continue to have material weaknesses.
- EIC: effective internal controls—observations of companies that have always disclosed a clean internal control report. Because the dataset consists of both auditors' internal control reports (starting from 2004) and management-only internal control reports (mostly starting from 2007), we use two criteria to construct the EIC group: 1) observations of companies that have disclosed a clean internal control report each year 2004–2012, and 2) observations of companies that have disclosed a clean internal control report each year 2007–2012 and no internal control reports before 2007. The second criterion is applied to capture smaller companies that began to disclose internal control reports only after 2007 in the control sample. Moreover, we exclude company-year observations with disclosed weaknesses based on section 302 requirements (see footnote 7).

After identifying these groups, the sample consists of 23,409 company-year observations: 4,267 observations in the MWEXIST group; 1,081 observations in the MWDISCL_{CLEAN} group; 1,858 observations in the MWDISCL_{REPEATED} group; and 16,203 observations in the EIC group.

The initial sample from Thomson Financial consists of financial data for 9,669 companies, and after excluding banks, insurance companies, and other financial companies (SIC codes 6XXX), and the observations with missing values on variables required to estimate real earnings management, the sample consists of 20,498 company-year observations (3,340 companies).¹⁰

After combining the Thomson Financial data with the Audit Analytics' data, and after excluding the observations with missing values on control variables, the final sample consists of 5,439 company-year observations: 532 observations in the MWEXIST group; 259 observations in the MWDISCL_{CLEAN} group; 194 observations in the MWDISCL_{REPEATED} group; and 4,454 observations in the EIC group. The number of sample observations by year, and by group, is presented in Table 1.

Year	MWEXIST	MWDISCL CLEAN	MWDISCL REPEATED	EIC	Total
2004	93	0	0	379	472
2005	85	47	37	378	547
2006	57	60	29	365	511
2007	95	35	9	517	656
2008	73	51	44	553	721
2009	36	28	36	556	656
2010	29	19	12	567	627
2011	31	6	17	579	633
2012	33	13	10	560	616
Total	532	259	194	4,454	5,439
N7 . C .	1. 1.0 . 11 1.0	er 1.1			

Table 1.Sample by year and by group

Notes: See Appendix 1 for variable definitions.

¹⁰ This sample is used to estimate the models of production costs, discretionary expenditure, and operating cash flows, and further, to determine measures for real earnings management (see the model specification section for details).

3.2 Propensity-score-matched sample

Companies with material weaknesses usually possess certain company characteristics (e.g., poor financial performance, complexity, and small size) (e.g., Ge and McVay 2005; Ashbaugh-Skaife et al. 2007; Doyle et al. 2007b) that could themselves incentivize earnings management. Therefore, we conduct analyses using a propensity-score-matched sample as well. This approach aims to match each material weakness company to a control company that is similar along observable dimensions other than the treatment effect. It is suggested that the advantage of this approach is its ability to better control for determinants of material weakness companies because it disregards functional form for confounding effects that inherently exist in cross-sectional regression analyses (Skaife et al. 2013). We match companies belonging to the MWEXIST, MWDISCL_{CLEAN}, and MWDIS-CL_{REPEATED} groups with EIC companies based on the predicted probabilities derived from the following probit regression:

$$\begin{split} MW &= \alpha + \beta_1 LOGMC + \beta_2 LOGAGE + \beta_3 FOREIGN + \beta_4 SQRTSEGMENTS + \\ \beta_5 RESTRUCTURE + \beta_6 ACQUISITION + \beta_7 INV + \beta_8 LOSS + \beta_9 ROA + \\ \beta_{10} LEV_{t-1} + \beta_{11}GC + \beta_{12} ZSCORE + \beta_{13} HIGHGROWTH + \beta_{14} BIG4 + \\ \beta_{15} AUDITORCHANGE + \beta_{16} LITIGATION + \beta_{17} ICREPORTM + annual \\ fixed effects + industry fixed effects + \varepsilon, \end{split}$$

(1)

where *MW* is assigned a value of one if it is either MWEXIST, MWDISCL_{CLEAN}, or MWDISCL_{REPEATED}, and zero if it is an EIC observation. Model (1) includes independent variables of measures for company size (*LOGMC*), age (*LOGAGE*), complexity (*FOREIGN*, *SQRTSEGMENTS*, *RESTRUCTURE*, *ACQUISITION*, *INV*), financial performance (*LOSS*, *ROA*), debt (*LEVt-1*), bankruptcy risk (*GC*, *ZSCORE*), extreme growth (*HIGHGROWTH*), auditor (*BIG4*, *AUDI-TORCHANGE*), litigation risk (*LITIGATION*), management-only internal control reports (*ICREPORTM*). Moreover, Model (1) includes year and industry (two-digit SIC codes) dummy variables. See Appendix 1 for the variable definitions.

Before the propensity-score matching, the sample consisted of 4,971 observations (after excluding observations with missing data on the independent variables). The estimation results of the probit regression are presented in column (1) of Appendix 2. The propensity-score-matched sample used in the analyses comprises 1,432 company-year observations.¹¹ The same propensity-score-matching proce-

¹¹ After the propensity score matching procedure, the sample size at the first stage is 1,710 observations. However, to tighten matching criteria even further, we include to the matched sample only those observations with maximum of ten percent difference in propensity scores between

dure was applied for the reduced sample where observations with missing values for the performance-matched real earnings management variables (see the model specification section for details) are dropped. After the matching, the propensityscore-matched sample for the analyses with performance-matched real earnings management variables comprises 748 observations.

3.3 Model specifications

Following Roychowdhury (2006), we investigate the following real earnings management methods: increasing earnings by reducing the cost of goods sold by overproducing inventory (proxied by the abnormal level of production costs); reducing discretionary expenditure (proxied by the abnormal level of discretionary expenditures); and sales manipulation (proxied by the abnormal level of cash flow from operations). Overproduction of inventories is considered real earnings management when producing an excessive volume of units is intended to reduce fixed costs per unit by spreading fixed overhead costs over a larger number of units (Cohen et al. 2008). Thus, overproduction can reduce sales costs in a current period and therefore increase earnings, if greater inventory holding costs or marginal cost per unit are not offsetting realized cost savings (Gunny 2010). However, in subsequent periods high inventory holding costs caused by excess of inventories are negatively affecting cash flows. The normal level of production costs is measured using the following model:

$$\frac{PROD_{t}}{TA_{t-1}} = \alpha + \beta_{1} \frac{1}{TA_{t-1}} + \beta_{2} \frac{S_{t}}{TA_{t-1}} + \beta_{3} \frac{\Delta S_{t}}{TA_{t-1}} + \beta_{4} \frac{\Delta S_{t-1}}{TA_{t-1}} + \varepsilon,$$
(2)

where $PROD^{12}$ is the sum of the cost of goods sold in year *t* and the change in inventory from *t*-1 to *t*, *S* is net sales, and *TA* is total assets.

treatment observation and control observation (84 percent of the 1,432 observations have a maximum of one percent difference in propensity scores). In the propensity score matched sample, the mean (median) difference in propensity scores is 0.0030 (0.0001). Moreover, a rerun of Model (1) using the propensity score matched sample shows that the global hypothesis that all of the explanatory variables are zero cannot be rejected, suggesting that the matching was successful (chi-square 18.29, Prob > Chi-Square 1.000). These results are presented in column (2) of Appendix 2.

¹² Following Roychowdhury (2006), we use production costs instead of cost of goods sold because of two reasons: Using cost of goods sold may also measure simultaneously accrual manipulation via inventory account, and it could be distorted by company's FIFO/LIFO choice.

Reducing discretionary expenditure refers to managers trying to decrease expenses such as advertising expense, research and development, and selling, general, and administrative (SG&A) expenses to achieve various income objectives for the current period (Cohen et al. 2008). While the reduction of those costs can boost current period earnings, the action is likely to have a negative effect on cash flow in the future and is therefore not an optimal long-term operational decision. For instance, delaying the start of a new project or shutting down a research site can weaken a company's competitive advantage in the markets (Graham et al. 2005). The normal level of discretionary expenditure is measured using the following model:

$$\frac{DISX_{t}}{TA_{t-1}} = \alpha + \beta_{1} \frac{1}{TA_{t-1}} + \beta_{2} \frac{S_{t-1}}{TA_{t-1}} + \varepsilon,$$
(3)

where *DISX* is the discretionary expenditure (i.e., the sum of R&D expenditure, advertising, and SG&A expenditure¹³) in year t.

Sales manipulation may be carried out using various techniques enabling managers to increase sales during the current year in an effort to increase reported earnings. Offering customers abnormal discounts or lenient credit terms can increase sales volume temporarily and pull sales from the next fiscal year into the current year, but is also likely to result in reduced cash flows overall and sacrifice future earnings (Cohen et al. 2008; Gunny 2010). Furthermore, re-establishing former prices and payment terms can adversely affect demand if customers defer purchases in anticipation of more favorable terms reappearing (Roychowdhury 2006; Gunny 2010). The normal level of cash flow from operations is measured using the following model:

$$\frac{OCF_{t}}{TA_{t-1}} = \alpha + \beta_{1} \frac{1}{TA_{t-1}} + \beta_{2} \frac{S_{t}}{TA_{t-1}} + \beta_{3} \frac{\Delta S_{t}}{TA_{t-1}} + \varepsilon,$$
(4)

where OCF is cash flow from operations in year t.

¹³ If advertising or R&D are missing, they are set to zero (see Roychowdhury 2006).

Models (2), (3), and (4) are estimated for industry-years with at least 15 observations, and the industry categorization is based on two-digit SIC industry groups. The estimated residuals from Model (2) represent the abnormal level of production costs (AB_PROD). Higher residuals indicate greater inventory overproduction, which reduces the cost of goods sold and increases earnings. The estimated residuals from Model (3) represent the abnormal level of discretionary expenditure (AB_DISX), with lower values indicating greater real earnings management to increase earnings. The estimated residuals from Model (4) represent the abnormal level of operating cash flows (AB_OCF), with lower values indicating greater real earnings management to increase earnings. Observations with extreme values are winsorized at the top and bottom one percent.

Following Kothari et al. (2005), who demonstrate an improvement in abnormal accruals estimations after taking performance into account, recent studies (e.g., Cohen et al. 2014; Kim and Park 2014) suggest that real earnings management estimations also include potential performance-related misspecification. Therefore, we construct two additional real earnings management variables adjusted by performance (return on assets). First, we estimate the above Models (2)-(4) after including the previous year's return on assets (ROA_{t-1}) in the models. The real earnings management variables from these estimations are labeled AB_PROD_LAGROAadded. AB_DISX_LAGROAadded, and AB OCF LAGROAadded. Second, we follow Kim and Park (2014) in employing the performance-matching procedure. We subtract the mean values of real earnings management measures of matched control companies based on the following criteria: fiscal year, two-digit SIC code and ROA between +/- 50 percent. The sample observation is dropped if there is no match between the ROA ranges. These real earnings management variables are labeled AB_PROD_ROAmatched, AB DISX ROAmatched, and AB OCF ROAmatched.

To test Hypotheses 1 and 2, we estimate the following OLS regression model:

$$\begin{split} RM &= \alpha + \beta_1 MWEXIST + \beta_2 MWDISCL_{CLEAN} + \beta_3 MWDISCL_{REPEATED} + \beta_4 LOGMC \\ &+ \beta_5 LEV_{t-1} + \beta_6 PB_{t-1} + \beta_7 \varDelta INCOME_{t-1} + \beta_8 ROA_{t-1} + \beta_9 BIG4 + \beta_{10} DACC + \\ & annual fixed effects + industry fixed effects + \varepsilon, \end{split}$$

(5)

where *RM* is either *AB_PROD*, *AB_DISX*, or *AB_OCF* (or their corresponding performance-adjusted measures). Congruent with groups described in the sample description section, the variables of interest are the following: *MWEXIST* is a dummy variable set to one, if the company year includes first-time material weaknesses in internal controls over financial reporting according to the 404 report, and zero otherwise; *MWDISCL*_{CLEAN} is a dummy variable assigned a value

of one if the previous company year included first-time material weaknesses in internal controls, and the subsequent report was a clean report, and zero otherwise; $MWDISCL_{REPEATED}$ is a dummy variable equal to one, if the previous company year included first-time material weaknesses in internal controls according to the 404 report, and the subsequent report still indicated material weaknesses, and zero otherwise; EIC group of companies without a history of 404 material weaknesses serves as the control group.

Model (5) includes control variables for various company characteristics expected to influence real earnings management. The following six control variables were adapted from Chi et al. (2011). We control for company size by including the natural logarithm of market capitalization (*LOGMC*) in the model. To control for the effect of indebtedness on real earnings management, we include the prior fiscal year's leverage (*LEV*) in the model. We add the previous year's price-to-book ratio (*PB*) into the model in order to control for growth opportunities. We control for firm performance by adding the previous year's change in net income before extraordinary items ($\Delta INCOME$) and return on assets (*ROA*) to the model. To control for the effect of the size of the audit firm on its client companies' real earnings management, we include an indicator variable for the four largest audit firms (*BIG4*).

Moreover, as Zang (2012) suggest that accrual-based earnings management is associated with real activities manipulation, we add discretionary accruals (*DACC*) to the model as a control variable.¹⁴ Year and industry (according to the two-digit SIC industry groups) fixed effects control for temporal variation and industry differences in the real earnings management. The variables are defined in Appendix 1.¹⁵

3.4 Descriptive statistics

Panel A of Table 2 displays the descriptive statistics of the variables used in the empirical analyses. The means of the real earnings management variables (*LAGROAadded*, and *ROAmatched* in parentheses, respectively) are: *AB_PROD* -

¹⁴ There may not be direct causal relationship between these variables, as Zang (2012) suggests that these two earnings management methods are sequential decisions, i.e. real earnings management (executed during the fiscal year) preceding accrual-based earnings management (executed after the fiscal-year end to "fine-tune the manipulation"). However, we include discretionary accruals to the model to control for accounting earnings management behavior.

¹⁵ The continuous control variables are winsorized at the top and bottom one percent to mitigate outliers.

0.013 (-0.007, -0.026), AB_DISX -0.007 (0.032, 0.005), and AB_OCF 0.033 (0.007, 0.013). The median values of the real earnings management variables (*LAGROAadded*, and *ROAmatched* in parentheses, respectively) are: AB_PROD - 0.009 (-0.001, -0.023), AB_DISX -0.067 (-0.032, -0.015), and AB_OCF 0.048 (0.021, 0.018). According to Cohen et al. (2014): "... in random samples where firms are selected without regard to any hypothesis or prediction about managerial incentives to manage real activities, the expected value of each REM measure is zero." Against this backdrop, our inferences from the analyses are based on those models where the real earnings management measure contains less bias (i.e., the mean and median are closer to zero). Our distributional statistics regarding the means (and medians) of real earnings management measures indicate that *LAGROAadded* variables are the most well-specified measures for abnormal production costs and abnormal operating cash flows¹⁶, while *ROAmatched* is the most well-specified measure for abnormal discretionary expenses.

Market capitalization ranges from 2.486 million to 113 billion dollars, with an average (median) value of 6,136 (819) million dollars. The mean (median) LEV_{t-1} is 0.196 (0.180), PB_{t-1} is 3.479 (2.265), $\Delta INCOME_{t-1}$ is 0.017 (0.010), and ROA_{t-1} is 0.010 (0.050). In addition, 76 percent of the company years are audited by one of the four largest audit firms. The mean (median) *DACC* is -0.069 (0.002).

Panel B of Table 2 displays the Pearson correlation coefficients across the independent variables. Overall, correlations between the variables are modest and the highest correlation (0.629) occurs between the indicator variable for a Big 4 audit firm (*BIG4*) and market capitalization (*LOGMC*). Untabulated variance inflation factors do not indicate problems with multicollinearity.

¹⁶ The median of the performance-matched measure of abnormal operating cash flows is slightly closer to zero com-pared to *LAGROAadded* measure. However, since the difference in mean values between these two measures is somewhat larger, with the mean of *LAGROAadded* being closer to zero, the inferences from the analyses are primarily based on results where *AB_OCF_LAGROAadded* is the dependent variable.

Table 2.Descriptive statistics

Variable	Mean	Std.	Min	25%tile	Median	75%tile	Max	Ν
AB_PROD	-0.013	0.228	-0.936	-0.120	-0.009	0.094	0.910	5,439
AB_PROD LAGROAadded	-0.007	0.224	-0.919	-0.107	-0.001	0.095	0.845	5,439
AB_PROD ROAmatched	-0.026	0.229	-1.151	-0.134	-0.023	0.079	1.330	3,839
AB_DISX	-0.007	0.646	-1.771	-0.196	-0.067	0.045	5.378	5,439
AB_DISX LAGROAadded	0.032	0.561	-1.473	-0.136	-0.032	0.078	4.893	5,439
AB_DISX ROAmatched	0.005	0.404	-6.352	-0.100	-0.015	0.084	5.708	3,839
AB_OCF	0.033	0.246	-1.830	-0.027	0.048	0.139	1.006	5,439
AB_OCF LAGROAadded	0.007	0.206	-1.523	-0.037	0.021	0.087	0.954	5,439
AB_OCF ROAmatched	0.013	0.173	-1.396	-0.037	0.018	0.073	2.661	3,839
MWEXIST	0.098	0.297	0.000	0.000	0.000	0.000	1.000	5,439
MWDISCL _{CLEAN}	0.048	0.213	0.000	0.000	0.000	0.000	1.000	5,439
MWDISCL _{REPEATED}	0.036	0.185	0.000	0.000	0.000	0.000	1.000	5,439
EIC	0.819	0.385	0.000	1.000	1.000	1.000	1.000	5,439
Market cap (millions)	6136.140	16616.390	2.486	161.788	819.321	3456.080	113240.000	5,439
LEV _{t-1}	0.196	0.169	0.000	0.028	0.180	0.307	0.649	5,439
PB _{t-1}	3.479	4.458	0.350	1.415	2.265	3.759	34.282	5,439
$\Delta INCOME_{t-1}$	0.017	0.155	-0.554	-0.020	0.010	0.042	0.745	5,439
ROA _{t-1}	0.010	0.181	-0.902	-0.004	0.050	0.093	0.335	5,439
BIG4	0.758	0.428	0.000	1.000	1.000	1.000	1.000	5,439
DACC	-0.069	3.244	-17.264	-0.210	0.002	0.290	13.966	5,439

Panel A: Distributional statistics

(
	2	3	4	5	6	7	8	9	10
MWEXIST (1)	-0.074	-0.063	-0.206	-0.022	0.033	0.038	-0.123	-0.139	-0.001
MWDISCL _{CLEAN} (2)		-0.043	-0.122	-0.016	0.016	-0.054	-0.094	-0.083	0.018
MWDISCL REPEATED(3)			-0.179	-0.007	0.053	-0.022	-0.124	-0.155	0.030
LOGMC (4)				0.174	0.033	0.000	0.388	0.629	-0.007
$LEV_{t-1}(5)$					0.108	-0.042	-0.022	0.168	0.032
$PB_{t-1}(6)$						0.021	-0.221	-0.074	0.032
$\Delta INCOME_{t-1}(7)$							0.354	0.010	0.004
$ROA_{t-1}(8)$								0.261	-0.009
BIG4 (9)									-0.012
DACC (10)									

Panel B: Pearson's correlation coefficients among independent variables (N=5.439)

Notes: See Appendix 1 for variable definitions. Bold text in the Pearson correlation matrix indicates significance at the 0.05 level.

In Table 3, we present the mean differences of the real earnings management measures between MW companies and EIC companies. The statistical significances are based on t-tests. In general, these univariate tests imply that MWEX-IST companies, MWDISCL_{CLEAN} companies, and MWDISCL_{REPEATED} companies, have higher levels of real earnings management than EIC companies. Specifically, across different real earnings management measures, abnormal production costs appear to be consistently higher in MW companies than in EIC companies. Abnormal operating cash flows are significantly smaller in MW companies, except when using the performance-matched measure. Performance-matched abnormal discretionary expenses are significantly smaller for companies with existing material weaknesses. Companies disclosing material weaknesses and subsequently disclosing a clean report have also significantly smaller discretionary expenses, but only when using the measure estimated from the models with lagged ROA added. However, we make our inferences based on the multivariate analyses reported in the following section.

	MWE	UST	MWDISCI		MWDISCL	EDEATED	Control	group:
	Mean	n	Mean	n	Mean	n	EI Mean	C n
AB_PROD	0.031 ***	532	0.038 ***	259	0.067 ***	194	-0.025	4,454
AB_PROD LAGROAadded	0.032 ***	532	0.043 ***	259	0.072 ***	194	-0.018	4,454
AB_PROD ROAmatched	0.004 **	271	0.026 **	136	-0.004	81	-0.031	3,351
AB_DISX	-0.027	532	-0.013	259	0.044	194	-0.007	4,454
AB_DISX LAGROAadded	0.013	532	-0.021 **	259	0.021	194	0.038	4,454
AB_DISX ROAmatched	-0.038 *	271	-0.044	136	-0.037	81	0.011	3,351
AB_OCF	-0.031 ***	532	-0.025 ***	259	-0.027 ***	194	0.047	4,454
AB_OCF LAGROAadded	-0.052 ***	532	-0.032 ***	259	-0.013 *	194	0.018	4,454
AB_OCF ROAmatched	-0.002	271	0.000	136	0.022	81	0.015	3,351

Table 3.Univariate results

Notes: See Appendix 1 for variable definitions. Statistical significance are based on t-tests.

4 Results

We estimate Model (5) to test whether *MWEXIST*, *MWDISCL*_{CLEAN}, and/or *MWDISCL*_{REPEATED} are associated with real activities manipulation. The OLS regression results (the statistical significances calculated by clustering the standard errors within companies) are reported in Table 4: regressions of abnormal production costs, abnormal discretionary expenditure, and abnormal operating cash flows in Panel A, Panel B, and Panel C of Table 4, respectively. In each panel, columns (1)-(3) report the results when using the total sample, while columns (4)-(6) report the results when using the propensity-score-matched sample.

4.1 Abnormal production costs

The results in Panel A of Table 4 indicate that the coefficient on *MWEXIST* is positive and significant at the 0.05 level when the dependent variable is either *AB_PROD* or *AB_PROD_LAGROAadded*. The findings are similar for the propensity-score-matched sample. When using the performance-matched measure of abnormal production costs (*AB_PROD_ROAmatched*), *MWEXIST* is insignificant. However, as explained in the descriptive statistics section, our distributional statistics suggest that the most well-specified measure for abnormal production costs is that calculated by adding the lagged ROA (*AB_PROD_LAGROAadded*) to the estimation models [Model (2)]. It should also be noted that using the propensity-score-matched sample should control for differences in performance, among oth-

ers. We therefore infer that the existence of material weaknesses is associated with inventory overproduction.

In the analyses on the total sample, the coefficients on $MWDISCL_{CLEAN}$ indicate a significant and positive association with AB_PROD (at the 0.10 level) and $AB_PROD_LAGROAadded$ (at the 0.05 level). These associations are similar when using the propensity-score-matched sample (a positive association with AB_PROD significant at the 0.05 level and with $AB_PROD_LAGROAadded$ significant at the 0.05 level and with $AB_PROD_LAGROAadded$ significant at the 0.01 level). Following the rationale outlined above, we infer that companies disclosing material weaknesses have greater inventory overproduction, although the performance-matched measure is not significantly associated with $MWDISCL_{CLEAN}$.

When using the total sample (propensity-score-matched sample) $MWDISCL_{RE-PEATED}$ is positive and significant at the 0.01 level (at the 0.05 level) when the dependent variable is AB_PROD , and positive and significant at the 0.01 level when the dependent variable is $AB_PROD_LAGROAadded$. Thus, we infer that companies disclosing material weaknesses and maintaining material weaknesses (as confirmed by a subsequent internal control report) have significantly more inventory overproduction compared to companies with effective internal controls.¹⁷ Again, the association is insignificant when using the performance-matched measure as a dependent variable [columns (3) and (6)].

¹⁷ In column (2) where the dependent variable is AB_PROD_LAGROAadded, the coefficients on MWEXIST, MWDISCL_{CLEAN}, and MWDISCL_{REPEATED} indicate 3, 3.7, and 7.1 percentage points higher abnormal production costs compared to the control group, respectively.

anel A: Abnorma	al prc	duction	n costs															
			(1)			(2)			(3)		(4)			(5)			(9)	
		Total	l sample	e	Tota	l sample		Total	sample	Ma	tched a	sample	[Matc]	hed sam	ple	Matche	ed samp	le
spendent variable		AB_{-}	PROD	_	AB	PROD	pe	AB_ ROA	PROD matched	1	AB_PR	OD	AB LAG	PROI ROAad) ded	ROA	PROD matched	
	Exp. sign	Est.	<u>t</u>		Est.	t		Est.	t	Est			Est.	<u>1</u>		Est.	t	
IWEXIST) +	0.032	2.26	* *	0.030	2.14	*	0.010	0.52	0.0	38 2.	01 **	0.038	2.08	* *	-0.007	-0.33	
IWDISCL _{CLEAN}	+	0.036	1.95	*	0.037	2.11	*	0.039	1.48	0.0	57 2.	45 **	0.059	2.72	* * *	0.045	1.56	
1WDISCL _{REPEATED}	+	0.066	3.15	* * *	0.071	3.38	***	0.008	0.29	0.0	57 2.	25 **	0.069	2.65	* * *	-0.007	-0.19	
OGMC	ı	-0.009	-2.42	*	-0.006	-1.77	*	-0.007	-1.52	-0.0-	16 -2.	*** 68	-0.013	-2.36	* *	-0.010	-1.65 *	*
EV _{t-1}	+	0.199	5.22	* * *	0.196	5.27	***	0.113	2.27 **	0.23	24 4.	18 ***	0.194	3.75	* * *	0.142	2.11 *	*
B_{t-1}	ı	-0.010	-6.34	* **	-0.011	-7.05	* *	-0.011	-4.88 **	.* -0.0(06 -2.	34 **	-0.007	-2.64	* * *	-0.017	-4.03	* *
INCOME _{t-1}	ı	-0.000	-2.06	*	-0.000	-2.08	*	-0.000	-0.54	0.0	26 0.	57	0.074	1.49		-0.046	-0.59	
OA_{t-1}	ı	-0.125	-3.73	* * *	-0.101	-3.03	* *	-0.094	-1.30	-0.0-	45 -0.	73	-0.041	-0.65		0.214	1.50	
IIG4	ı	-0.005	-0.28		-0.007	-0.39		-0.024	-0.94	0.0	17 0.	76	0.018	0.80		0.036	1.16	
ACC	ċ	0.000	0.00		0.000	0.14		0.002	1.39	0.0(0.00	73	0.003	0.99		-0.001	-0.13	
ntercept		0.069	1.37		0.092	2.01	*	0.135	1.85 *	0.0	93 0.	66	0.101	1.22		-0.053	-0.36	
nnual fixed ffects?		Yes			Yes			Yes		Υ	es		Yes			Yes		
ndustry fixed ffects?		Yes			Yes			Yes		Υ	es		Yes			Yes		
dj. R ²		0.08			0.09			0.06		0.0)5		0.05			0.06		
		5,439			5,439			3,839		1,4,	32		1,432			748		

80 Acta Wasaensia

Multivariate results

Table 4.

4.2 Abnormal discretionary expenses

The results in Panel B of Table 4 indicate that *MWEXIST* is negative and significant at the 0.01 level when the dependent variable is *AB_DISX*, and negative and marginally significant (at the 0.10 level) when the dependent variable is *AB_DISX_ROAmatched*. We draw our inferences from the model with a performance-matched measure as a dependent variable, since the distributional statistics in Panel A Table 2 suggest that *ROAmatched* variable is the most well-specified measure for abnormal discretionary expenses. Moreover, in the propensity score matched sample, *MWEXIST* is marginally significantly associated with *AB_DISX* (at the 0.10 level). We therefore conclude that the existence of material weaknesses has a modest association with the reduction of discretionary expenses.

In the total sample, *MWDISCL_{CLEAN}* is marginally significantly (at the 0.10 level) associated with the performance-matched measure of abnormal discretionary expenses (*AB_DISX_ROAmatched*). When using the propensity-score-matched sample, *MWDISCL_{CLEAN}* is significant at the 0.01 level when the dependent variable is *AB_DISX* or *AB_DISX_LAGROAadded*, and significant at the 0.05 level when the dependent variable is *AB_DISX_ROAmatched*. In sum, we find evidence that companies disclosing material weaknesses are cutting discretionary expenses to manage earnings.

 $MWDISCL_{REPEATED}$ is marginally significant in the total sample when the performance-matched measure of abnormal discretionary expenses $(AB_DISX_ROAmatched)$ is the dependent variable. Moreover, when using the propensity-score-matched sample, $MWDISCL_{REPEATED}$ is significant at the 0.05 level, but only if the dependent variable is $AB_DISX_ROAmatched$. Accordingly, these results offer evidence that companies disclosing material weaknesses and continuing to have material weaknesses are managing earnings by reducing discretionary expenses, albeit that evidence is somewhat modest.¹⁸

¹⁸ In column (3) where the dependent variable is AB_DISX_ROAmatched, the coefficients on MWEXIST, MWDISCL_{CLEAN}, and MWDISCL_{REPEATED} indicate 5.5, 7.1, and 7.9 percentage points lower abnormal discretionary expenses compared to the control group, respectively.

	(3)
	(2)
Panel B: Abnormal discretionary expenses	(1)

			[]			(2)			(3)		(4)			(5)		Ŭ	(9)	
		Tota	ul sampl	e	Totā	ıl sampl	e	Tota	l sample	Matc	hed sam	ple	Match	ed sam	ole	Matche	d samp	ole
Dependent variable		AB	SIQ_8		AB LAGF	E DISX	led	AB ROA	DISX	Α	B_DISX		AB LAGR	DISX	ed	AB_ ROAr	DISX	q
	Exp. sign	Est.	t		Est.	<u>t</u>		Est.	t	Est.	t		Est.	1t		Est.	<u>I</u> t	
MWEXIST) 1	-0.082	-2.81	* **	-0.043	-1.60		-0.055	-1.84 *	-0.072	-1.92	*	-0.044	-1.26	I	0.011	0.41	
MWDISCL _{CLEAN}	ı	-0.042	-1.13		-0.050	-1.55		-0.071	-1.82 *	-0.108	-3.04	* * *	-0.101	-3.05	۱ * *	0.102 .	.1.99	*
MWDISCL _{REPEATED}	ı	-0.030	-0.58		-0.033	-0.70		-0.079	-1.85 *	-0.050	-0.80		-0.067	-1.19	ľ	0.100	2.26	*
LOGMC	+	-0.020	-2.43	* *	-0.013	-1.95	*	-0.024	-2.06 **	-0.006	-0.49		-0.010	-0.86		0.000	0.03	
LEV_{t-1}	ı	-0.257	-4.14	* * *	-0.258	-4.75	* * *	-0.152	-2.29 **	-0.193	-1.74	*	-0.224	-2.21	۱ * *	0.192 .	2.19	*
PB _{t-1}	+	0.022	5.30	* * *	0.018	5.48	* * *	0.019	3.84 **	** 0.014	2.56	* *	0.014	3.05	* * *	0.020	3.04	* * *
ΔINCOME _{t-1}	+	0.002	0.02		0.091	1.04		-0.220	-1.46	0.124	0.86		0.150	1.14	ı	0.077	0.50	
ROA_{t-1}	+	-0.307	-2.76	* * *	0.065	0.67		0.164	0.70	-0.443	-2.97	* * *	0.030	0.24	1	0.298 -	.1.34	
BIG4	+	0.007	0.19		-0.005	-0.17		0.060	1.47	0.022	0.48		0.018	0.42	1	0.076	2.08	*
DACC	ċ	-0.021	-4.53	* * *	-0.022	-5.10	* * *	-0.002	-0.69	-0.023	-2.25	* *	-0.023	-2.34	* *	0.007	0.89	
Intercept		-0.275	-2.76	* * *	-0.102	-1.08		-0.113	-0.69	-0.394	-2.52	* *	-0.141	-0.98		0.003	0.02	
Annual fixed effects?		Yes			Yes			Yes		Yes			Yes			Yes		
Industry fixed effects?		Yes			Yes			Yes		Yes			Yes			Yes		
Adj. R ²		0.15			0.16			0,04		0.14			0.13			0.03		
Ν		5,439			5,439			3,839		1,432			1,432			748		

4.3 Abnormal operating cash flows

The results in Panel C of Table 4 do not show any consistent evidence of an association between abnormal operating cash flows and material weaknesses. That is, *MWEXIST* is negative and significant at the 0.05 level (at the 0.10 level) when using the total sample (the propensity-score-matched sample) and the dependent variable is $AB_OCF_LAGROAadded$ (which is the most well-specified measure of abnormal operating cash flows based on distributional statistics). Thus, we conclude that there is some evidence of sales manipulation when companies have material weaknesses. However, it appears that companies disclosing material weaknesses are not employing sales manipulation to manage earnings. When the dependent variable is $AB_OCF_ROAmatched$, $MWDISCL_{REPEATED}$ is positive and marginally significant in columns (3 and (6), but the statistical power of these models is almost non-existent.¹⁹

¹⁹ In column (2) where the dependent variable is AB_OCF_LAGROAadded, the coefficient on MWEXIST indicate 2.8 percentage points lower abnormal operating cash flows compared to the control group.
Panel C: Abnorma	l opera	ting ca	sh flow	S											
			(1)			(2)			(3)		(4)		(5)		(9)
		Toti	al sample	0	Tota	samp	le	Tota	l sample	[Matc]	hed sampl	e Mat	ched sample	Matche	d sample
Dependent variable		Ŋ	B_OCF		AE LAGR	OCF	ded	AE ROA	OCF matched	A	B_OCF	LAC	AB_OCF GROAadded	ROAr	OCF
	Exp. sign	Est.	t		Est.	lt.		Est.	1¢	Est.	<u>t</u>	<u>Est.</u>	۱ ۱	Est.	I¢
MWEXIST	ı	-0.020	-1.63		-0.028	2.56	* *	-0.001	-0.09	-0.015	- 1.01	-0.02	4 * 1.83	-0.008	-0.57
MWDISCL _{CLEAN}	ı	-0.024	-1.53		-0.023	- 1.63		0.000	0.00	-0.004	- 0.24	-0.00	3 0.23	0.013	0.51
MWDISCL _{REPEATED}	·	0.009	0.52		0.021	1.21		0.031	1.83 *	0.001	0.05	0.01	1 0.57	0.030	1.69 *
LOGMC	+	0.011	4.41	* * * ·	0.008	4.16	* * · * ·	0.011	3.11 ***	0.007	1.59	0.00	8 1.86 *	0.008	1.28
LEV _{t-1}	ı	0.044	1.91	*	0.045	2.32	* *	-0.003	-0.11	0.062	1.62	0.05	3 1.60	0.004	0.11
PB_{t-1}	+	-0.002	-1.47		0.000	0.20		-0.001	-0.95	-0.002	- 0.98	0.00	0 0.16	0.001	0.42
Δ INCOME _{t-1}	+	-0.063	-1.85	*	-0.104	3.08	* * *	0.049	0.85	-0.113	2.41 *	* -0.110	5 2.18 **	0.087	1.24
ROA_{t-1}	+	0.494	12.38	* * *	0.263	7.16	* * *	0.087	1.32	0.493	8.51 *:	** 0.210) 3.59 ***	-0.024	0.21
BIG4	+	-0.001	-0.07		0.008	0.83		-0.006	-0.46	0.000	0.01	-0.00	2 0.12	0.008	0.39
DACC	ċ	0.005	3.42	* * *	0.006	3.82	* * *	0.003	1.69 *	0.004	1.07	00.00	4 1.19	-0.003	- 0.69
Intercept		0.114	2.69	* * *	0.008	0.27		-0.037	-0.56	0.074	1.07	00.00	7 0.12	-0.110	- 0.70
Annual fixed effects?		Yes			Yes			Yes		Yes		Ye	S	Yes	
Industry fixed effects?		Yes			Yes			Yes		Yes		Ye	S	Yes	
Adj. R ²		0.24			0.17			0.01		0.23		0.1	_	-0.00	
Ν		5,439			5,439			3,839		1,432		1,432	2	748	
<i>Notes:</i> See Appendix 1 1 and *, respectively. State fixed effects are not report	or the va stical siξ rted.	riable def mificance	initions. S s are calc	statistic ulated	cal signifi by cluste	cances ring th	based c e standa	on two-tai ard errors	led tests at th within comp	le 1 percentanies (Pet	t, 5 percen ersen 2009	t, and 10 pe). For brevi	rcent levels ar ty, results for	e denoted b the year an	y ***, **, d industry

4.4 Additional tests

In the main analyses, we attempted to alleviate the concern that the performance of material weakness companies is systematically lower than that in the EIC control companies by 1) using a propensity-score-matched sample, and 2) employing performance-adjusted measures of real earnings management. Moreover, in order to scrutinize the *incentives* for real earnings management, we investigate whether the association between MW and real earnings management differs based on previous performance.

We construct two sub-samples based on performance in the *previous* fiscal year: 1) poor performing companies with a negative change in net income (*DINCOME*_t. $_{I}<0$), or a negative return on assets ($ROA_{t-I}<0$), and 2) better performing companies with both a positive change in net income (*DINCOME*_{t-I}>0), and a positive return on assets ($ROA_{t-I}>0$). Moreover, we include in those sub-samples only those observations that show an increase in net income in the current fiscal year (*DINCOME*_t>0). We rerun the analyses using those sub-samples. The results reported in Table 5 indicate that when the sample consists of previously poorly performing companies, the existence of material weaknesses (*MWEXIST*) is associated with higher levels of abnormal production costs (Panel A columns 1–3) and lower levels of abnormal discretionary expenses (Panel B columns 1–2).²⁰ This suggests that, although the incentive is similar (being previous poor performance), the management of a material weakness company is the one employing these real earnings management methods.

When the sample consists of previously better performing companies, companies disclosing material weaknesses (*MWDISCL*_{CLEAN}) have higher levels of abnormal production costs (Panel A columns 4–6) and lower levels of discretionary expenses (Panel B columns 4–6) than companies that have never reported 404 material weaknesses. That is, when investigating a sample of companies that have all previously showed good financial performance (i.e., the sample companies do not have a clear performance-related incentive to manage earnings), the disclosure of material weaknesses (*MWDISCL*_{CLEAN}) is significantly associated with real earnings management, suggesting that the negative publicity associated with the disclosure of material weakness induces management to employ these real earnings management methods.

²⁰ The association with the performance-matched measure is insignificant (column 3), but it should be noted that the model fit is weak.

The results of the abnormal operating cash flows are somewhat inconsistent (Panel C of Table 5). In the sample of previously poor performing companies, $MWDISCL_{CLEAN}$ is marginally negatively associated with abnormal operating cash flows, but not when the dependent variable is the performance-adjusted measure. Moreover, in the sample of better performing companies, the coefficient on MWEXIST is marginally significant when the dependent variable is $AB_OCF_LAGROAadded$.

Panel A: Abnorm	al pro	oductior	1 costs															
	-		(1)			(2)		(3)			(4)			(5)		(9)		
		Poor	perform		Poor J	perform	P_{C}	or perform	n	Better	perforn	u	Better	r perform	B	etter po	erforn	u
Dependent variable		AB_{-}	PROD		AB_ LAGR(PROD DAadded	A. RC	AB_PROD DAmatche	d -	AB_	PROD		AB	PROD OAadded		AB_P] OAma	ROD atched	_
	Exp. sign	Est.	<u>t</u>		Est.	<u>t</u>	Est.			Est.	It .		Est.	<u>t</u>	Es	انب	t.	
MWEXIST) +	0.083	3.25 *	*	0.084	3.32 **	* 0.06	7 1.87	*	0.003	0.12		0.006	0.23	-0.0	11 -0	.36	
MWDISCL _{CLEAN}	+	0.019	0.65		0.029	1.04	0.03	1 0.81		0.063	1.96 *	*	0.063	2.06 **	0.0	68 1	* 62.	
MWDISCL _{REPEATED}	+	0.054	1.57		0.061	1.68 *	0.05	7 1.45		0.014	0.24		0.031	0.53	-0.0	19 -0	.32	
LOGMC	ı	-0.008	-1.60	I	0.003	-0.63	-0.00	1.12	·	-0.010	-1.68 *	*	-0.007	-1.15	-0.0	13 -2	.16 *	*
LEV _{t-1}	+	0.196	4.40 *	* *	0.210	4.67 **	* 0.11	9 2.21	*	0.144	2.08 \$	*	0.122	1.85 *	0.1	14 1	.60	
PB_{t-1}	ı	-0.007	-3.86 *	۱ * *	0.008	-4.50 **	* -0.00	8 -3.16	* * *	-0.013	-4.36 ¥	* *	-0.013	-4.62 **	0.0	11 -3	* 88.	* *
AINCOME _{t-1}	ı	0.004	0.09		0.007	0.15	0.13	8 1.80	'	-0.096	-1.07	-	-0.078	-0.90	-0.0	47 -0	.50	
ROA_{t-1}	ı	-0.038	-0.84	I	0.041	-0.86	-0.00	7 -0.07	·	.1.017	-5.00 \$	* *	-0.944	-4.75 **	:* -0.3	75 -1	.82	
BIG4	ı	-0.010	-0.39	'	0.019	-0.77	-0.02	3 -0.77	'	-0.025	-0.69	-	-0.027	-0.78	-0.0	37 -0	76.	
DACC	ė	-0.002	-1.02	I	0.002	-0.79	-0.00	12 -0.46		0.001	0.45		0.000	-0.20	0.0	03 1	.26	
Intercept		-0.018	-0.21	I	0.002	-0.03	0.14	0 1.10		0.184	2.92 *	* *	0.202	3.36 **	** 0.1	33 2	.01 *	*
Annual fixed effects?		Yes			Yes		Ύε	SS		Yes			Yes		Y	es		
Industry fixed effects?		Yes			Yes		Υє	SS		Yes			Yes		Y	es		
Adj. R ²		0.06			0.07		0.0	15		0.18			0.18		0.	08		
Z		1 561			1 561		1 03	9		1 638			1 638		16	26		

Additional tests for sub-samples based on previous performance and current year increase in net income Table 5.

Panel B: Abnorm	al dis(cretion	ary ex	pens	ses												Í
			(1)			(2)			(3)		(4)			(5)		(9)	
		Poor	perfori	ц	Pool	perfor	m.	Poor	perform	Bett	er perfo	rm	Bettei	r perform	Bett	er perfo	гш
Dependent variable		AB	DISX		AE LAGF	S DISX TOAad	ر ded	ROA	_DISX matched	Α	B_DISA		AB LAGR	DISX OAadded	RO.	B_DISX Amatch	ed
	Exp. sign	Est.	<u>t</u>		Est.	1t		Est.	t	Est.	<u>t</u>		Est.	۱ ر	Est.	lt.	
MWEXIST) 1	-0.230	-4.53	* * *	-0.187	-4.11	* * *	-0.044	-1.16	-0.002	-0.04		0.024	0.54	0.009	0.24	
MWDISCL _{CLEAN}	,	0.010	0.13		-0.051	-0.74		-0.094	-1.00	-0.109	-2.04	* *	-0.094	-1.94 *	-0.084	-2.12	* *
MWDISCL _{REPEATED}	ı	-0.085	-1.22		-0.089	-1.41		-0.093	-1.44	-0.064	1.05		-0.047	-0.86	-0.127	-1.23	
LOGMC	+	-0.008	-0.57		-0.007	-0.63		-0.016	-0.97	-0.026	5 -2.25	* *	-0.013	-1.30	-0.027	-2.44	*
LEV_{t-1}	ı	-0.378	-3.90	* * *	-0.393	-4.72	* * *	-0.215	-2.27 **	* -0.109	0.85		-0.159	-1.51	-0.045	-0.38	
PB_{t-1}	+	0.014	3.09	* * *	0.011	3.10	* * *	0.017	2.41 **	* 0.023	4.60	* * *	0.019	4.28 ***	0.018	4.12	* * *
AINCOME _{t-1}	+	0.022	0.14		0.102	0.76		-0.048	-0.25	0.056	0.30		0.168	1.10	-0.220	-1.11	
ROA_{t-1}	+	-0.412	-2.86	* * *	0.021	0.18		0.065	0.22	0.739	1.18		1.006	1.82 *	0.434	0.75	
BIG4	+	0.006	0.11		0.020	0.38		0.056	1.25	0.026	0.50		0.034	0.74	0.047	0.95	
DACC	ė	-0.018	-1.85	*	-0.018	-2.07	* *	-0.006	-0.84	-0.037	-2.83	* * *	-0.036	-3.09 ***	-0.005	-0.82	
Intercept		-0.391	-2.95	* * *	-0.194	-1.49		-0.272	-1.95 *	-0.322	: -1.64		-0.237	-1.27	0.039	0.19	
Annual fixed effects?		Yes			Yes			Yes		Yes			Yes		Yes		
Industry fixed effects?		Yes			Yes			Yes		Yes			Yes		Yes		
$Adj. R^2$		0.17			0.18			0.01		0.13			0.16		0.04		
Ν		1,561			1,561			1,036		1,638			1,638		1,626		

Panel C: Abnorma	l opera	ting ca:	sh flo	SW													
			(1)			(2)			(3)		(4)			5)		(9)	
		Poor	perfor	m.	Poor	perfor	щ	Poor J	perform	Betto	ar perform	_	Better	perform	Bettei	. perform	_
Dependent variable		AF	3_0CF		AF LAGF	OCF OAad	ded	AB ROAr	_OCF natched	A	B_OCF		LAGR	OCF DAadded	AB ROA	_OCF matched	
	Exp. sign	Est.	<u>t</u>		Est.	<u>t</u>		Est.	lt.	Est.	I C		Est.	t	Est.	<u>t</u>	
MWEXIST	ı	0.024	1.04		0.024	1.21		-0.031	-1.30	-0.011	- 0.49	-	.029	- * 1.70	-0.008	- 0.36	
MWDISCL _{CLEAN}	ı	-0.057	- 1.76	*	-0.055	- 1.86	*	0.003	0.06	0.009	0.42	0	0.005	0.42	0.015	0.76	
MWDISCL _{REPEATED} LOGMC	· +	0.018 0.004	0.60 0.70		0.031 0.003	1.22 0.78		0.023 0.003	0.77 0.42	0.005 0.013	0.16 3.39 *:)) **).016).006	0.68 1.95 *	0.043 0.016	1.22 4.46 **	*
LEV _{t-1}	·	0.114	2.67	* * *	0.116	3.21	* * *	0.063	1.41	-0.045	- 1.00	-	.023	- 0.66	-0.051	- 1.25	
PB_{t-1}	+	-0.002	- 1.18		-0.001	- 0.73		-0.002	-1.07	0.000	0.25	0	.001	0.66	-0.001	- 0.38	
$\Delta INCOME_{t-1}$	+	-0.115	2.10	* *	-0.140	2.72	* * *	-0.135	-1.86 *	0.041	0.51	0	020)	0.32	0.045	0.58	
$\mathrm{ROA}_{\mathrm{t-l}}$	+	0.516	8.84	* * *	0.258	5.27	* * *	0.157	1.88 *	0.598	3.53 *:) **	.426	3.46 ***	0.193	1.31	
BIG4	+	-0.003	0.13		-0.004	- 0.24		0.000	0.02	-0.011	- 0.57	0	0.002	0.12	-0.006	- 0.30	
DACC	ż	0.006	1.77	*	0.007	2.34	*	0.005	1.24	0.009	2.03 *:	0 *	001.007	1.95 *	0.006	1.55	
Intercept		0.278	5.08	* * *	0.150	3.46	* * *	0.160	1.73 *	0.077	1.05	0	0.001	0.03	-0.103	- 1.17	
Annual fixed effects?		Yes			Yes			Yes		Yes			Yes		Yes		
Industry fixed effects?		Yes			Yes			Yes		Yes			Yes		Yes		
Adj. R ²		0.26			0.18			0.00		0.15			0.15		0.02		
Ν		1,561			1,561			1,036		1,638		1	,638		1,626		
<i>Notes</i> : See Appendix 1 f and *, respectively. Stati fixed effects are not repo	or the var stical sigr rted.	iable defi nificances	initions. 3 are cal	. Statist culated	ical signi I by cluste	ficance: sring the	s based e standé	on two-té ard errors	uiled tests a within con	tt the 1 per npanies (P	cent, 5 pero etersen 200	cent, an 9). For	id 10 pe brevity	rcent levels a , results for th	re denotec he year an	l by ***, * d industry	* ~

4.5 Robustness checks

As of 2007, Auditing Standard No. 2 was replaced by Auditing Standard No. 5 (AS5), which recognizes scalability in evaluating controls, and requires a topdown risk-based approach to controls testing. Although including indicator variables for fiscal years should capture the yearly effects on the material weakness disclosures, we perform the propensity-score matching by replacing the year dummy variables in Model (1) with an indicator variable for the company years after 2006. The regression results of estimating Model (5) using the alternative propensity-score-matched sample are presented in Table 6. The inferences from these results are essentially similar to those from the main analyses, although some significance levels vary. For example, $MWDISCL_{CLEAN}$ is significantly associated with each measure of abnormal production costs (Panel A of Table 6). Moreover, MWEXIST and abnormal discretionary expenses is insignificant when the dependent variable is the basic measure, but significant at the 0.05 level when the dependent variable is the performance-matched measure (Panel B of Table 6).

Real earnings management through overproduction is applicable to manufacturing companies, which is why we also estimate the model of abnormal production costs for manufacturing companies (two-digit SIC codes between 20 and 39) (Cohen et al. 2008). The inferences from the results presented in Table 7 are similar to those from the main analysis (Panel A of Table 4).

Because our first hypothesis relates to the existence but not the disclosure of material weaknesses and our second hypothesis relates specifically to the disclosure, we exclude those *MWEXIST* observations that disclosed 302 weaknesses²¹ in the first and/or the second quarter from the sample. The inferences from these (untabulated) results are essentially similar to those from the main analyses.

²¹ No distinction is made between material weaknesses and significant deficiencies.

	P									
			(1)			(2)			(3)	
Dependent variable		AR)	AB	_PROI)	AB	PROD)
Dependent variable		AD	_1 KO1	,	LAGE	ROAad	ded	ROA	matche	d
	Exp.	Est.	t		Est.	t		Est.	t	
	sıgn		-			-			-	
MWEXIST	+	0.030	1.69	*	0.034	1.97	**	0.031	1.41	
MWDISCL _{CLEAN}	+	0.051	2.30	**	0.056	2.66	***	0.070	2.37	**
MWDISCL _{REPEATED}	+	0.041	1.69	*	0.056	2.19	**	0.033	0.93	
LOGMC	-	-0.012	-2.27	**	-0.009	-1.66	*	-0.001	-0.10	
LEV _{t-1}	+	0.186	3.78	***	0.176	3.68	***	0.079	1.11	
PB _{t-1}	-	-0.004	-1.79	*	-0.006	-2.27	**	-0.013	-1.87	*
$\Delta INCOME_{t-1}$	-	-0.002	-0.05		0.036	0.81		0.006	0.07	
ROA _{t-1}	-	-0.049	-0.91		-0.056	-1.02		-0.184	-1.08	
BIG4	-	0.014	0.63		0.011	0.51		0.002	0.08	
DACC	?	0.000	0.15		0.002	0.58		-0.001	-0.29	
Intercept		0.000	-0.01		0.021	0.28		0.128	0.75	
Annual fixed effects?		Yes			Yes			Yes		
Industry fixed effects?		Yes			Yes			Yes		
Adj. R ²		0.04			0.03			0.05		
N		1,468			1,468			792		

Table 6. Propensity-score-matched sample with AS5 controlled

Panel B: Abnormal discretionary expenses

			(1)			(2)			(3)	
Dependent variable		AB			AB	_DISX	-	AB	_DISX	-
Dependent variable		110		-	LAGE	ROAad	ded	ROA	match	ed
	Exp. sign	<u>Est.</u>	<u>t</u>		<u>Est.</u>	<u>t</u>		<u>Est.</u>	<u>t</u>	
MWEXIST	-	-0.024	-0.68		-0.009	-0.26		-0.054	-1.99	**
MWDISCL _{CLEAN}	-	-0.052	-1.73	*	-0.054	-1.89	*	-0.113	-2.30	**
MWDISCL _{REPEATED}	-	0.005	0.09		-0.020	-0.37		-0.126	-2.55	**
LOGMC	+	-0.004	-0.34		-0.006	-0.60		0.017	1.36	
LEV _{t-1}	-	-0.241	-2.40	**	-0.304	-3.53	***	-0.281	-3.16	***
PB _{t-1}	+	0.012	2.64	***	0.012	2.81	***	0.015	2.07	**
$\Delta INCOME_{t-1}$	+	0.182	1.27		0.266	1.98	**	-0.049	-0.34	
ROA _{t-1}	+	-0.551	-3.95	***	-0.080	-0.58		-0.003	-0.01	
BIG4	+	-0.040	-0.86		-0.023	-0.54		-0.020	-0.55	
DACC	?	-0.020	-2.29	**	-0.022	-2.64	***	0.003	0.48	
Intercept		-0.278	-1.84	*	-0.034	-0.23		-0.112	-0.67	
Annual fixed effects?		Yes			Yes			Yes		
Industry fixed effects?		Yes			Yes			Yes		
Adj. R ²		0.11			0.10			0.07		
N		1,468			1,468			792		

Tallel C. Ablioffiai	operat	ing casi	I HOW	3						
			(1)			(2)			(3)	
Dependent variable		ΔŦ	3 OCF		AI	3_OCF		AB	_OCF	
Dependent variable		111	_001		LAGI	ROAado	ded	ROA	matched	
	Exp.	Est.	t		Est.	t		Est.	t	
	sıgn	<u></u>	-		<u></u>	-			-	
MWEXIST	-	-0.025	-1.62		-0.028	-2.11	**	-0.009	-0.60	
MWDISCL _{CLEAN}	-	-0.020	-1.28		-0.014	-1.01		-0.002	-0.08	
MWDISCL _{REPEATED}	-	-0.004	-0.20		0.008	0.38		0.025	1.20	
LOGMC	+	0.008	1.75	*	0.009	2.03	**	-0.006	-0.90	
LEV _{t-1}	-	0.116	3.09	***	0.112	3.22	***	0.114	2.42	**
PB _{t-1}	+	-0.002	-0.76		0.000	-0.04		-0.002	-0.58	
$\Delta INCOME_{t-1}$	+	-0.090	-1.70	*	-0.145	-2.62	***	0.068	0.74	
ROA _{t-1}	+	0.502	7.99	***	0.218	3.60	***	0.094	0.57	
BIG4	+	0.010	0.54		0.010	0.61		0.019	0.93	
DACC	?	0.007	2.12	**	0.007	2.37	**	0.005	1.40	
Intercept		0.074	1.04		-0.017	-0.34		-0.151	-1.20	
Annual fixed effects?		Yes			Yes			Yes		
Industry fixed effects?		Yes			Yes			Yes		
$Adj. R^2$		0.22			0.11			0.01		
Ν		1,468			1,468			792		

Panel C: Abnormal operating cash flows

Notes: See Appendix 1 for the variable definitions. Statistical significances based on two-tailed tests at the 1 percent, 5 percent, and 10 percent levels are denoted by ***, **, and *, respectively. Statistical significances are calculated by clustering the standard errors within companies (Petersen 2009). For brevity, results for the year and industry fixed effects are not reported.

			(1)			(2)			(3)	
Dependent variable		AE	B_PRO	D	AB_ LAGR	_PROD OAadde	ed	AB ROA	_PROE) ed
	Exp. sign	<u>Est.</u>	<u>t</u>		<u>Est.</u>	<u>t</u>		<u>Est.</u>	<u>t</u>	
MWEXIST	+	0.049	3.41	***	0.048	3.33	***	0.022	1.19	
MWDISCL _{CLEAN}	+	0.051	2.74	***	0.052	2.83	***	0.024	1.10	
MWDISCL _{REPEATED}	+	0.070	3.44	***	0.081	3.94	***	0.007	0.25	
LOGMC	-	-0.008	-2.07	**	-0.007	-1.67	*	-0.008	-1.70	*
LEV _{t-1}	+	0.243	6.37	***	0.244	6.47	***	0.209	4.58	***
PB _{t-1}	-	-0.012	-6.51	***	-0.012	-6.58	***	-0.014	-5.59	***
$\Delta INCOME_{t-1}$	-	0.000	-2.13	**	0.000	-1.83	*	0.000	-0.64	
ROA _{t-1}	-	-0.130	-3.60	***	-0.125	-3.44	***	-0.075	-1.01	
BIG4	-	-0.003	-0.12		-0.004	-0.22		-0.024	-0.97	
DACC	?	0.001	1.09		0.002	1.41		0.002	1.45	
Intercept		0.019	0.45		0.018	0.41		0.055	1.09	
Annual fixed effects?		Yes			Yes			Yes		
Industry fixed effects?		Yes			Yes			Yes		
Adj. R ²		0.12			0.12			0.11		
Ν		3,742			3,742			2,631		

Table 7. Abnormal production costs – manufacturing companies SIC 20XX-39XX

Notes: See Appendix 1 for the variable definitions. Statistical significances based on two-tailed tests at the 1 percent, 5 percent, and 10 percent levels are denoted by ***, **, and *, respectively. Statistical significances are calculated by clustering the standard errors within companies (Petersen 2009). For brevity, results for the year and industry fixed effects are not reported.

5 Conclusions

This study examines the association between SOX 404 material weaknesses and earnings management by manipulating real operational activities. Our empirical findings suggest that both the existence and the disclosure of SOX 404 material weaknesses are related to real earnings management. For companies with existing material weaknesses, the results consistently indicate overproduction being used as a method of real earnings management. Although the association may stem from the behavior of an opportunistic management, the poor 'tone at the top' may also flow through to inventory-related operational decisions via low quality internal information caused by material weaknesses (see Feng et al. 2015). Furthermore, the findings show some evidence that the existence of material weaknesses is associated with the reduction of discretionary expenses.

Companies disclosing material weaknesses are using inventory overproduction and reduction of discretionary expenses as real earnings management methods. Cutting discretionary expenses in particular is probably the most feasible real earnings management method after disclosure. We do not find evidence of sales manipulation among companies disclosing material weaknesses, which might be due to reduced customer demand after the disclosure of internal control weaknesses (Su et al. 2014), making sales manipulation as a dysfunctional real earnings management method for material weakness companies.

The additional analyses reveal that among previously poor performing companies, the *existence* of material weaknesses in particular is associated with real earnings management, implying that the management of a company with material weaknesses employs real earnings management methods. However, when the sample consists of previously better performing companies, it is mainly the companies *disclosing* material weaknesses that appear to be employing real earnings management methods, suggesting that the incentive stems from the negative publicity flowing from material weakness disclosures.

Our results offer several avenues for future research. First, while we interpret our results as indicating that material weakness companies do engage in real earnings management, whether intentionally or unintentionally, future research could strive to investigate the question of intentionality further. Second, as our data set is in-adequate to address the underlying reason for the variance in these real earnings management methods, we surmise that one possible reason is the desirability of a specific method in a certain situation and a fear of real earnings management being detected by outsiders determining the preferred order of the application of real earnings management methods.

In summary, the findings of this study suggest that, after controlling for various company characteristics, material weaknesses reflect pervasive control problems and the 'tone at the top', enabling real earnings management behavior. Material weaknesses hence identify a setting where there is a greater risk of the value of stockholders' investments in the company being reduced in the long run via real activities manipulation. The findings of this study extend the research on internal control over financial reporting and the research on real earnings management by providing new information on situations that can provide fertile ground for real earnings management, and on what might prompt it.

We acknowledge as a limitation of the study that some material weaknesses might not be detected and disclosed (see e.g., Rice and Weber 2012). However, this concern is not likely to distort the statistical significance of our findings if the underlying reasons for undetected weaknesses are random. In addition, there is a risk that the models used in this study to capture real earnings management might measure it erroneously, and thus bias our results. This is, however, the same risk faced by prior real earnings management studies (e.g., Roychowdhury 2006; Cohen et al. 2008; Zang 2012). Finally, we acknowledge that the common problem of endogeneity cannot be completely ruled out, although we do address this issue by employing a propensity-score-matching technique and using the matched sample in our main analyses.

References

Ashbaugh-Skaife, H., D. W. Collins & W. R. Kinney, Jr. (2007). The discovery and reporting of internal control deficiencies prior to SOX-mandated audits. *Journal of Accounting and Economics* 44:1-2, 166–192.

Ashbaugh-Skaife, H., D. W. Collins, W. R. Kinney, Jr. & R. LaFond (2008). The effect of SOX internal control deficiencies and their remediation on accrual quality. *The Accounting Review* 83:1, 217–250.

Ashbaugh-Skaife, H., D. W. Collins, W. R. Kinney, Jr. & R. LaFond (2009). The effect of SOX internal control deficiencies on firm risk and cost of equity. *Journal of Accounting Research* 47:1, 1–43.

Badertscher, B. (2011). Overvaluation and the choice of alternative earnings management mechanisms. *The Accounting Review* 86:5, 1491–1518.

Bauer, A. (2014). Tax avoidance and the implications of weak internal controls. *Working paper*. University of Illinois at Urbana-Champaign.

Beneish, M. D., M. B. Billings & L. D. Hodder (2009). Internal control weaknesses and information uncertainty. *The Accounting Review* 83:3, 665–703.

Chan, K.C., B. Farrell & P. Lee (2008). Earnings management of firms reporting material internal control weaknesses under section 404 of the Sarbanes-Oxley act. *Auditing: A Journal of Practice & Theory* 27:2, 161–179.

Cheng, M., D. Dhaliwal & Y. Zhang (2013). Does investment efficiency improve after the disclosure of material weaknesses in internal control over financial reporting? *Journal of Accounting and Economics* 56:1, 1–18.

Chi, W., L. L. Lisic & M. Pevzner (2011). Is enhanced audit quality associated with greater real earnings management? *Accounting Horizons* 25:2, 315–335.

Clinton, S., A. Pinello & H. A. Skaife (2014). The implications of ineffective internal control and SOX 404 reporting for financial analysts. *Journal of Accounting and Public Policy* 33, 303–327.

Cohen, D. A., A. Dey & T. Z. Lys (2008). Real and accrual-based earnings management in the pre- and post-Sarbanes-Oxley periods. *The Accounting Review* 83:3, 757–787. Cohen D., S. Pandit, C. Wasley & T. Zach (2014). Measuring real activity management. *Working paper*. University of Texas at Dallas, University of Illinois at Chicago, University of Rochester, The Ohio State University.

Cohen, D. A. & P. Zarowin (2010). Accrual-based and real earnings management activities around seasoned equity offerings. *Journal of Accounting and Economics* 50:1, 2–19.

COSO (1992). Committee of Sponsoring Organization of Treadway Commission. Internal Control – Integrated Framework. New York: AICPA.

Doyle, J., W. Ge & S. McVay (2007a). Accruals quality and internal control over financial reporting. *The Accounting Review* 82:5, 1141–1170.

Doyle, J., W. Ge & S. McVay (2007b). Determinants of weaknesses in internal control over financial reporting. *Journal of Accounting and Economics* 44:1-2, 193–223.

Feng, M., C. Li, S. E. McVay & H. Skaife (2015). Does ineffective internal control over financial reporting affect a firm's operations? Evidence from firms' inventory management. *The Accounting Review* 90:2, 529–557.

Ge, W. & S. McVay (2005). The disclosure of material weaknesses in internal control after the Sarbanes-Oxley act. *Accounting Horizons* 19:3, 137–158.

Graham, J. R., C. R. Harvey & S. Rajgopal (2005). The economic implications of corporate financial reporting. *Journal of Accounting and Economics* 40:1-3, 3–73.

Gunny, K. (2010). The relation between earnings management using real activities manipulation and future performance: Evidence from meeting earnings benchmarks. *Contemporary Accounting Research* 27:3, 855–888.

Healy, P. M. & J. M. Wahlen (1999). A review of the earnings management literature and its implications for standard setting. *Accounting Horizons* 13:4, 365– 383.

Hoitash, R., U. Hoitash & J. C. Bedard (2009). Corporate governance and internal control over financial reporting: A comparison of regulatory regimes. *The Accounting Review* 84:3, 839–867.

Jones, J. (1991). Earnings management during import relief investigations. *Journal of Accounting Research* 29:2, 193–228.

Kim, B., L. L. Lisic & M. Pevzner (2010). Debt covenant slacks and real earnings management. *Working paper*. American University, George Mason University.

Kim, J., B. Y. Song & L. Zhang (2011). Internal control weakness and bank loan contracting: Evidence from SOX Section 404 disclosures. *The Accounting Review* 86:4, 1157–1188.

Kim, Y. & M. S. Park (2014). Real activities manipulation and auditors' clientretention decisions. *The Accounting Review* 89:1, 367–401.

Kothari, S. P., A. J. Leone & C. E. Wasley (2005). Performance matched discretionary accrual measures. *Journal of Accounting and Economics* 39:1, 163–197.

Kothari, S. P., N. Mizik & S. Roychowdhury (2012). Managing for the moment: The role of real activity versus accruals earnings management in SEO valuation. *Working paper*. Boston College, Massachusetts Institute of Technology, University of Washington.

Li, C., L. Sun & M. Ettredge (2010). Financial executive qualifications, financial executive turnover, and adverse SOX 404 opinions. *Journal of Accounting and Economics* 50:1, 93–110.

Lopez, T. J., S. D. Vandervelde & Y. Wu (2009). Investor perceptions of an auditor's adverse internal control opinion. *Journal of Accounting and Public Policy* 28:3, 231–250.

Myllymäki, E-R. (2014). The persistence in the association between Section 404 material weaknesses and financial reporting quality. *Auditing: A Journal of Practice & Theory* 33:1, 93–116.

PAIB (2011). Professional Accountants in Business Committee. *Global survey on risk management and internal control. Results, analysis, and proposed next steps.* International Federation of Accountants. Available from: http://www.ifac.org/.

PCAOB (2007). Public Company Accounting Oversight Board. Auditing Standard No. 5 – An audit of internal control over financial reporting that is integrated with an audit of financial statements.

Petersen, M. A. (2009). Estimating standard errors in finance panel data sets: Comparing approaches. *Review of Financial Studies* 22:1, 435–480.

Rezee, Z., R. Espahbodi, P. Espahbodi & H. Espahbodi (2012). Firm characteristics and stock price reaction to SOX 404 compliance. *Abacus* 48:4, 473–501.

Rice, S. C. & D. P. Weber (2012). How effective is internal control reporting under SOX 404? Determinants of the (non-) disclosure of existing material weaknesses. *Journal of Accounting Research* 50:3, 811–843.

Roychowdhury, S. (2006). Earnings management through real activities manipulation. *Journal of Accounting and Economics* 42:3, 335–370.

Schneider, A. & B. K. Church (2008). The effect of auditors' internal control opinions on loan decisions. *Journal of Accounting and Public Policy* 27:1, 1–18.

Schneider, A., A. A. Gramling, D. R. Hermanson & Z. Ye (2009). A review of academic literature on internal control reporting under SOX. *Journal of Accounting Literature* 28, 1–46.

Shelton, S.W. & O. R. Whittington (2008). The influence of the auditor's report on investors' evaluations after the Sarbanes-Oxley Act. *Managerial Auditing Journal* 23:2, 142–160.

Skaife, H. A., D. Veenman & D. Wangerin (2013). Internal control over financial reporting and managerial rent extraction: Evidence from the profitability of insider trading. *Journal of Accounting and Economics* 55:1, 91–110.

SOX (2002). *The Sarbanes-Oxley act of 2002*. United States House of Representatives, Public Law 107–204 [H.R. 3763]. Washington D.C.

Su, L., X. Zhao & G. Zhou (2014). Do customers respond to the disclosure of internal control weaknesses? *Journal of Business Research* 67:7, 1508–1518.

Zang, A. Y. (2012). Evidence on the trade-off between real activities manipulation and accrual-based earnings management. *The Accounting Review* 87:2, 675– 703.

Zhang, Y., J. Zhou & N. Zhou (2007). Audit committee quality, auditor independence, and internal control weaknesses. *Journal of Accounting and Public Policy* 26:3, 300–327.

Appendices

Appendix 1. Variable definitions

Variable	Definition
AB_PROD	The estimated residuals from the following industry-year regression $\frac{PROD_t}{TA_{t-1}} = \alpha + \beta_1 \frac{1}{TA_{t-1}} + \beta_2 \frac{S_t}{TA_{t-1}} + \beta_3 \frac{\Delta S_t}{TA_{t-1}} + \beta_4 \frac{\Delta S_{t-1}}{TA_{t-1}} + \varepsilon.$
AB_DISX	The estimated residuals from the following industry-year regression $\frac{DISX_t}{TA_{t-1}} = \alpha + \beta_1 \frac{1}{TA_{t-1}} + \beta_2 \frac{S_{t-1}}{TA_{t-1}} + \varepsilon.$
AB_OCF	The estimated residuals from the following industry-year regression $\frac{OCF_t}{TA_{t-1}} = \alpha + \beta_1 \frac{1}{TA_{t-1}} + \beta_2 \frac{S_t}{TA_{t-1}} + \beta_2 \frac{\Delta S_t}{TA_{t-1}} + \varepsilon.$
PROD	The sum of the cost of goods sold in year t and the change in inventory from t-1 to t.
DISX	The discretionary expenditure that are the sum of R&D expenditure, adver- tising, and selling, general, and administrative (SG&A) expenditure.
OCF	Cash flow from operations.
ТА	Total assets.
S	Net sales.
AB_PROD LAGROAadded	The estimated residuals from the following industry-year regression $\frac{PROD_t}{TA_{t-1}} = \alpha + \beta_1 \frac{1}{TA_{t-1}} + \beta_2 \frac{S_t}{TA_{t-1}} + \beta_2 \frac{\Delta S_t}{TA_{t-1}} + \beta_4 \frac{\Delta S_{t-1}}{TA_{t-1}} + \beta_5 ROA_{t-1} - \beta_5 ROA_{t-1}$
AB_DISX LAGROAadded	The estimated residuals from the following industry-year regression $\frac{DISX_{t}}{TA_{t-1}} = \alpha + \beta_{1} \frac{1}{TA_{t-1}} + \beta_{2} \frac{S_{t-1}}{TA_{t-1}} + \beta_{3}ROA_{t-1} + \varepsilon.$
AB_OCF LAGROAadded	The estimated residuals from the following industry-year regression $\frac{OCF_t}{TA_{t-1}} = \alpha + \beta_1 \frac{1}{TA_{t-1}} + \beta_2 \frac{S_t}{TA_{t-1}} + \beta_2 \frac{\Delta S_t}{TA_{t-1}} + \beta_4 ROA_{t-1} + \varepsilon.$
AB_PROD ROAmatched	AB_PROD minus average AB_PROD of matched control companies based on fiscal-year, two-digit SIC code and return on assets between +/- 50
AB_DISX ROAmatched	AB_DISX minus average AB_DISX of matched control companies based on fiscal-year, two-digit SIC code and return on assets between +/- 50 percent.
AB_OCF ROAmatched	AB_OCF minus average AB_OCF of matched control companies based on fiscal-year, two-digit SIC code and return on assets between +/- 50 percent.
MWEXIST	A dummy variable with a value of one if the company-year includes first time material weaknesses in internal controls according to the subsequent 404 report, and zero otherwise.
MWDISCL _{CLEAN}	A dummy variable with a value of one if the previous company-year in- cluded first time material weaknesses, but the current company-year does not include weaknesses based on subsequent 404 report, and zero other- wise.

MWDISCL _{REPEATED}	A dummy variable with a value of one if the previous company-year in- cluded first time material weaknesses and the current company-year again includes weaknesses based on subsequent 404 report, and zero otherwise.
LOGMC	The natural logarithm of market capitalization.
LEV _{t-1}	Beginning of the year total debt divided by total assets.
PB _{t-1}	Lagged price-to-book ratio.
$\Delta INCOME_{t-1}$	Lagged change in a company's annual net income before extraordinary items, scaled by the beginning of the year total assets.
ROA _{t-1}	Lagged ROA calculated as net income before extraordinary items, scaled by the beginning of the year total assets.
BIG4	A dummy variable with a value of one if the auditor is one of the four largest audit firms, and zero otherwise.
DACC	Discretionary accruals estimated using the modified Jones (1991) model with ROA_{t-1} included to the model.
YEAR	Dummy variables for the fiscal years 2004–2012.
INDUSTRY	Dummy variables for two-digit SIC industry groups.
Variables used in the	propensity score matching [Model (1)] if not defined above:
MW	A dummy variable with a value of one if <i>MWEXIST</i> =1, <i>MWDISCL_{CLEAN}</i> =1, or <i>MWDISCL_{REPEATED}</i> =1, and zero otherwise.
LOGAGE	A natural logarithm of the number of years since the company was found- ed.
FOREIGN	A dummy variable with a value of one if a company reported geographic segment(s) other than the US, and zero otherwise.
SQRTSEGMENTS	The square root of the number of business segments reported by the com- pany.
RESTRUCTURE	A dummy variable with a value of one if a company reported restructuring expenses / charges, and zero otherwise.
ACQUISITION	A dummy variable with a value of one if an exemption to the assessment of internal controls indicates acquisition(s) during the past year, and zero otherwise.
INV	Total inventory divided by total assets.
LOSS	A dummy variable with a value of one if the net income before extraordinary items is negative in year <i>t</i> or <i>t</i> -1, and zero otherwise.
GC	A dummy variable with a value of one if a company received a going con- cern opinion, and zero otherwise.
ZSCORE	Altman's Z-score: 1.2*(working capital / total assets) + 1.4*(retained earn- ings / total assets) + 3.3*(earnings before interests and taxes / total assets) + 0.6*(market capitalization/total liabilities) + 1*(sales/total assets).
HIGHGROWTH	A dummy variable with a value of one if a company's sales growth from previous fiscal year falls into the top quartile in the sample, and zero oth- erwise
AUDI-	A dummy variable with a value of one if the signing audit firm changed
TORCHANGE	from previous year, and zero otherwise.
LITIGATION	A dummy variable with a value of one if the company operates in a liti- gious industry (SIC codes 2833–2836, 3570–3577, 3600–3674, 5200–
	5961, 7370-7374), and zero otherwise.
ICREPORTM	A dummy variable with a value of one if the internal control report is man- agement-only report (no auditor's internal control report), and zero other-

	(1	1)		(2)	
Dependent variable	М	W		MW	
	Estimate	<u>Z</u>		Estimate	<u>Z</u>
LOGMC	-0.218	-10.68	***	0.015	0.48
LOGAGE	-0.147	-4.78	***	0.013	0.31
FOREIGN	-0.120	-1.97	**	0.039	0.47
SQRTSEGMENTS	0.276	5.03	***	0.031	0.39
RESTRUCTURE	0.057	0.92		-0.033	-0.37
ACQUISITION	-0.057	-0.68		-0.003	-0.02
INV	0.197	0.83		-0.289	-0.91
LOSS	0.175	2.82	***	0.021	0.24
ROA	-0.185	-1.62		-0.016	-0.10
LEV _{t-1}	0.428	2.85	***	-0.076	-0.37
GC	0.526	3.94	***	-0.055	-0.28
ZSCORE	-0.000	-0.87		0.003	0.87
HIGHGROWTH	0.130	2.08	**	-0.115	-1.37
BIG4	-0.154	-2.09	**	0.013	0.13
AUDITORCHANGE	0.431	5.02	***	0.081	0.70
LITIGATION	0.225	2.83	***	0.011	0.10
ICREPORTM	-0.096	-1.03		0.071	0.54
Intercept	0.318	1.22		0.001	0.00
Annual fixed effects?	Yes			Yes	
Industry fixed effects?	Yes			Yes	
Ν	4,971			1,432	
Likelihood ratio	Chi-Squar Prob > Chi	e 1,073.98 Sg 0.000	3 0	Chi-Square, Prob > ChiSa	18.29 1.0000
Log likelihood	-174	4.89	-	-983.44	

Appendix 2. Probit regression results on estimating Model (1)

Notes: See Appendix 1 for the variable definitions. Statistical significance based on two-tailed tests at the 1 percent, 5 percent, and 10 percent levels are denoted by ***, **, and *, respectively. For brevity, results for the year and industry fixed effects are not reported.

INCUMBENT AUDIT FIRM-PROVIDED TAX SERVICES AND CLIENTS WITH LOW FINANCIAL REPORTING QUALITY⁺

The co-authors of this essay are H. Gin Chong and Erkki K. Laitinen

ABSTRACT

This study investigates whether tax services being provided by incumbent audit firms enhances or impairs the likelihood of acknowledging client companies' low financial reporting quality. In particular, we examine the association between taxrelated fees and the likelihood of timely restatements, and internal control weakness disclosures among a sample of US companies that all have misstatements in their financial information. The empirical findings indicate that companies paying higher tax-related fees are less likely to disclose SOX 404 internal control weaknesses, implying that underlying control problems are unacknowledged when incumbent audit firm-provided tax-related fees are higher. However, the findings suggest that merely providing both audit and tax-related services does not have an impact on audit quality per se, but rather it is the magnitude of the tax-related fees in particular that counts. We also find some evidence suggesting that companies paying higher tax-related fees have a greater likelihood of restatement lags, whereas companies paying smaller tax-related fees to their audit firm restate financial statements in a more timely manner. Overall, the findings suggest that audit scrutiny of client companies with low quality financial reporting is weaker when the magnitude of tax-related fees is higher.

Keywords: tax fees, financial reporting quality, internal control weakness

Acknowledgements: The authors gratefully appreciate the comments of the seminar participants at the Umeå School of Business and Economics, at the Business Department of Universitat Autònoma de Barcelona (Spain), and at the European Accounting Association Annual Congress in Tallinn (Estonia) (2014). Myllymäki gratefully acknowledges the financial support received from the Finnish Foundation for Economic Education, Jenny and Antti Wihuri Foundation, Nordea Bank Foundation, Oskar Öflund's Foundation, and Evald and Hilda Nissi Foundation. All remaining errors are our own.

1 Introduction

This study investigates whether tax services provided by incumbent audit firms enhance or impair the likelihood of acknowledging client companies' low financial reporting quality. Since the audit scandal involving Arthur Andersen, investors and regulators have expressed their concerns over the magnitude of non-audit services provided by incumbent audit firms, because they are perceived as a threat to auditors' independence, in that non-audit services might increase auditors' economic dependence on their clients, leading to lower audit quality. In the USA, the Sarbanes Oxley Act (SOX) (2002) prohibits audit firms from providing most non-audit services to their audit clients, but permits tax services due to the potential benefits from knowledge spillover to audit quality. After all, audit quality consists of both the likelihood that the auditor discovers a breach (implying expertise and audit effort) and the likelihood that the auditor discloses the breach (implying objectivity and independence) (DeAngelo 1981).

The bulk of prior studies have investigated the association between non-audit fees and audit/financial reporting quality, and reported somewhat mixed results (see Schneider et al. 2006 for a review). After the restrictions on the provision of most non-audit services by incumbent audit firms (SOX 2002), prior research has particularly focused on the influence of tax services in order to examine whether the expected benefits from knowledge spillover outweigh the problems of economic dependence (e.g., Seetharaman et al. 2011; Krishnan and Visvanathan 2011; Paterson and Valencia 2011). In this study, we approach the issue of incumbent audit firm-provided tax services using a sample of companies that all have poor financial reporting quality, that is, they are companies with misstatements that were eventually restated. The misstatements are determined from the restated periods indicated by restatement data. First, we investigate whether tax fees are associated with those misstatements in financial information that remain undiscovered in a fiscal year in question, to which we refer to as restatement lag. We therefore compare companies with restatement lag to those companies with misstatements restated more quickly. Secondly, we investigate whether tax fees are associated with the likelihood of Section 404 internal control weakness disclosures. Based on the view that an occurrence of a misstatement indicates existing internal control weaknesses (e.g., Eilifsen and Messier 2000; Rice and Weber 2012), disclosing material weakness would suggest that the company acknowledges the control weaknesses. The research setting featuring a sample of similar companies in terms of poor accounting quality allows examining the auditors' professional

skepticism in particular.¹ Thus, the underlying assumption is that high quality auditing should lead to the discovery of any material misstatements and internal control weaknesses, and auditors should require management to acknowledge them. Enhanced knowledge about the client via tax-services could make restatements more timely and material weakness disclosures more likely for companies with poor accounting quality. However, economic dependence may reduce an auditor's professional skepticism, resulting in restatement lags and unacknowl-edged control problems.

The inferences of the findings in prior studies that investigate the association between incumbent audit firm-provided tax services and financial reporting quality mostly support the knowledge spillover view. For example, Kinney et al. (2004) suggest that tax-related fees reduce the likelihood of restatements, implying there are benefits from knowledge spillover. See tharaman et al. (2011), however, report insignificant association between tax-related fees and restated periods, but significant negative association with tax-related restatements. These studies examine whether or not restatements/misstatements occur. Harris and Zhou (2013) suggest that tax services lead to reduced likelihood of non-tax internal control weaknesses but do not have an effect on tax-related weaknesses. A lower likelihood of restatements and/or internal control weaknesses could, however, also signify a reluctance to acknowledge and disclose them. Rice and Weber (2012) examine a sample of companies with existing internal control weaknesses and find that nonaudit fees makes it less likely that material weaknesses are disclosed, supporting the economic dependence view. We extend the findings of the prior studies on tax services and investigate the role of audit firm-provided tax services among companies with poor financial reporting quality.

This study uses a sample of fiscal-year observations of US companies from 2005–2012. Only the company-years involving misstatements are included in the sample used in the analyses. That analysis investigates the probability of a restatement lag (an indicator variable for company-years where restatement is disclosed after the filing date of internal control opinion) and the probability of internal control material weakness disclosure (an indicator variable for internal control reports disclosing material weaknesses). Our independent variables of interest are

¹ Nelson (2009) defines professional skepticism as "indicated by auditor judgments and decisions that reflect a heightened assessment of the risk that an assertion is incorrect, conditional on the information available to the auditor." He elaborates that "In many circumstances the assertion in question will be a client's assertion that the financial statements are free of material misstatement, but the definition could apply to other assertions as well (e.g., attesting to the effectiveness of a client's internal controls)."

(1) an indicator variable with a value of one if the tax fees paid to the incumbent auditor are greater than zero, and zero otherwise (*DTAXFEES*), (2) the tax fees paid to the incumbent auditor divided by the square root of total assets (*TAXFEES*), (3) the ratio of tax fees divided by total fees paid to the incumbent auditor (*TAXFEES_TF*), and (4) the ratio of tax fees divided by audit fees paid to the incumbent auditor (*TAXFEES_AF*).

The empirical findings of this study suggest that in all different model specifications, the indicator variable for tax services is insignificant. This implies that tax services provided by incumbent audit firms do not have either a qualityenhancing or quality-impairing impact. However, the results indicate that the continuous variables measuring the magnitude of tax-related fees (TAXFEES, TAXFEES TF, and TAXFEES AF) have negative effects on the likelihood of a material weakness disclosure, although the significance levels vary across different model specifications. That is, we find evidence implying that the higher (proportional) magnitude of tax fees results in unacknowledged control problems. In most model specifications, tax-related fees are not associated with the likelihood of restatement lags. However, when we further examine the impact of tax-related fees using a sample of only those companies that have paid tax-related fees to their auditors (i.e., observations of zero tax fees are excluded), the results indicate that tax fees divided by total fees and tax fees divided by audit fees are (mostly marginally) significantly associated with higher likelihood of restatement lags and lower likelihood of material weakness disclosures. These findings support the inference that it is the magnitude of the fees in particular that reduce the likelihood of acknowledging client companies' financial reporting problems. In summary, this study contributes to the literature investigating non-audit fees charged by incumbent audit firms by suggesting that, among companies with poor accounting quality, a stronger economic bond with the client resulting from higher levels of tax-related fees may jeopardize the auditor's independence and impair auditor's professional skepticism.

The remainder of the paper is organized as follows. The next section summarizes the literature on the relation between non-audit services and financial reporting quality, and develops the hypotheses. Section three describes the data and the methodology. Results are reported in Section four, and are followed by conclusions.

2 Background and hypotheses development

Over the past 30 years, audit firms have expanded their business areas to consultancy services, such as taxation, mergers and acquisitions, and risk management. Research on audit quality has investigated the possible benefits and disadvantages of an audit firm providing both audit and non-audit services to the same client concurrently. There are two conflicting hypotheses on the association between non-audit services being provided by an incumbent auditor and audit quality. The knowledge spillover view suggests that information acquired in the course of consulting flows to the audit partner, improving the quality of the audit (and vice versa) (e.g., Simunic 1984). However, these services are economically important to the audit firms. High non-audit fees increase an auditor's economic dependence on the client, thereby possibly impairing audit quality (e.g., Simunic 1984).²

Prior studies have reported mixed results on the association between non-audit fees and audit/financial reporting quality (see Schneider et al. 2006 for a review). While the findings of Frankel et al. (2002), Ferguson et al. (2004), and Kanagaretnam et al. (2011) imply that non-audit fees impair auditor independence, DeFond et al. (2002), Ashbaugh et al. (2003), Chung and Kallapur (2003), Raghunandan et al. (2003), and Reynolds et al. (2004) do not find a statistically significant association between non-audit fees and audit/financial reporting quality.

The findings of an experiment by Joe and Vandervelde (2007) indicate that although knowledge spillover improves audit risk assessment, auditors seem to be less skeptical in identifying specific factors indicative of fraud when they provide both audit and non-audit services. Rice and Weber (2012) examine a sample of companies with existing internal control weaknesses and find that non-audit fees make it less likely that material weaknesses are disclosed. Their findings therefore support the economic dependence view. Prior research has also investigated market perceptions of non-audit services being provided by incumbent audit firms (e.g., Krishnan et al. 2005; Francis and Ke 2006; Khurana and Raman 2006), and the findings generally suggest that investors perceive the provision of non-audit services impairs auditor independence.

In 2002, the US Congress passed the Sarbanes-Oxley Act (SOX), which brought greater regulation to audit firms, among other things. In order to enhance audit

² As defined by DeAngelo (1981), audit quality consists of both the likelihood that the auditor discovers a breach and the likelihood that the auditor discloses the breach.

quality by securing auditor independence, SOX provisions and Securities and Exchange Commission (SEC) rules prohibit audit firms from offering audit and certain non-audit services to the same client concurrently. Because of the potential benefits from knowledge spillover, providing tax services is permitted, but is hedged with specific requirements. For example, an audit committee must approve in advance all audit firm-provided tax services, tax-related fees paid to the audit firm must be reported separately, and there are limitations to the scope of the tax consulting (SEC 2003; PCAOB 2005).

There are some prior studies examining the effects of tax services on the clients' financial reporting quality. The underlying assumption in most of these studies is that better quality auditing ought to manifest in better quality financial reporting by the client company. Kinney et al. (2004) find a negative association between restatements of financial statements and tax fees. They interpret this result as suggesting that there are benefits from tax consulting that flow to the audit quality. See tharaman et al. (2011) examine the association between restated periods and tax-related fees, but do not find statistically significant results. However, they report a significant negative relation when the restatements relate to tax issues. Choi and Lee (2009) find that tax services being provided by an incumbent audit firm results in lower client discretionary accruals. Huang et al. (2007) find mostly insignificant associations between tax fees and proxies for financial reporting quality (some weak evidence of lower abnormal accruals, but insignificant association with meeting or beating earnings benchmarks). Krishnan and Visvanathan (2011) suggest that tax services have a negative impact on loss avoidance. Gleason and Mills (2011) find improvement in estimates for tax reserves when the audit firm also provides tax services. Harris and Zhou (2013) suggest that tax services lead to lower likelihood of non-tax internal control weak-nesses but do not have an effect on tax-related weaknesses. In addition to the studies examining the client company's financial reporting quality, Robinson (2008) finds a positive association between tax fees and the likelihood that the auditor issues a going concern opinion prior to bankruptcy filing. In summary, these studies support the view of knowledge spillover or find insignificant associations. However, a few prior studies have found indications suggesting that tax fees may have a negative effect on audit quality. Using an experiment, Favere-Marchesi (2006) find that the joint provision of audit and tax services lead to significantly lower fraud-risk assessments. Paterson and Valencia (2011) find that recurring audit firm-provided tax services create knowledge spillover, but nonrecurring tax services seem to have a detrimental impact on auditor independence.

In this study, we approach the issue of tax services being provided by an incumbent audit firm via a sample of companies with poor financial reporting quality, that is, companies with misstatements. The misstatements are determined from restatement data, and the focus is on the restated periods. Rather than examining the years of restatement announcements, it is more appropriate to concentrate on the restated periods including misstatement in order to determine the companyyears with poor accounting quality. First, we investigate whether tax-related fees are associated with those misstatements in financial information that remain undiscovered in a fiscal year in question, to which we refer as restatement lag. Second, we investigate whether tax-related fees are associated with the likelihood of internal control weakness disclosures among companies with misstatements. SOX Section 404 requires the management of listed companies to disclose their assessment on the effectiveness of internal control over financial reporting.³ The underlying assumption is that an occurrence of a misstatement indicates internal control weaknesses (e.g., Eilifsen and Messier 2000; Leone 2007; Rice and Weber 2012). Therefore, utilizing a sample of companies with misstatements (eventually restated), we examine whether tax fees has an impact on whether or not a company disclosed the existing material weaknesses.

A research setting utilizing a sample of companies that all have low financial reporting quality allows the examination of auditors' professional skepticism in the context of the joint provision of tax services and audit services. Given the cost (a threat to auditor independence) and benefit (knowledge spillover) associated with tax services, it is not obvious how tax services affect audit quality. The enhanced knowledge of clients acquired via the provision of tax services could make restatements more timely and material weakness disclosures more likely in companies with poor accounting quality. However, economic dependence might disrupt an auditor's professional skepticism, resulting in restatement lags and unacknowledged control problems. Thus, we state the following two hypotheses without directional expectations:

H1: There is a significant association between restatement lag and incumbent audit firm-provided tax services among companies with poor financial reporting quality.

³ This section became effective for accelerated filers in 2004. In addition to the management assessment, Section 404 requires the company's external auditor to attest the assessment. Since 2007 smaller listed companies have also been reporting their assessment of internal control effectiveness annually, but auditor's internal control disclosure is not required. Thus, our sample includes company-year observations with both auditor internal control reports and management-only internal control reports.

H2: There is a significant association between disclosure of internal control weaknesses and incumbent audit firm-provided tax services among companies with poor financial reporting quality.

3 Data and methodology

3.1 Sample and model specification

The data used in this study consist of company-year observations of US listed companies from 2005–2012. We start the sample selection by gathering from Audit Analytics all the companies located in the US that have issued the SOX Section 404 mandated internal control reports. Then we obtain restatement data and audit fee data (also from Audit Analytics), and combine those with the internal control data. Finally, we extract the financial data from the Thomson Financial database.

The samples used in the analyses consist of companies with poor financial reporting quality based on occurrences of misstatements (eventually restated). Specifically, we examine the restatement disclosure data, focusing on the restated periods, that is, the periods including misstatements. The restatement data include the period beginning and ending dates for which the company is restating. The restatement can affect an entire fiscal year, multiple fiscal years, or only a part of the fiscal year.

In order to investigate the first hypothesis, we estimate the following logit model⁴:

 $REST_LAG = \alpha + \beta_{1}taxservices + \beta_{2}AUDITFEES + \beta_{3}AUDITRELATED + \beta_{4}OTHERFEES + \beta_{5}BIG4 + \beta_{6}GC + \beta_{7}ACHANGE + \beta_{8}LOGTA + \beta_{9}LOGNBS + \beta_{10}FOREIGN + \beta_{11}RESTRUCT + \beta_{12}ACQ + \beta_{13}LOSS + \beta_{14}LEV + \beta_{15}PRIORICW + annual fixed effects + in$ $dustry fixed effects + <math>\varepsilon$

(1)

⁴ Because we are estimating conditional probability, we use logistic analysis, where normal distributions of variables are not required.

The dependent variable in Model (1) is *REST_LAG*, which is a dummy variable equal to one if the restatement announcement took place after the filing date of internal control report, and zero otherwise. If more than one restatement announcement affects the same fiscal-year, *REST_LAG* is coded as one only if each announcement date takes place after the filing date of the internal control report. The rationale in the analyses using Model (1) is to compare company-years where misstatements were restated in a more timely manner to those company-years where misstatements were restated only at the later periods (restatement lag). We assume that better audit quality would manifest in a reduced likelihood of restatement lags.

In order to investigate the second hypothesis, we estimate the following logit model:

 $MW404 = \alpha + \beta_{1}taxservices + \beta_{2}AUDITFEES + \beta_{3}AUDITRELATED + \beta_{4}OTHERFEES + \beta_{5}BIG4 + \beta_{6}GC + \beta_{7}ACHANGE + \beta_{8}LOGTA + \beta_{9}LOGNBS + \beta_{10}FOREIGN + \beta_{11}RESTRUCT + \beta_{12}ACQ + \beta_{13}LOSS + \beta_{14}LEV + \beta_{15}PRIORICW + annual fixed effects + industry fixed effects + \varepsilon$ (2)

The dependent variable in Model (2) is *MW404*, which is a dummy variable equal to one if a company disclosed an internal control report indicating material weaknesses, and zero otherwise. Since the sample used in the analyses consists of company-years with misstatements, material weakness disclosures could be expected to indicate that control problems have been acknowledged. The data on section 404 material weaknesses consists of both auditor internal control reports and management-only reports (see also footnote 3). The assumption is that, in case of high quality auditing, auditors would require management to acknowledge existing material weaknesses also when they do not issue their own internal control report. Additionally, we use the variable *MW404FIRST*, which is set to one if a company discloses internal control material weaknesses for the first time, and zero otherwise.

Both Model (1) and Model (2) include the same explanatory variables. *Tax-services* represents the four alternative variables for tax-related fees, which are examined separately in the analyses: *DTAXFEES* is set as one, if tax-related fees are greater than zero, and zero otherwise; *TAXFEES* measures the magnitude of tax fees, and is calculated by dividing tax-related fees paid to the incumbent auditor by the square root of total assets (see Kinney et al. 2004); *TAXFEES_TF* is the

ratio of tax-related fees paid to the incumbent auditor divided by total fees⁵; *TAXFEES_AF* is the ratio of tax-related fees paid to the incumbent auditor divided by audit fees.

The control variables are adapted from prior studies investigating the likelihood of restated financial statements (Seetharaman et al. 2011) and the non-disclosure of existing material weaknesses (Rice and Weber 2012). AUDITFEES controls for the effect of audit effort, and is calculated as audit fees scaled by the square root of total assets (see e.g., Kinney et al. 2004; Rice and Weber et al. 2012). Similarly, audit related fees (AUDITRELATED) and all other auditor fees (OTHER-*FEES*) are scaled by the square root of total assets. To control for the effect of audit firm size, an indicator variable for the Big 4 audit firms (BIG4) is included in the model (see e.g., Seetharaman et al. 2011). Indicator variables for going concern opinion (GC) and for change of the signatory audit firm from previous year (ACHANGE) are also included in the model. Company size is controlled by including the natural logarithm of total assets (LOGTA) in the model (e.g., See tharaman et al. 2011). Complex companies might have more difficulties to establish and maintain higher quality accounting system. To control for the effect of company complexity, we include the natural logarithm of the number of business segments (LOGNBS) in the model (e.g., Seetharaman et al. 2011). Moreover, we include indicator variables for companies reporting geographic segment(s) other than the USA (FOREIGN), restructuring expenses/charges (RESTRUCT) and exemptions to the internal control assessment due to acquisition (ACQ).⁶ Poorly performing companies may have more incentive to manage earnings, and/or have less resource to maintain high quality accounting systems. Thus, we add to the model an indicator variable for companies reporting negative income in either previous or current company year (LOSS) to control for financial performance. Highly leveraged companies might try to avoid violating debt covenants by managing earnings (e.g., Seetharaman et al. 2011), but those companies might also be subject to closer monitoring leading to more conservative reporting (e.g., Seetharaman et al. 2011). Thus, we include leverage (LEV) calculated as total debt divided by total assets in the model.⁷ Finally, we control for the disclosure of the previous year's internal control effectiveness (PRIORICW). PRIORICW is set to one if a company identified internal control weaknesses in the previous year based on either Section 404 disclosure or Section 302 quarterly disclosure. SOX

⁵ Total fees is the sum of audit fees, audit related fees, tax-related fees, and other fees.

⁶ This information is based on Audit Analytics data where it is indicated if exemptions to the assessment of internal controls over financial reporting were identified.

⁷ Observations with *LEV*>1 are excluded to mitigate outliers.

Section 302 relates to quarterly reporting and requires management to assess the effectiveness of disclosure controls and procedures. Finally, year and industry (according to the 17 industry classifications by Fama and French) fixed effects control for temporal variation and industry differences. All the variables used in the analyses are defined in Table 1.

Table 1.Variable de	finitions
---------------------	-----------

Variable	Definition
REST_LAG	A dummy variable with a value of one if the restatement announcement took place after the filing date of internal control opinion, and zero otherwise.
MW404	internal controls over financial reporting based on Section 404, and zero otherwise.
MW404FIRST	A dummy variable with a value of one if a company disclosed weaknesses in internal controls over financial reporting based on Section 404 for the first time, and zero otherwise.
DTAXFEES	A dummy variable with a value of one if a company paid tax related fees to the auditor, zero otherwise.
TAXFEES	Tax related fees paid to the incumbent auditor scaled by the square root of total assets.
TAXFEES_TF	Tax related fees paid to the incumbent auditor scaled by total fees.
TAXFEES_AF	Tax related fees paid to the incumbent auditor scaled by audit fees.
AFEES	Audit fees scaled by the square root of total assets.
AUDITRELATED	Audit related fees scaled by the square root of total assets.
OTHERFEES	Other fees scaled by the square root of total assets.
BIG4	A dummy variable with a value of one if the auditor is one of the four larg- est audit firms, and zero otherwise.
GC	A dummy variable with a value of one if a company received a going con- cern opinion, and zero otherwise.
ACHANGE	A dummy variable with a value of one if the signing audit firm changed from the previous fiscal year, and zero otherwise.
LOGTA	The natural logarithm of total assets
LOGNBS	The natural logarithm of the number of business segments reported by the company.
FOREIGN	A dummy variable with a value of one if a company reported geographic segment(s) other than the US, and zero otherwise.
RESTRUCT	A dummy variable with a value of one if a company reported restructuring expenses / charges, and zero otherwise.
ACQ	A dummy variables with a value of one if an exemption to the assessment of internal controls indicates acquisition(s) during the past year, and zero otherwise.
LOSS	A dummy variable with a value of one if the net income before extraordi- nary items is negative in year t or t-1, and zero otherwise.
LEV	Total debt divided by total assets
PRIORICW	A dummy variable with a value of one if the company disclosed either Sec- tion 404 material weaknesses or Section 302 weaknesses in the previous year, and zero otherwise.
YEAR	A dummy variable for the fiscal years 2005–2012
INDUSTRY	A dummy variable for Fama and French 17 industry classifications

The analyses are conducted using 1) total sample of restated company-years and 2) sample of company-years where only the first restated years are included. The number of observations in the samples is as follows⁸:

- 1) Total samples of restated company-years
 - Model (1) used to examine the probability of a restatement lag: 2,749 observations (1,343 unique companies)
 - Model (2) used to examine the probability of a material weakness disclosure: 2,584 observations (1,312 unique companies). If a restated internal control report was issued and the material weaknesses were stated only in the restated report, the observation is excluded from the sample.
 - Additionally, we examine the probability of a first time material weakness disclosure: 2,262 observations (1,174 unique companies). If an internal control report indicating material weaknesses is issued repeatedly (i.e., other than a first time material weakness report), the observation is excluded from the sample.
- 2) Sample of restated company-years: only first restated years are included. A restatement may affect multiple fiscal years. We also conduct the analyses using a sample where continuous misstatement years have been excluded and examine only the first restated fiscal years.
 - Model (1) used to examine the probability of a restatement lag: 1,374 observations (1,096 unique companies)
 - Model (2) used to examine the probability of a material weakness disclosure: 1,297 observations (1,051 unique companies)
 - Additional investigation of first time material weaknesses: 1,131 observations (941 unique companies)

⁸ Exclusions from the sample are primarily related to missing values on variables needed in the analyses, but also to some extreme cases or possible data errors in financial dates (which are used to determine years including misstatements) and inconsistencies in auditor names between datasets.

3.2 Descriptive statistics

The descriptive statistics of the variables used in the empirical analyses are presented in Panel A of Table 2. The mean of the dependent variable REST_LAG indicates that 63 percent of the sample observations are restatement lag observations. Moreover, 21 percent of the sample observations include 404 material weaknesses and 10 percent first time material weaknesses. The dummy variable for tax fees (DTAXFEES) indicates that 70 percent of the company-year observations include incumbent auditor-provided tax services. The means (medians) of tax fees scaled by the square root of total assets (TAXFEES) is 0.165 (0.045), the ratio of tax fees to total fees (TAXFEES_TF) is 0.074 (0.039), and the ratio of tax fees to audit fees (TAXFEES_AF) is 0.105 (0.044). The means (medians) of audit fees divided by the square root of total assets (AFEES) is 1.596 (1.263), audit related fees divided by the square root of total assets (AUDITRELATED) is 0.124 (0.027), and other fees divided by the square root of total assets (OTHERFEES) is 0.028 (0.000). Seventy-three percent of the company-years are audited by one of the Big 4 audit firms, seven percent of the sample observations have a going concern opinion, and approximately eight percent have changed their signatory auditor from the previous year.

The average value of total assets is 9,730 million dollars with a median value of 735 million dollars. The average number of business segments is 2.625. Approximately 46 percent of the observations have foreign operations, 24 percent have reported restructuring expenses/charges, and 10 percent indicate an exception to the assessment of internal controls due to acquisition. In about 49 percent of the company-years, the companies report a loss in either the current or previous year. The mean (median) leverage is 0.247 (0.197). On average, 23 percent of the sample observations have disclosed internal control weaknesses based on section 404 disclosure, or section 302 disclosure in the previous year.

Panel B of Table 2 displays the Pearson correlation coefficients among the selected variables. The two highest correlations among the independent variables occur between *BIG4* and *LOGTA* (0.563), and *TAXFEES* and *AUDITRELATED* (0.462). The variance inflation factors (untabulated) do not indicate problems with multicollinearity.

Table 2.Descriptive statistics

Variable	Mean	Std.	Min	25%tile	Median	75%tile	Max	Ν
REST_LAG	0.628	0.483	0.000	0.000	1.000	1.000	1.000	2,749
MW404	0.214	0.410	0.000	0.000	0.000	0.000	1.000	2,584
MW404FIRST	0.103	0.303	0.000	0.000	0.000	0.000	1.000	2,262
DTAXFEES	0.696	0.460	0.000	0.000	1.000	1.000	1.000	2,749
TAXFEES	0.165	0.384	0.000	0.000	0.045	0.192	8.818	2,749
TAXFEES_TF	0.074	0.094	0.000	0.000	0.039	0.115	0.553	2,749
TAXFEES_AF	0.105	0.164	0.000	0.000	0.044	0.142	1.693	2,749
AFEES	1.596	1.379	0.029	0.717	1.263	2.052	24.417	2,749
AUDITRELATED	0.124	0.358	0.000	0.000	0.027	0.118	7.727	2,749
OTHERFEES	0.028	0.239	0.000	0.000	0.000	0.002	7.115	2,749
BIG4	0.730	0.444	0.000	0.000	1.000	1.000	1.000	2,749
GC	0.069	0.253	0.000	0.000	0.000	0.000	1.000	2,749
ACHANGE	0.084	0.277	0.000	0.000	0.000	0.000	1.000	2,749
Total assets (millions)	9729.710	96053.390	0.003	182.666	735.445	2728.510	3222422.000	2,749
Segments	2.625	1.809	1.000	1.000	2.000	4.000	10.000	2,749
FOREIGN	0.458	0.498	0.000	0.000	0.000	1.000	1.000	2,749
RESTRUCT	0.242	0.428	0.000	0.000	0.000	0.000	1.000	2,749
ACQ	0.101	0.302	0.000	0.000	0.000	0.000	1.000	2,749
LOSS	0.492	0.500	0.000	0.000	0.000	1.000	1.000	2,749
LEV	0.247	0.230	0.000	0.040	0.197	0.384	0.996	2,749
PRIORICW	0.225	0.418	0.000	0.000	0.000	0.000	1.000	2,749

Panel A: Distributional statistics

	2	3	4	5	9	7	8	6	10	11	12	13	14	15	16	17	18
DTAXFEES (1)	0.284	0.525	0.425	0.081	0.054	0.028	0.222	-0.155	-0.171	0.262	0.158	0.194	0.098	0.034	-0.101	0.000	-0.096
TAXFEES (2)		0.627	0.709	0.299	0.462	0.287	0.141	-0.055	-0.068	0.105	0.120	0.174	0.141	0.025	0.004	-0.013	-0.044
TAXFEES_TF (3)			0.943	-0.018	0.055	0.023	0.121	-0.089	-0.086	0.136	0.093	0.097	0.089	0.014	-0.076	-0.002	-0.113
TAXFEES_AF (4)				-0.014	0.235	0.163	0.096	-0.078	-0.071	0.123	0.091	0.079	0.078	0.012	-0.056	0.000	-0.101
AFEES (5)					0.171	0.068	0.184	0.016	-0.051	-0.067	0.123	0.378	0.271	0.079	0.145	-0.088	0.214
AUDITRELATED (6)						0.415	0.044	-0.016	-0.017	0.056	0.074	0.076	0.064	0.055	0.027	0.035	-0.010
OTHERFEES (7)							0.056	-0.027	-0.027	0.114	0.049	0.070	0.041	0.018	-0.005	-0.004	0.015
BIG4 (8)								-0.321	-0.272	0.563	0.251	0.257	0.218	0.115	-0.204	0.082	-0.191
GC (9)									0.183	-0.458	-0.166	-0.157	-0.063	-0.091	0.256	0.063	0.194
ACHANGE (10)										-0.253	-0.104	-0.104	-0.069	-0.054	0.131	0.002	0.139
LOGTA (11)											0.370	0.166	0.122	0.122	-0.344	0.175	-0.230
LOGNBS (12)												0.181	0.128	0.107	-0.148	0.077	-0.054
FOREIGN(13)													0.341	0.121	-0.068	-0.146	-0.025
RESTRUCT (14)														0.066	0.072	-0.020	-0.013
ACQ (15)															-0.090	0.005	-0.054
LOSS (16)																0.085	0.166
LEV (17)																	-0.006
PRIORICW (18)																	

ign

3.3 Univariate tests

Panel A of Table 3 reports the univariate results when the sample is partitioned into two groups based on the restatement lag variable (REST_LAG). The mean differences examined are DTAXFEES, TAXFEES, TAXFEES TF, and TAXFEES_AF. The statistical significances are based on t-tests for continuous variables and a chi-square test for dummy variables. The descriptive test results for the total sample indicate that companies with restatement lags are more likely to have tax fees (DTAXFEES has a marginally significant difference at the 0.10 level), and have higher proportions of tax fees to total fees (TAXFEES_TF) and tax fees to audit fees (TAXFEES_AF) (significant at the 0.01 level) compared to companies with misstatements discovered before filing the internal control report. Moreover, the results of a sample that includes only the first restated years indicate that companies with restatement lags have higher tax-related fees (TAXFEES, TAXFEES_TF, and TAXFEES_AF).

Panel B of Table 3 reports the univariate results when the sample is partitioned into two groups based on the material weakness disclosure variable (*MW404*). These descriptive results indicate that companies disclosing material weaknesses have significantly lower tax-related fees compared to those that do not acknowledge their control problems. In the total sample, the differences are significant with the variables *DTAXFEES*, *TAXFEES_TF*, and *TAXFEES_AF*. In the sample that includes only the first restated years, all the variables of tax-related fees are significantly different between companies that disclose weaknesses and those that do not disclose them.

Table 3.Univariate results

	Total sampl compar	e of restated 1y-years		Sample of restated only first restate	l company-years: d year included	
	<u>REST_LAG=1</u> <u>n=1,727</u>	<u>REST_LAG=0</u> <u>n=1,022</u>		<u>REST_LAG=1</u> <u>n=908</u>	<u>REST_LAG=0</u> <u>n=466</u>	
DTAXFEES	0.708	0.674	*	0.666	0.663	
TAXFEES	0.170	0.156		0.161	0.126	**
TAXFEES_TF	0.080	0.066	***	0.074	0.062	**
TAXFEES_AF	0.114	0.090	***	0.107	0.082	***

Panel A: Restatement lags

Panel B: Section 404 material weaknesses

	<u>Total sample</u> <u>company</u>	of restated -years		Sample of restated of nestated of nestated of sectors of the sector of t	company-years: year included	
	<u>MW404=1</u> <u>n=554</u>	<u>MW404=0</u> <u>n=2,030</u>		$\frac{\text{MW404}=1}{\text{n}=281}$	<u>MW404=0</u> <u>n=1,016</u>	
DTAXFEES	0.610	0.724	***	0.577	0.696	***
TAXFEES	0.153	0.172		0.111	0.164	***
TAXFEES_TF	0.054	0.080	***	0.047	0.078	***
TAXFEES_AF	0.075	0.114	***	0.064	0.110	***

Notes: See Table 1 for the variable definitions. The statistical significances are based on t-tests for continuous variables and chi-square test for dummy variables.

4 Results

We estimate Model (1) to test whether tax services are associated with the likelihood of a restatement lag, and Model (2) to test whether tax services are associated with the likelihood of a material weakness disclosure. The logistic regression results are reported in Table 4. Panel A of Table 4 reports the results for the dummy variable of tax services (*DTAXFEES*). These results indicate that in each model specification (*REST_LAG*, *MW404*, *MW404FIRST*), the indicator variable for tax-related fees is insignificant, suggesting that tax services being provided by incumbent audit firms is not associated with audit quality.

Panel B of Table 4 reports the results for the continuous variable *TAXFEES*, which represents the magnitude of tax-related fees. Columns (1) and (4) indicate that tax-related fees are not associated with restatement lags. However, *TAXFEES* is marginally significantly associated with lower likelihood of material weakness disclosures in the sample that includes only the first restated years [column (5)]. Moreover, higher tax-related fees are negatively associated with the likelihood of first time material weakness disclosures [columns (3) and (6)].
Panel C of Table 4 reports the results for the variable tax-related fees to total fees (*TAXFEES_TF*). These results show that *TAXFEES_TF* is not significantly associated with the likelihood of restatement lags [columns (1) and (4)]. However, the results indicate that the proportion of tax-related fees of total fees is significantly and negatively associated (at the 0.05 level) with the likelihood of material weakness disclosures (columns (2), (3), (5), and (6)].

Panel D of Table 4 shows the results for the ratio of tax-related fees to audit fees. *TAXFEES_AF* is significantly (at the 0.05 level) and positively associated with the likelihood of restatement lags, but only in the total sample of misstatements [column (1)]. The results in columns (3) and (6) suggest that the likelihood of first time material weakness disclosures is lower when tax-related fees are higher.

Panel A: The indicate	or variat	le for tax	-relat	ted fee	S>0 (L	XAX	TEES)										
		Total	sampl	e of res	tated co	mpan	y-years			Sample c	f restated c	ompany-y	ears: onl	y first	restated y	ear inclu	Ided
		(1)		Ŭ	2)			(3)			(4)		(5)			(9)	
Dependent variable	RES	ST_LAG		MV	V404		MW4	04FIRS1	r	RES	T_LAG		MW404		MW4	04FIRS7	<u> </u>
	Est.	Wald		Est.	Wald		Est.	Wald		Est.	Wald	Est.	Wald		Est.	Wald	
DTAXFEES	-0.063	0.44	9	066	0.25		-0.226	1.75		-0.203	2.04	-0.103	0.30		-0.209	0.80	
AUDITFEES	-0.196	20.52 **	.* ()	.417	51.52	***	0.466	40.27	* * *	-0.264	17.73 ***	0.339	17.68	* * *	0.247	6.48	* * *
AUDITRELATED	0.150	1.39	9	0.122	0.57		-0.009	0.00		0.455	5.32 **	-0.213	0.76		-0.226	0.26	
OTHERFEES	-0.196	0.75	0	0.032	0.03		-0.037	0.02		-0.130	0.20	-0.018	0.00	_	-1.500	1.41	
BIG4	0.452	11.69 **	- *	.845	27.07	***	-0.841	13.96	* * *	0.922	26.40 ***	-0.976	17.96	* * *	-1.023	9.42	* * *
GC	-0.159	0.72	0	0.601	6.61	*	0.381	1.21		0.113	0.21	0.225	0.52		0.193	0.19	
ACHANGE	-0.518	9.85 **	.* **	.886	22.67	***	1.083	20.65	* * *	-0.548	7.37 ***	1.044	16.00	***	1.092	12.10	* * *
LOGTA	-0.014	0.24	9	0.083	4.35	*	-0.121	4.92	* *	-0.044	1.14	-0.059	1.01		-0.049	0.37	
LOGNBS	0.095	2.13	9	0.103	1.22		-0.036	0.09		0.133	1.67	0.122	0.76		0.126	0.47	
FOREIGN	0.188	3.17 *	9	.348	5.32	*	-0.384	3.92	* *	0.244	2.21	-0.501	4.95	* *	-0.221	0.62	
RESTRUCT	0.038	0.11	9	660.0	0.39		-0.107	0.28		-0.268	2.66	-0.064	0.08		-0.025	0.01	
ACQ	0.146	0.99	0	0.028	0.02		0.223	0.89		0.098	0.21	0.207	0.52		0.339	1.03	
TOSS	-0.214	4.93 **	0	.434	10.13	***	0.484	7.39	* *	-0.519	13.46 ***	0.846	18.04	* *	0.852	9.64	* * *
LEV	0.221	1.26	9	600.0	0.00		-0.114	0.10		0.205	0.50	-0.493	1.71		-0.468	0.88	
PRIORICW	-0.908	68.89 **	** 1	.113	77.44	***	-1.963	31.30	* *	-0.409	5.70 **	1.576	68.99	* * *	-1.247	7.42	* * *
Intercept	0.088	0.17	9	.854	8.28	* **	-1.326	11.19	* *	0.528	2.85 *	-1.180	8.09	* * *	-1.790	9.85	* * *
Annual fixed effects?	Yes			Yes			Yes			Yes		Yes			Yes		
Industry fixed effects?	Yes			Yes			Yes			Yes		Yes			Yes		
Likelihood ratio																	
Chi-Square	353.5	<.0001		612.1 <	<.0001		257.2	<.0001		185.4	<.0001	373.	4 <.0001		119.7	<.0001	
\mathbb{R}^2		0.12			0.21			0.11			0.13		0.25			0.10	
Ν		2,749			2,584			2,262			1,374		1,297			1,131	

Multivariate results Table 4. Acta Wasaensia 121

I ULVI D. I IN IIIUGUII		nonini vini	777 1777									
		Total sa	imple of re	sstated company	y-years	-	Sample o	of restated cor	npany-yea	ars: only first	restated y	ear included
		(1)		(2)		(3)		(4)		(5)		(9)
Dependent variable	RES	T_LAG	Z	1W404	MW4	04FIRST	RES	T_LAG	M	W404	MW4)4FIRST
	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald
TAXFEES	0.132	1.16	-0.205	1.46	-0.394	3.16 *	0.389	2.22	-0.880	3.74 *	-1.970	6.15 **
AUDITFEES	-0.206	20.89 ***	0.431	49.89 ***	0.494	42.78 ***	-0.292	18.91 ***	0.376	17.79 ***	0.318	8.12 ***
AUDITRELATED	0.090	0.37	-0.031	0.03	0.192	0.98	0.355	2.85	-0.153	0.31	-0.204	0.21
OTHERFEES	-0.214	0.77	0.077	0.16	0.114	0.26	-0.098	0.12	-0.032	0.01	-1.603	1.51
BIG4	0.447	11.41 ***	-0.845	27.05 ***	-0.857	14.60 ***	0.904	25.52 ***	-0.953	17.09 ***	-0.980	8.67 ***
GC	-0.158	0.70	0.605	6.69 ***	0.386	1.24	0.113	0.21	0.231	0.55	0.239	0.30
ACHANGE	-0.506	9.52 ***	0.893	23.13 ***	1.102	21.05 ***	-0.507	6.46 **	1.035	15.89 ***	1.098	12.15 ***
LOGTA	-0.017	0.38	-0.082	4.16 **	-0.120	4.78 **	-0.053	1.68	-0.057	0.93	-0.051	0.39
LOGNBS	0.091	1.98	-0.102	1.21	-0.033	0.07	0.125	1.48	0.128	0.84	0.148	0.64
FOREIGN	0.181	2.97 *	-0.357	5.64 **	-0.407	4.45 **	0.225	1.87	-0.506	5.08 **	-0.229	0.66
RESTRUCT	0.036	0.10	-0.101	0.40	-0.110	0.30	-0.274	2.75 *	-0.056	0.06	0.035	0.01
ACQ	0.156	1.12	0.013	0.00	0.196	0.70	0.122	0.32	0.198	0.47	0.350	1.08
LOSS	-0.213	4.89 **	0.437	10.27 ***	0.496	7.75 ***	-0.504	12.78 ***	0.827	17.06 ***	0.763	7.86 ***
LEV	0.223	1.28	-0.007	0.00	-0.094	0.07	0.197	0.46	-0.478	1.60	-0.425	0.72
PRIORICW	-0.898	67.49 ***	1.104	75.72 ***	-2.010	31.49 ***	-0.384	4.98 **	1.552	65.75 ***	-1.305	7.70 ***
Intercept	0.072	0.12	-0.896	9.34 ***	-1.456	13.81 ***	0.459	2.20	-1.207	8.41 ***	-1.798	10.07 ***
Annual fixed effects?	Yes		Yes		Yes		Yes		Yes		Yes	
Industry fixed effects?	Yes		Yes		Yes		Yes		Yes		Yes	
Likelihood ratio												
Chi-Square	354.1	<.0001	613.	5 < .0001	259.1	<.0001	185.9	<.0001	378.4	<.0001	128.5 -	<.0001
\mathbb{R}^2		0.12		0.21		0.11		0.13		0.25		0.11
Z		2,749		2,584		2,262		1,374		1,297		1,131

T MILLI C. TITL TMILL OT	TOT VINI		AT THINK		(1								
		Total s	ample of re	estated compa	ny-years		Sample (of restated con	npany-ye	ars: only fir	st restated	year include	led
		(1)		(2)		(3)		(4)		(5)		(9)	
Dependent variable	RES	T_LAG	N	[W404	MW4	04FIRST	RES	T_LAG	Μ	W404	MW	404FIRST	
	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald	
TAXFEES_TF	0.700	2.40	-1.364	4.31 **	-2.350	5.28 **	0.592	0.71	-2.356	5.25 **	-3.981	5.47 **	*
AUDITFEES	-0.194	20.12 ***	0.412	50.44 ***	0.458	40.30 ***	* -0.268	17.74 ***	0.333	17.00 **:	• 0.238	5.85 **	*
AUDITRELATED	0.135	1.13	-0.099	0.40	0.047	0.06	0.428	4.73 **	-0.216	0.73	-0.256	0.30	
OTHERFEES	-0.186	0.66	0.029	0.02	-0.044	0.02	-0.102	0.12	-0.040	0.02	-1.651	1.62	
BIG4	0.443	11.21 ***	-0.839	26.33 ***	-0.839	13.82 ***	* 0.911	26.03 ***	-0.968	17.67 ***	-1.000	9.07 **	*
GC	-0.156	0.68	0.592	6.45 **	0.380	1.22	0.120	0.24	0.211	0.46	0.217	0.25	
ACHANGE	-0.499	9.22 ***	0.875	22.14 ***	1.082	20.48 ***	* -0.503	6.37 **	1.013	15.22 **:	• 1.071	11.69 **	*
LOGTA	-0.017	0.38	-0.081	4.10 **	-0.119	4.70 **	-0.052	1.60	-0.056	0.93	-0.044	0.29	
LOGNBS	0.089	1.90	-0.098	1.13	-0.033	0.07	0.126	1.51	0.125	0.80	0.137	0.54	
FOREIGN	0.177	2.81 *	-0.357	5.65 **	-0.396	4.24 **	0.223	1.83	-0.511	5.18 **	-0.222	0.62	
RESTRUCT	0.031	0.08	-0.092	0.34	-0.095	0.22	-0.268	2.64	-0.062	0.08	-0.003	0.00	
ACQ	0.153	1.09	0.014	0.00	0.202	0.75	0.117	0.30	0.195	0.46	0.329	0.97	
TOSS	-0.211	4.79 **	0.427	9.84 ***	0.467	6.91 ***	* -0.510	13.08 ***	0.821	16.83 **:	• 0.785	8.23 **	*
LEV	0.222	1.28	-0.008	0.00	-0.106	0.09	0.193	0.44	-0.461	1.50	-0.442	0.80	
PRIORICW	-0.898	67.50 ***	1.104	75.99 ***	-1.981	31.54 ***	* -0.397	5.33 ***	1.563	68.01 **:	-1.259	7.50 **	*
Intercept	0.021	0.01	-0.800	7.53 ***	-1.295	11.08 ***	* 0.419	1.84	-1.080	6.70 **:	-1.668	9.01 **	*
Annual fixed effects?	Yes		Yes		Yes		Yes		Yes		Yes		
Industry fixed effects?	Yes		Yes		Yes		Yes		Yes		Yes		
Likelihood ratio													
Chi-Square	355.2	<.0001	616.1	<.0001	261.6	<.0001	184.0	<.0001	378.2	<.0001	126.1	<.0001	
\mathbb{R}^2		0.12		0.21		0.11		0.13		0.25		0.11	
N		2,749		2,584		2,262		1,374		1,297		1,131	

Panel C: The ratio of tax-related fees to total fees (TAXFEES TF)

Panel D: The ratio of	tax-relat	ted fees	to al	udit fee	s (TAX	FEE	S_AF									
		Total	samp	ole of res	tated cor	npany	-years			Sample (of restated co	mpany-ye	ars: only fir	st restated	year included	
	Ŭ	(1)			(2)			(3)			(4)		(5)		(9)	
Dependent variable	REST	[_LAG		M	W404		MW4	04FIRS	E	RES	T_LAG	2	IW404	MM	404FIRST	
	Est.	Wald		Est.	Wald		Est.	Wald		Est.	Wald	Est.	Wald	Est.	Wald	
TAXFEES_AF	0.570	4.56	*	-0.644	2.54		-1.330	4.59	* *	0.594	2.08	-0.992	1.90	-2.286	3.49 *	
AUDITFEES	-0.192	19.69	* **	0.412	50.62	***	0.458	40.27	* * *	-0.266	17.49 ***	0.332	17.22 ***	0.234	5.81 **	
AUDITRELATED	0.087	0.46		-0.063	0.15		0.114	0.36		0.384	3.73 *	-0.171	0.43	-0.211	0.21	
OTHERFEES	-0.210	0.74		0.050	0.07		0.041	0.02		-0.102	0.12	-0.021	0.01	-1.593	1.53	
BIG4	0.444	11.21	* *	-0.844	26.80	***	-0.846	14.10	* * *	0.909	25.86 ***	-0.970	17.83 ***	-1.007	9.18 ***	
GC	-0.155	0.67		0.594	6.47	*	0.379	1.21		0.120	0.24	0.217	0.49	0.210	0.24	
ACHANGE	-0.498	9.23	* *	0.881	22.46	* * *	1.088	20.65	* * *	-0.502	6.36 ***	1.033	15.76 ***	1.082	11.86 ***	
LOGTA	-0.018	0.41		-0.082	4.18	*	-0.118	4.68	* *	-0.052	1.62	-0.058	0.99	-0.046	0.31	
LOGNBS	0.088	1.85		-0.100	1.17		-0.034	0.08		0.125	1.47	0.124	0.78	0.135	0.53	
FOREIGN	0.177	2.83	~	-0.358	5.66	*	-0.401	4.35	* *	0.222	1.83	-0.511	5.21 **	-0.229	0.66	
RESTRUCT	0.029	0.07		-0.094	0.35		-0.102	0.26		-0.270	2.67	-0.063	0.08	-0.012	0.00	
ACQ	0.157	1.14		0.015	0.01		0.201	0.74		0.121	0.32	0.197	0.47	0.332	0.99	
LOSS	-0.212	4.82	*	0.432	10.04	***	0.478	7.23	* * *	-0.508	12.95 ***	0.830	17.29 ***	0.799	8.55 ***	
LEV	0.226	1.32		-0.012	0.00		-0.103	0.08		0.190	0.43	-0.476	1.60	-0.440	0.80	
PRIORICW	-0.898	67.62	* *	1.108	76.72	* * *	-1.980	31.50	* * *	-0.392	5.19 **	1.563	68.04 ***	-1.256	7.50 ***	
Intercept	0.016	0.01		-0.827	8.09	* * *	-1.339	11.89	* * *	0.405	1.72	-1.135	7.60 ***	-1.710	9.59 ***	
Annual fixed effects?	Yes			Yes			Yes			Yes		Yes		Yes		
Industry fixed effects?	Yes			Yes			Yes			Yes		Yes		Yes		
Likelihood ratio																
Chi-Square	357.1	<.0001		614.5	<.0001		260.6	<.0001		185.2	<.0001	375.6	<.0001	124.3	<.0001	
\mathbb{R}^2		0.12			0.21			0.11			0.13		0.25		0.10	
Ν		2,749			2,584			2,262			1,374		1,297		1,131	
<i>Notes:</i> See Table 1 for vari respectively. Statistical signi are not reported.	able defini ficances ar	tions. Stat e calculate	istical d by 6	l significa clustering	nces base the stand	od on 1 ard err	two-tailed ors within	tests at compar	the 1 nies (P	percent, 5 etersen 20	percent, and 99). For brevit	10 percen y, the resu	t levels are d ts for the indu	enoted by * 1stry and ye	**, **, and * ar fixed effect	

Additionally, we estimate Model (1) and Model (2) for only those companies that have paid tax-related fees to their audit firm. That is, we exclude from the sample those observations where tax-related fees are zero. The results are reported in Table 5. These results indicate that the magnitude of tax-related fees is associated with a greater likelihood of restatement lags: *TAXFEES* is significant at the 0.05 level [column (4) of Panel A]; *TAXFEES_TF* is significant at the 0.10 level [columns (1) and (4) of Panel B]; *TAXFEES_AF* is significant at the 0.05 level [columns (1) and (4) of Panel C]. Moreover, the magnitude of tax-related fees is associated with a lower likelihood of material weakness disclosures: *TAXFEES_TF* is significant at the 0.10 level [columns (3), (5), (6) of Panel A]; *TAXFEES_TF* is significant at the 0.10 level [columns (2), (3), (6) of Panel B] and significant at the 0.10 level [column (3)].

In summary, the current research does uncover some evidence suggesting that companies paying higher tax-related fees to their audit firm are more likely to have a restatement lag on a misstated financial statement. The results show more consistently across different model specifications that tax-related fees are associated with a lower likelihood of material weakness disclosures. Especially the higher proportion of tax fees over total fees or audit fees appears to have a significant influence on the material weakness disclosures. However, providing taxrelated services does not in itself affect the likelihood of restatement lags or material weakness disclosures. Rather it is the magnitude of the fees that counts.

The results of the control variables indicate that *AUDITFEES* is significantly and negatively (positively) associated with the likelihood of restatement lag (material weakness disclosure) in nearly every model specification, which suggests that greater audit effort leads to a greater probability that misstatements are restated and control problems acknowledged (e.g., Blankley et al. 2012). Surprisingly, the dummy variable for Big 4 audit firms is positive (negative) in restatement lag (material weakness disclosure) estimations. Auditor change from the previous year is negatively (positively) associated with restatement lag (material weakness disclosure). The indicator variable *LOSS* is negatively (positively) associated with restatement lag (material weakness disclosure) is less likely (more likely) if control problems have been acknowledged in the previous year (*PRIORICW*). Other control variables have greater variations in their significance levels in different model specifications.

(DTAXFEES=1)
\sim
tax-related fees
with a
observations
Only
tests: (
Additional
Table 5 .

Panel A: The magnitude of tax-related fees (TAXFEES)

0												
		Total sai	mple of re	stated compa	ny-years		Sample of	of restated co	mpany-ye	ars: only firs	st restated	year included
		(1)		(2)		(3)		(4)		(5)		(9)
Dependent variable	RES	T_LAG	Μ	W404	MW4	04FIRST	RES	T_LAG	Μ	W404	MW ²	404FIRST
	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald	Est.	Wald
TAXFEES	0.195	1.91	-0.234	1.97	-0.372	2.83 *	0.671	4.17 **	-1.007	3.19 *	-1.957	3.62 *
AUDITFEES	-0.226	17.41 ***	0.455	40.85 ***	0.521	30.96 ***	-0.318	13.66 ***	0.405	16.87 ***	0.364	4.26 **
AUDITRELATED	0.159	0.95	-0.054	0.09	0.212	0.70	0.353	2.12	0.012	0.00	-0.065	0.03
OTHERFEES	-0.455	3.82 *	0.337	1.93	0.273	1.10	-0.951	1.08	0.766	0.40	0.581	0.14
BIG4	0.551	9.39 ***	-1.213	31.61 ***	-1.248	19.35 ***	1.081	19.80 ***	-1.442	20.37 ***	-1.777	16.99 ***
GC	-0.606	4.10 **	0.634	3.36 *	0.381	0.46	-0.469	1.58	0.540	1.33	0.538	0.54
ACHANGE	-0.438	3.14 *	0.483	2.76 *	0.587	2.32	-0.315	0.97	0.242	0.26	0.315	0.31
LOGTA	-0.056	2.36	-0.049	0.94	-0.062	0.73	-0.088	2.46	0.002	0.00	0.035	0.10
LOGNBS	0.163	3.99 **	-0.078	0.48	-0.132	0.70	0.069	0.26	0.352	3.51 *	0.351	1.77
FOREIGN	0.086	0.42	-0.408	4.77 **	-0.575	5.08 **	0.102	0.22	-0.817	6.74 ***	-0.501	1.28
RESTRUCT	0.127	0.90	-0.113	0.36	-0.042	0.03	-0.153	0.52	-0.145	0.26	-0.199	0.25
ACQ	0.272	2.49	-0.185	0.56	-0.028	0.01	0.488	3.43 *	0.105	0.08	0.392	0.76
TOSS	-0.259	4.54 **	0.523	8.99 ***	0.566	6.11 **	-0.571	10.46 ***	0.976	12.78 ***	0.988	7.39 ***
LEV	0.181	0.50	0.144	0.18	-0.044	0.01	0.209	0.30	-0.279	0.28	-0.274	0.17
PRIORICW	-1.079	61.37 ***	1.129	52.57 ***	-2.202	19.24 ***	-0.599	7.20 ***	1.598	36.29 ***	-1.417	4.47 **
Intercept	0.239	0.75	-1.102	8.36 ***	0.194	0.48	0.524	1.76	-1.493	7.37 ***	-2.479	9.55 ***
Annual fixed effects?	Yes		Yes		Yes		Yes		Yes		Yes	
Industry fixed effects?	Yes		Yes		Yes		Yes		Yes		Yes	
Likelihood ratio												
Chi-Square	293.5	<.0001	388.2	<.0001	175.0	<.0001	169.4	<.0001	246.0	<.0001	100.8	<.0001
\mathbb{R}^2		0.14		0.19		0.10		0.17		0.25		0.12
N		1,912		1,807		1,612		914		869		773

ranel b. The raulo 0.	I Lax-relai	on seer ner	lolal lee	S (IAAF EE	(11)							
		Total sai	nple of res	stated compan	y-years		Sample c	f restated cor	npany-ye	ars: only fu	rst restated	year included
)	1)		(2)		(3)	1	(4)		(5)		(9)
Dependent variable	REST	LAG	Z	IW404	MW4	04FIRST	RES	T_LAG	Μ	W404	MM	404FIRST
	Est.	Wald	Est.	Wald	Est.	<u>Wald</u>	Est.	Wald	Est.	Wald	Est.	<u>Wald</u>
TAXFEES_TF	1.051	3.51 *	-1.483	3.67 *	-2.070	2.95 *	1.530	2.97 *	-3.182	5.21 **	-4.623	3.69 *
AUDITFEES	-0.199	15.19 ***	0.422	39.83 ***	0.463	26.78 ***	-0.258	10.83 ***	0.325	13.76 **	* 0.208	2.66
AUDITRELATED	0.228	2.87 *	-0.140	0.66	0.040	0.03	0.454	4.29 **	-0.100	0.24	-0.115	0.11
OTHERFEES	-0.396	3.50 *	0.262	1.23	0.177	0.39	-0.940	1.04	0.707	0.34	0.617	0.16
BIG4	0.549	9.28 ***	-1.213	31.10 ***	-1.235	18.70 ***	1.090	20.33 ***	-1.460	21.22 **	* -1.767	17.32 ***
GC	-0.612	4.17 **	0.632	3.38 *	0.382	0.47	-0.464	1.55	0.530	1.28	0.539	0.56
ACHANGE	-0.443	3.20 *	0.475	2.62	0.591	2.36	-0.319	1.00	0.220	0.20	0.268	0.23
LOGTA	-0.054	2.14	-0.052	1.08	-0.070	0.93	-0.083	2.22	-0.006	0.01	0.030	0.08
LOGNBS	0.161	3.94 **	-0.079	0.49	-0.130	0.69	0.071	0.28	0.339	3.28 *	0.336	1.63
FOREIGN	0.087	0.43	-0.419	5.05 **	-0.570	5.13 **	0.101	0.22	-0.831	7.00 **	* -0.488	1.29
RESTRUCT	0.121	0.81	-0.097	0.26	-0.015	0.00	-0.138	0.43	-0.160	0.32	-0.264	0.42
ACQ	0.265	2.41	-0.179	0.53	-0.005	0.00	0.476	3.29 *	0.112	0.10	0.371	0.70
SSOT	-0.256	4.40 **	0.512	8.65 ***	0.542	5.63 **	-0.576	10.65 ***	0.958	12.26 **	* 1.003	7.55 ***
LEV	0.178	0.48	0.152	0.20	-0.031	0.00	0.185	0.23	-0.244	0.22	-0.255	0.15
PRIORICW	-1.081	61.83 ***	1.131	52.99 ***	-2.161	18.75 ***	-0.615	7.62 ***	1.616	38.25 **	* -1.356	4.07 **
Intercept	0.099	0.12	-0.905	5.53 **	-1.486	7.55 ***	0.361	0.80	-1.148	4.10 **	-2.100	7.19 ***
Annual fixed effects?	Yes		Yes		Yes		Yes		Yes		Yes	
Industry fixed effects?	Yes		Yes		Yes		Yes		Yes		Yes	
Likelihood ratio												
Chi-Square	295.1	<.0001	389.9	<.0001	175.6	<.0001	167.0	<.0001	246.3	<.0001	99.	8 < 0001
\mathbb{R}^2		0.14		0.19		0.10		0.17		0.25		0.12
Z		1.912		1,807		1.612		914		869		773

Panel B. The ratio of tax-related fees to total fees (TAYFFES TF)

Panel C: The ratio of	tax-rela	ted fees t	o auc	lit fees	(TAXI	FEES	(AF)			Comolo	f waatatad ac		and the second	+ 40040404	populori noo
		1 0tal (1)	sampi	e or res	tated coi	mpany	/-years	(٤		Sample (ot restated cc	mpany-y	ears: only fir (5)	st restated	year included
Dependent variable	RES	T LAG		M	V404		MW40	(J) (4FIRS	L	RES	T LAG	4	1W404	7MM	404FIRST
4	Est.	Wald		Est.	Wald		Est.	Wald		Est.	Wald	Est.	Wald	Est.	Wald
TAXFEES_AF	0.755	5.71 *	Ť).646	1.99	-	-1.211	2.99	*	1.059	4.44 **	-1.168	1.81	-2.238	2.11
AUDITFEES	-0.195	14.61 **)).423	39.92	* * *	0.463	26.69	***	-0.253	10.52 ***	0.328	14.13 ***	0.210	2.81 *
AUDITRELATED	0.160	1.29	Ŷ	0.095	0.29		0.130	0.29		0.372	2.73 *	-0.005	0.00	-0.031	0.01
OTHERFEES	-0.453	4.53 *	_ _).302	1.53		0.264	1.03		-0.994	1.16	0.848	0.49	0.794	0.27
BIG4	0.549	9.24 *	- *	1.216	31.59	* *	-1.235	18.76	* * *	1.086	20.14 ***	-1.445	21.10 ***	-1.750	17.30 ***
GC	-0.608	4.13 *	_ _).627	3.32	*	0.375	0.45		-0.464	1.57	0.518	1.23	0.497	0.48
ACHANGE	-0.444	3.22 *	U	0.473	2.61		0.588	2.34		-0.326	1.04	0.230	0.23	0.275	0.24
LOGTA	-0.054	2.14	Ŷ	0.051	1.05	-	-0.067	0.86		-0.083	2.20	-0.005	0.00	0.036	0.10
LOGNBS	0.160	3.87 *	Ť	0.079	0.49	-	-0.131	0.70		0.069	0.26	0.341	3.31 *	0.322	1.50
FOREIGN	0.088	0.45	Ŷ).412	4.89	*	-0.574	5.18	* *	0.101	0.22	-0.805	6.60 **	-0.472	1.21
RESTRUCT	0.118	0.78	Ŷ	0.101	0.29	-	-0.020	0.01		-0.141	0.45	-0.163	0.33	-0.291	0.52
ACQ	0.270	2.47	Ŷ	0.176	0.51	-	-0.012	0.00		0.482	3.38 *	0.103	0.08	0.352	0.63
LOSS	-0.257	4.44 *	_ *).518	8.82	* * *	0.549	5.77	*	-0.577	10.65 ***	0.981	12.91 ***	1.030	8.00 ***
LEV	0.186	0.53	U).138	0.17	-	-0.044	0.01		0.189	0.24	-0.293	0.31	-0.297	0.20
PRIORICW	-1.082	62.18 **	- **	1.137	53.65	* *	-2.159	18.83	* * *	-0.609	7.48 ***	1.620	38.76 ***	-1.349	4.11 **
Intercept	0.099	0.12	Ŷ	086.0	6.58	*	-1.559	8.52	**	0.374	0.88	-1.355	6.18 **	-2.312	8.96 ***
Annual fixed effects?	Yes			Yes			Yes			Yes		Yes		Yes	
Industry fixed effects?	Yes			Yes			Yes			Yes		Yes		Yes	
Likelihood ratio															
Chi-Square	297.1	<.0001		388.5 <	<.0001		175.5 <	<.0001		168.5	<.0001	243.	< .0001	97.8	<.0001
\mathbb{R}^2		0.14			0.19			0.10			0.17		0.24		0.12
N		1,912			1,807			1,612			914		869		773
<i>Notes:</i> See Table 1 for vari respectively. Statistical signiare not reported.	able defini ficances ar	tions. Statis e calculated	stical s I by clu	ignificar ıstering 1	nces base the stands	d on t ard erre	wo-tailed ors within	tests at compan	the 1 ies (Pe	percent, 5 stersen 20	percent, and 39). For brevit	10 percer y, the resu	it levels are d	enoted by * 1stry and yea	**, **, and * ar fixed effects

5 Conclusions

Although SOX (2002) prohibits the provision of most non-audit services by incumbent audit firms, tax services remain allowed because of the potential benefits from knowledge spillover. We have empirically examined whether tax-related fees paid to the incumbent auditor are associated with the likelihood of 1) restatement lag of misstated financial statement and 2) material weakness disclosure of misstatement companies. Collectively, our findings suggest that providing taxrelated services does not in itself have a significantly positive or negative impact on audit quality. However, the empirical findings of this study provide some evidence that higher levels of tax-related fees are associated with restatement lags, and stronger evidence on the association between higher tax-related fees and unacknowledged internal control problems. That is, the findings of the current study indicate that, among companies with misstatements in financial statements, the misstatements and control problems are more likely to remain undiscovered during the fiscal year in question when the tax-related fees paid to the audit firm are higher. These findings have implications for regulators and company stakeholders as well as for the audit profession.

References

Ashbaugh, H., R. LaFond & B. Mayhew (2003). Do nonaudit services compromise auditor independence? Further evidence. *The Accounting Review* 78:3, 611–639.

Blankley, A. I., D. N. Hurtt & J. E. MacGregor (2012). Abnormal audit fees and restatements. *Auditing: A Journal of Practice & Theory* 31:1, 79–96.

Choi, W.-W. & H.-Y. Lee (2009). The provision of tax services by incumbent auditors and earnings management: evidence from Korea. *Journal of International Financial Management and Accounting* 20:1, 79–103.

Chung, H. & S. Kallapur (2003). Client importance, nonaudit services, and abnormal accruals. *The Accounting Review* 78:4, 931–955.

DeAngelo, L. E. (1981). Auditor size and audit quality. *Journal of Accounting* and *Economics* 3:3, 183–199.

DeFond, M., K. Raghunandan & K. Subramanyam (2002). Do nonaudit service fees impair auditor independence? Evidence from going-concern audit opinions. *Journal of Accounting Research* 40:4, 1247–1274.

Eilifsen, A. & W. F. Messier, Jr. (2000). The incidence and detection of misstatements: A review and integration of archival research. *Journal of Accounting Literature* 19, 1–43.

Favere-Marchesi, M. (2006). The impact of tax services on auditors' fraud-risk assessment. *Advances in Accounting* 22, 149–165.

Ferguson, M., G. Seow & D. Young (2004). Nonaudit services and earnings management: UK evidence. *Contemporary Accounting Research* 21:4, 813–841.

Francis, J. & B. Ke (2006). Disclosure of fees paid to auditors and the market valuation of earnings surprises. *Review of Accounting Studies* 11:4, 495–523.

Frankel, R., M. Johnson & K. Nelson (2002). The relation between auditors' fees for nonaudit services and earnings management. *The Accounting Review* 77:4, 71–105.

Gleason, C. A. & L. F. Mills (2011). Do auditor-provided tax services improve the estimate of tax reserves? *Contemporary Accounting Research* 28:5, 1484–1509.

Harris, D. & J. Zhou (2013). Auditor-provided tax consulting, knowledge spillovers, and reported weaknesses in internal control. *Working paper*. Syracuse University, University of Hawaii at Manoa.

Huang, H-W., S. Mishra & K. Raghunandan (2007). Types of nonaudit fees and financial reporting quality. *Auditing: A Journal of Practice & Theory* 26:1, 133–145.

Joe, J. R. & S. D. Vandervelde (2007). Do auditor-provided nonaudit services improve audit effectiveness? *Contemporary Accounting Research* 24:2, 467–487.

Kanagaretnam, K., G. Krishnan & G. Lobo (2011). An empirical analysis of auditor independence in the banking industry. *The Accounting Review* 85:6, 2011–2046.

Khurana, I. & K.K. Raman (2006). Do investors care about the auditor's economic dependence on the client? *Contemporary Accounting Research* 23:4, 977– 1016.

Kinney Jr., W., Z.-V. Palmrose & S. Scholz (2004). Auditor independence, nonaudit services, and restatements: was the U.S. government right? *Journal of Accounting Research* 42:3, 561–588.

Krishnan, J., H. Sami & Y. Zhang (2005). Does the provision of nonaudit services affect investor perceptions of auditor independence? *Auditing: A Journal of Practice & Theory* 24:2, 111–135.

Krishnan, G. & G. Visvanathan (2011). Is there an association between earnings management and auditor-provided tax services? *Journal of the American Taxa-tion Association* 33:2, 111–135.

Leone, A. (2007). Factors related to internal control disclosure: a discussion of Ashbaugh, Collins, and Kinney (2007) and Doyle, Ge, and McVay (2007). *Journal of Accounting and Economics* 44, 224–237.

Nelson, M. W. (2009). A model and literature review of professional skepticism in auditing. *Auditing: A Journal of Practice & Theory* 28:2, 1–34.

Paterson, J. S. & A. Valencia (2011). The effects of recurring and nonrecurring tax, audit-related, and other nonaudit services on auditor independence. *Contemporary Accounting Research* 28:5, 1510–1536.

Petersen, M. A. (2009). Estimating standard errors in finance panel data sets: Comparing approaches. *Review of Financial Studies* 22:1, 435–480.

PCAOB (2005). Public Company Accounting Oversight Board. *Ethics and independence rules concerning independence, tax services, and contingent fees.* PCAOB Rulemaking Docket Matter No. 017, Release No. 2005-014. Available from: http://www.pcaobus.org/.

Raghunandan, K., W. Read & S. Whisenant (2003). Initial evidence on the association between nonaudit fees and restated financial statements. *Accounting Horizons* 17:3, 223–234.

Reynolds, J.K., D.R. Deis, Jr. & J.R. Francis (2004). Professional service fees and auditor objectivity. *Auditing: A Journal of Practice & Theory* 23:1, 29–52.

Rice, S. C. & D. P. Weber (2012). How effective is internal control reporting under SOX 404? Determinants of the (non-)disclosure of existing material weaknesses. *Journal of Accounting Research* 50:3, 811–843.

Robinson, D. (2008). Auditor independence and auditor-provided tax service: Evidence from going-concern audit opinions prior to bankruptcy filings. *Auditing: A Journal of Practice & Theory* 27:2, 31–54.

SEC (2003). Securities and Exchange Commission. *Final rule: Strengthening the commission's requirements regarding auditor independence*. Washington, DC. Available from: http://www.sec.gov/.

SOX (2002). *The Sarbanes-Oxley act of 2002*. United States House of Representatives, Public Law 107–204 [H.R. 3763]. Washington D.C.

Schneider, A., B. Church & K. Ely (2006). Non-audit services and auditor independence: a review of the literature. *Journal of Accounting Literature* 25, 169–211.

Seetharaman, A., Y. Sun & W. Wang (2011). Tax-related financial statement restatements and auditor-provided tax services. *Journal of Accounting, Auditing & Finance* 26:4, 677–698. Simunic, D. (1984). Auditing, consulting, and auditor independence. *Journal of Accounting Research* 22, 679–702.

European Accounting Review, 2014 http://dx.doi.org/10.1080/09638180.2014.906315

Audit Partner Public-Client Specialisation and Client Abnormal Accruals

KIM ITTONEN*, KARLA JOHNSTONE** and EMMA-RIIKKA MYLLYMÄKI*

*Department of Accounting and Finance, University of Vaasa, PO Box 700, 65101 Vaasa, Finland and **Wisconsin School of Business, University of Wisconsin– Madison, 975 University Avenue, Madison, WI 53706, USA

(Received: November 2011; accepted: February 2014)

ABSTRACT We examine the association of Big 4 audit partners' public-client specialisation with client companies' audit quality. Using a sample of NASDAQ OMX companies in Finland, we identify the audit partner assigned to each public-client engagement. We expect that partners with greater public-client specialisation provide higher quality auditing, since they have likely developed deep domain-specific knowledge and a keen sense of the litigation and reputational risks posed by public clients. In addition, the willingness to resist client pressure likely increases with the number of public clients in the partner's portfolio because dependence on any one client diminishes, which should help to ensure audit quality. The results show that public-client specialisation is negatively associated with abnormal accruals, and this result is attributable to audit partners with three to six public clients. The results of supplemental tests imply that public-client specialisation is more important when general auditing experience is lower. Further, the results reveal that in our setting of high-tax and high alignment between financial reporting and tax reporting, greater public-client specialisation is particularly associated with smaller income-decreasing abnormal accruals, suggesting that auditors with greater public-client specialisation likely recognise the downside reputational implications and achieve audit quality by discouraging tax avoidance.

1. Introduction

The purpose of this paper is to examine how the extent of audit partners' specialisation in public clients is associated with abnormal accruals, a proxy for client companies' audit quality.¹ Prior research has provided extensive evidence of the effects of *audit firm* or *local audit office-level* characteristics on both audit quality (Balsam, Krishnan, & Yang, 2003; Krishnan, 2005; Reichelt & Wang, 2010; Reynolds & Francis, 2001) and audit fees (Craswell, Francis, & Taylor, 1995; Ferguson, Francis, & Stokes, 2003). However, the empirical evidence on how *individual audit partner* specialisation affects client financial reporting outcomes is limited due to the widespread lack of audit partner signature on audit reports, which enables identification of individual partners with specific client engagements. Chin and Chi (2009) and Chi and Chin (2011) examine the

Correspondence Address: Karla Johnstone, EY Professor, University of Wisconsin School of Business, 975 University Avenue, Madison, WI 53706, USA. Email: kjohnstone@bus.wisc.edu

Paper accepted by Ann Vanstraelen.

¹Abnormal accruals reflect the quality, or lack thereof, of reported earnings. Reported earnings are the output of the financial reporting and the audit process, and thus represent a signal of actual audit quality since the audit process affects the client's audited financial statements (Becker et al., 1998; Francis, 2011).

© 2014 European Accounting Association

effects of audit partner industry expertise and audit quality, whereas Zerni (2012) focuses on audit partner specialisation and audit fees. The current study extends this prior literature by examining how the extent of audit partners' public-client specialisation is associated with abnormal accruals.

Recent literature suggests that it may be more informative to focus on office (Ferguson et al., 2003; Francis, Reichelt, & Wang, 2005; Francis, Stokes, & Anderson, 1999; Reichelt & Wang, 2010; Reynolds & Francis, 2001) and audit partner level characteristics (Chi & Chin, 2011; Chin & Chi, 2009; Zerni, 2012) than on overall audit firm-level analyses. More specifically, auditor expertise is tied to the deep knowledge of individual professionals and at least partly this knowledge cannot be captured and distributed by the firm to other offices (Ferguson et al., 2003) or even to other partners (Chi & Chin, 2011; Zerni, 2012). Because signing partners plan and implement the audit and ultimately determine the type of audit report to be issued to the client (Ferguson et al., 2003), it is logical to assume that the characteristics of these individuals have a substantial effect on the quality of the audit engagement (Chin & Chi, 2009). In addition, signing the public audit report increases the risk of personal reputation loss in case of an audit failure, and hence might work as an incentive for performing high-quality auditing (Chi & Chin, 2011).

Prior research suggests that domain-specific knowledge is an important determinant of expertise, and this kind of knowledge can be acquired through the repetition of similar tasks, i.e. specialisation (Bonner & Walker, 1994; Solomon, Shields, & Whittington, 1999; Thibodeau, 2003; Zerni, 2012). Prior research has mainly investigated auditors' specialisation in different industries, and audit partner industry specialisation is positively associated with audit quality (Chi & Chin, 2011; Chin & Chi, 2009). Specialisation in public companies is another means to gain domain-specific knowledge (Zerni, 2012). Auditing public clients requires specialised knowledge concerning financial reporting (international financial reporting standards (IFRS)) and auditing requirements. In addition to the enhanced expertise achieved through publicclient specialisation, the willingness to resist client pressure is likely to increase with the number of public clients in the partner's portfolio as the partner's dependence on any one client diminishes, which should help to ensure audit quality (Johnstone, Sutton, & Warfield, 2001). Zerni (2012) finds that both industry specialisation and specialisation in public companies are associated with higher audit fees. His findings suggest that auditors with greater public-client specialisation put more effort into performing the audit and/or that they are valued by financial statement users and corporate insiders - implying higher *perceived* audit quality. However, prior research has not investigated whether public-client specialisation is associated with actual audit outcomes that are indicative of higher quality. Building on Chin and Chi (2009), Zerni (2012) and Chi and Chin (2011), we examine the relationship between the extent of individual Big 4 audit partner's public-client specialisation and the absolute value of client abnormal working capital accruals (WCA), a proxy for *actual* audit quality. Specifically, we predict lower absolute abnormal accruals for clients having partners with greater public-client specialisation compared to those with lesser public-client specialisation.

To conduct our analyses, we use a sample of 420 company-year observations from the NASDAQ OMX Exchange in Finland. We track the identities of audit partners assigned to public-client engagements,² and link the partner's identity with client financial statement data. Thus, we are able to examine the association between the extent of audit partners' public-client specialisation and abnormal accruals. Our primary analyses show a significant negative association between greater public-client specialisation and absolute abnormal WCA.

²Audit partner signatures have been available in EU member states since the implementation of the Directive on Statutory Audit (Council of the European Union, 2006 (2006/43/EC)). However, Finland has a long tradition of publishing audit partner signatures in the annual reports.

We also explore the extent of public-client specialisation that may be optimal versus potentially dysfunctional. On one hand, audit quality is likely to improve with the number of publicclients in the partner's portfolio as both the partner's expertise and independence is likely to be increased. In contrast, and analogous to studies on busy directors (Beasley, 1996; Core, Holthausen, & Larcker, 1999; Fich & Shivdasani, 2006), it is possible that having too many public clients at some point may lead to overcommitted audit partners, which could threaten audit quality. To investigate these possibilities, we analyse the linearity of the association between public-client specialisation and abnormal accruals. We find that the negative association between public-client specialisation is not evident for partners with seven or more public clients. Thus, a moderate level of public-client specialisation appears optimal, and the busyness that goes with having a high level of public-client specialisation appears to mitigate the knowledge and independence benefits associated with this type of specialisation.

In supplemental analyses, we examine whether public-client specialisation reduces the potentially negative association between shorter audit experience and audit quality. We re-estimate our primary hypothesis testing model and include an interaction term of public-client specialisation and experience. The results show a significant and positive interaction of public-client specialisation and experience, implying that public-client specialisation compensates for potential lower audit quality of less experienced auditors. We also split the sample into higher versus lower experience groups, and find that public-specialisation is significantly and negatively associated with abnormal accruals in the below-median experience sub-sample, but insignificant in the above-median experience sub-sample. Therefore, public-client specialisation appears to be most important for less experienced partners.

Finally, we investigate the association between public-client specialisation and abnormal accruals separately for income-increasing and income-decreasing sub-samples. Incomeincreasing accruals are usually considered as the primary signal of earnings management, but some prior research (Collins & Hribar, 2000; Peasnell, Pope, & Young, 2005) show that incentives and opportunities for income-decreasing accruals exist, for example when managers desire to mitigate the magnitude of a positive earnings surprise. In addition, Shackelford and Shevlin (2001) illustrate that managers also may seek to decrease earnings in order to minimise taxation, especially in high-tax countries. Further, Van Tendeloo and Vanstraelen (2008) show that Big 4 auditors constrain earnings management in countries with a high alignment between financial reporting and tax accounting because the probability that an audit failure is detected is larger when tax authorities have higher interest. Tax-driven choices of accounting policies in unconsolidated financial statements might 'flow through' to consolidated financial statements because of administrative convenience or because tax authorities expect such conformity (Gee, Haller, & Nobes, 2010). With respect to the tax authorities' expectations, it is important to note that in Finland the taxes of listed companies and their subsidiaries are administered and audited by a special unit for taxation of consolidated corporations. To conclude, in our setting with hightax and high alignment between financial reporting and tax accounting, the tax authorities are likely to monitor aggressive tax minimisation strategies of consolidated corporations and thus the group auditors with greater public-client specialisation likely recognise the downside reputational implications of client misreporting and achieve audit quality by discouraging aggressive tax avoidance. The results reveal that companies audited by partners with greater public-client specialisation have significantly smaller income-decreasing abnormal accruals. Thus, in addition to providing the insight that audit partner public-client specialisation is associated with improved audit quality, this study further contributes to the literature by revealing the extent of public-client specialisation that appears to be optimal, and the nature of earnings management that is mitigated by partner public-client specialisation.

The remainder of this paper proceeds as follows. The next section develops our hypothesis. The following sections present research methods and describe results. The final section first presents limitations of the analyses and then discusses conclusions.

2. Background and Hypothesis Development

2.1. Literature on Audit Quality

Until recently, research on audit quality has mainly focused on examining audit firm characteristics and their impact on audit engagements (Balsam et al., 2003; Becker, DeFond, Jiambalvo, & Subramanyam, 1998; Francis & Krishnan, 1999). Considering portfolios at the audit-firm level, DeAngelo (1981) shows that the diversified client portfolios of large audit firms are less dependent on the outcome of any individual client relationship, which should be associated with higher audit quality. Studies on industry specialisation at the national level indicate that clients audited by industry-specialist audit firms have higher earnings quality than clients audited by non-specialists (Balsam et al., 2003; Krishnan, 2003).

More recent research demonstrates the importance of local office-level audit firm characteristics in determining both audit quality (usually proxied by measures of conservative financial reporting or abnormal accruals) and audit fees (Francis, Richard, & Vanstraelen, 2005; Francis & Yu, 2009; Krishnan, 2005; Reichelt & Wang, 2010; Reynolds & Francis, 2001). This office-level perspective suggests that expertise and knowledge within the audit office are difficult to capture and distribute by the firm to other offices (Ferguson et al., 2003; Johnstone, Li, & Luo, in press).

We extend this prior research by examining individual audit partner characteristics at the audit partner portfolio level, as suggested by DeFond and Francis (2005). Auditor expertise is tied to the deep knowledge of individual professionals and it is difficult to distribute this knowledge to other partners (Chi & Chin, 2011; Zerni, 2012). Because signing partners plan and implement the audit and ultimately determine the type of audit report to be issued to the client (Ferguson et al., 2003), it is logical to assume that these engagement partners have a substantial effect on audit quality (Chin & Chi, 2009). In addition, signing the public audit report leads to an increase in the risk of personal reputation loss in case of an audit failure (Chi & Chin, 2011) or attention from taxing authorities (Van Tendeloo & Vanstraelen, 2008). Following the European Union's (EU's) 8th directive (2006/43/EC) requiring the disclosure of engagement partner identity and a Public Company Accounting Oversight Board (PCAOB, 2009) release on requiring individual partner signatures, Zerni (2012) notes that these initiatives also demonstrate that the quality of public-client audits rests to a large extent on the input of individual auditors exercising their own professional judgement.

2.2. Public-Client Specialisation

Underlying audit task knowledge develops early, including both explicit and tacit knowledge, and is an important determinant of audit task performance (Libby & Luft, 1993). As the auditor specialises in certain domains, domain-specific expertise grows (Bonner & Walker, 1994; Solomon et al., 1999; Thibodeau, 2003; Zerni, 2012). More specialised individuals have generally acquired their expertise through relevant experience, education and training, and consequently they are expected to have the ability to outperform less specialised individuals in their domain (Bedard, 1989; Bedard & Chi, 1993). Over time, the auditor establishes personal and professional contacts in the business community, which enables networking across individuals through repeated interactions (Borgatti & Cross, 2003). These interactions yield enhancements in the acquisition of domain-specific knowledge.

As the auditor gains further experience, the likelihood increases that the auditor will work on engagements within a given domain. Ultimately, over time and with experience, the auditor develops an understanding of commonalities and shared risks/opportunities among clients in a given domain, which constitutes individual auditor-level domain-specific knowledge. This type of domain-specific auditing knowledge is associated with superior performance (Jamal & Tan, 2001; Libby & Tan, 1994; Ramsay, 1994). In terms of resolving difficult financial reporting issues, Johnstone, Bedard, and Biggs (2002) show that auditors with greater specialised knowledge about how to resolve a complex financial reporting issue generate a greater number and monetary range of alternatives for evaluation as they prepare to negotiate with an aggressive client. Brown and Johnstone (2009) find that auditors with higher levels of negotiation experience are better able to negotiate a relatively more conservative financial reporting outcome, even when the client initially prefers an aggressive financial reporting outcome. Specialised auditors possess characteristics that enable them to achieve higher quality auditing and financial reporting outcomes. Taken together, this line of research suggests that specialisation is a critical determinant of developing expertise in a particular subject matter.³

Prior research has mainly investigated auditors' specialisation in different industries (Balsam et al., 2003; Chi & Chin, 2011; Chin & Chi, 2009; Ferguson et al., 2003; Francis et al., 2005; Krishnan, 2003, 2005; Reichelt and Wang, 2010). Examining industry specialisation at the audit partner level, Chin and Chi (2009) find that differences in restatement likelihood due to industry expertise are mainly attributable to the partner-level experts rather than firm-level experts. In addition, Chi and Chin (2011) report that individual audit partner industry expertise, together with audit firm industry specialisation, increase audit quality.

Another means to differentiate audit service is through specialisation in public versus private companies. It is logical to assume that the auditor's specialised knowledge of auditing public clients increases with the number of public clients audited. Auditing large, complex, high-profile clients is likely to require a great deal of auditor expertise (Zerni, 2012). Public clients require specialised knowledge concerning financial reporting and the associated auditing requirements mandated by government and trading exchanges. It is reasonable that public-client specialists possess the required domain-specific knowledge, which results in a higher level of expertise and higher audit quality. Moreover, auditing listed clients requires an indepth understanding of the litigation risks that such clients represent to individual audit partners and their audit firms as a whole. Public-client specialists bear the litigation and regulatory attention (Zerni, 2012). Thus, partners with high levels of public-client specialisation are likely high-profile leaders of their respective offices and audit firms, and, in addition, have valuable reputational capital at stake, which should encourage them to ensure that their clients present the highest quality financial statements. Further, the willingness to resist client pressure is likely

³We recognise that experience, knowledge, and specialisation are distinct, but related concepts, both conceptually and empirically. Experience involves 'the learning of action-outcome connections ... learning occurs through outcome feedback' (Einhorn & Hogarth, 1981, p. 78). 'Of great importance to the issue of learning from experience is the role of awareness of the task factors that can influence outcomes' (Einhorn & Hogarth, 1981, p. 79.) Knowledge 'is generally rich in content and complex in form. It includes appearance, function, relation to other objects, and any other property of the object that can be deduced from our general knowledge of the world' (Tversky, 1977, p. 329). Libby and Luft (1993) help explain how repeated experiences affect knowledge development, noting that '... the content and the organization of knowledge can be changed by decision makers' learning opportunities' (p. 428). Expertise develops through repeated experiences and the associated development of knowledge is thus a component of auditor expertise in addition to the general knowledge base required for all audits' (p. 301). Industry specialisation is associated with product differentiation and fee premia (Craswell et al., 1995).

to increase with the number of public clients in the partner's portfolio as the partner's dependence on any one client diminishes, which should help to ensure audit quality (Johnstone et al., 2001). Therefore, public-client specialisation may positively affect audit quality because it assists in enhancing auditor independence.

Zerni (2012) finds that both industry specialisation and specialisation in public companies are associated with higher audit fees, suggestive of higher *perceived* audit quality. Chin and Chi (2009) and Chi and Chin (2011) document the association between industry specialisation and *actual* audit quality. Prior research has not, however, examined whether the public-client specialisation actually results in higher quality audit outcomes, i.e. higher quality financial reporting of client companies. We extend Zerni (2012) by investigating whether public-client specialisation is associated with a financial reporting measure that proxies for *actual* audit quality, abnormal WCA.

Based on this discussion, our expectation is that auditors with a greater extent of specialisation in auditing public clients will be associated with higher quality audits:

H1: Audit partner public-client specialisation is negatively associated with the absolute value of client abnormal WCA.

A factor that may potentially mitigate the impact of specialisation on audit quality is audit partner over-commitment. For example, while we hypothesise that public-client specialisation will enhance audit quality, having too many public clients may at some point lead to overcommitted audit partners who do not have time to focus on all clients. Of course, it is difficult to predict how many is 'too many' public clients. Prior research on busy board members shows that companies with busy directors have weaker performance and corporate governance (Beasley, 1996; Core et al., 1999; Fich & Shivdasani, 2006). Based on this discussion, we propose the following research question:

Research Question 1: Is there a point at which the audit quality benefits associated with public-client specialisation are mitigated due to over-commitment?

3. Method

3.1. Data

Our sample includes publicly listed companies of the NASDAQ OMX Helsinki exchange in Finland covering fiscal years 2006–2010.⁴ The NASDAQ OMX exchange provides an expedient environment for studying public-client specialisation because audit partners are required to personally sign the audit reports on behalf of the audit firm, and thus, can be identified for each client company. This is in contrast with, for example, the USA, where only the name of the audit firm is public information.

Table 1 describes the sample selection. First, to construct our research variable we search all NASDAQ OMX Helsinki listed companies during years 2005–2010, yielding 793 engagements. We remove observations for companies traded in First North Finland, companies for which auditor information is unavailable, and companies with a foreign auditor. This yields 779 engagements for which we are able to identify the signing auditor (the engagement partner) after locating and reviewing the audit opinions included in the annual reports. In constructing the final sample, we exclude banks, insurance companies, and other financial companies (Standard Industrial Classification (SIC) codes 6000–6900) due to their unique regulations. To avoid

⁴As of 1 January 2005 all listed companies are required to follow the IFRS.

Fiscal year ended	2005	2006	2007	2008	2009	2010	Total
Companies (listed on NASDAQ OMX Helsinki)	136	136	135	131	129	126	793
Less companies traded in First North Finland	0	1	2	2	2	2	9
=	136	135	133	129	127	124	784
Less companies with auditor information not available	2	0	0	0	1	0	3
=	134	135	133	129	126	124	781
Less companies with a foreign auditor	0	0	0	1	1	0	2
= Companies with identified audit partner	134	135	133	128	125	124	779
Less observations with fiscal year length six months	1	2	0	0	0	0	3
=	133	133	133	128	125	124	776
Less financial companies (SIC Codes 6XXX)	13	13	13	13	14	13	79
=	120	120	120	115	111	111	697
Less companies with non-Big4 audit firm	8	9	8	8	6	4	43
=	112	111	112	107	105	107	654
Less companies with two audit firms (joint audits)	2	2	2	0	0	0	6
=	110	109	110	107	105	107	648
Less observations with missing data	52	36	28	20	16	18	170
=	58	73	82	87	89	89	478
Less observations from fiscal year 2005	58	0	0	0	0	0	58
Final sample of companies used in the analyses	0	73	82	87	89	89	420

 Table 1.
 Sample selection

potential audit quality differences based on audit firm size, we exclude clients of non-Big 4 audit firms. In addition, we exclude the few rare observations with joint audits, where engagement partners from two audit firms have signed the audit report, again to avoid potential audit quality differences across audit firms. After removing the observations with missing data on our dependent and independent variables, the sample includes 478 observations. Since our public-client specialisation variables are determined based on the number of public clients in the *previous* fiscal year,⁵ the first year included in the sample is the year 2006. Our final sample includes 420 company-year observations. The financial statement data are obtained from the Thomson Financial Worldscope database.

3.2. Model Specifications

Our empirical tests focus on the association between public-client specialisation and abnormal WCA, which is a proxy for actual audit quality. The abnormal part of WCA is defined as the difference between actual WCA and expected WCA. WCA are calculated as the change in current assets (less the change in cash and cash equivalents) minus the change in current liabilities (less the change in short-term debt and current portion of long-term debt). In determining the expected part of WCA, we use the DeFond and Park (2001) model:

Expected WCA = Sales
$$x \left(\frac{WCA_{t-1}}{Sales_{t-1}} \right)$$
. (1)

Following prior audit studies using non-US data (Carey & Simnett, 2006; Francis et al., 2009; Francis & Wang, 2008; Maijoor & Vanstraelen, 2006; Zerni, Haapamäki, Järvinen, & Niemi,

⁵We employ the previous year's number of public clients in order to capture the public-client specialisation at the beginning of a company-year.

Variable	Definition
ABWCA	Absolute value of the abnormal WCA calculated as actual WCA minus expected WCA based on the model by DeFond and Park (2001).
PUBS	The number of public clients for which the engagement partner signs the audit report in the previous year. In cases for which two partners sign the audit report, the number of public clients is measured for the partner with greater specialisation.
$PUBS \ge 3$	One if the number of public clients in previous year is greater than or equal to three, zero otherwise.
3 < PUBS < 6	One if the number of public clients in previous year is greater than or equal to three but less than six, zero otherwise.
$PUBS \ge 6$	One if the number of public clients in previous year is greater than or equal to six, zero otherwise.
LOGEXP	The natural logarithm of the number of years since the partner's Certified Public Accountant (CPA) examination. In cases for which two partners sign the audit report, experience is measured for the partner with greater experience.
INDSPEC	One if the audit partner audited at least two clients belonging to a particular industry in year <i>t</i> , zero otherwise.
LOGASSETS	The natural logarithm of total assets.
GROWTH	The percentage change in sales from previous year.
PB	Price-to-book ratio.
LEVERAGE	Total debt divided by total assets.
INV	Total inventory divided by total assets.
LOSS	One if net income is negative, and zero otherwise.
OCF	Operating cash flow divided by total assets.
AC	One if the company has an audit committee, and zero otherwise.
LAGWCA	Lagged working capital accruals scaled by lagged total assets.
STDSALES	The standard deviation of sales over the years $t-3$ to t
AUDITFIRM	Audit firm fixed effects (Deloitte, EY, KPMG, and PWC).
YEAK	Year fixed effects (2006–2010).
INDUSTRY	I wo-digit SIC industry classifications.

2012), we employ the DeFond and Park (2001) model because the Jones (1991) model and its modifications may underperform when the industry samples are small (Francis et al., 2009; Peek, Meuwissen, Moers, & Vanstraelen, 2013; Zerni et al., 2012).

To examine whether audit partner public-client specialisation is associated with audit quality, we estimate the following fixed effects panel regression:

$$|ABWCA| = a + public-client specialisation + \beta_1 LOGEXP + \beta_2 INDSPEC + \beta_3 LOGASSETS + \beta_4 GROWTH + \beta_5 PB + \beta_6 LEVERAGE + \beta_7 INV + \beta_8 LOSS + \beta_9 OCF + \beta_{10}AC + \beta_{11} LAGWCA + \beta_{12} STDSALES + fixed effects + \varepsilon,$$
(2)

where ABWCA is the abnormal working capital accruals (ABWCA) calculated as actual WCA minus expected WCA, scaled by lagged total assets. The absolute values of ABWCA are used to capture both income-increasing and income-decreasing ABWCA. Larger values of |ABWCA| suggest lower actual audit quality. Observations with extreme values are dropped, that is, if the absolute value of ABWCA scaled by lagged total assets is above one. See variable definitions in Table 2.

Turning to the independent variables of interest, we use the previous year's number of public clients in order to measure public-client specialisation at the beginning of a company-year.⁶ Public-client specialisation in Model (2) is measured in multiple ways. PUBS is a continuous variable measured by the number of the audit partner's public clients in the previous year. In addition to the continuous variable, we construct dummy variables for public-client specialisation. The appendix provides descriptive information about the number of public clients audited by partners in our sample. We observe that the average number of public clients per audit partner is 1.92, and, furthermore, that only 20% of partners audit three or more public clients. Consequently, we regard as specialists the minority of the engagement partners with three or more public clients. PUBS ≥ 3 is a dichotomous variable that equals one if the auditor had at least three public clients in the previous year, and zero otherwise. Moreover, we consider different levels of public-client specialisation and employ the following dichotomous variables:

- auditors with three to five clients ($3 \le PUBS \le 6$) and
- auditors with six or more clients (PUBS ≥ 6) (the top 5% of all auditors auditing listed companies audit six or more public clients (Appendix)).

Auditors with two or fewer public clients (PUBS ≤ 2), are used as the base. We assume that the highest level of specialisation is represented in the 5% of auditors who have six or more clients, followed by the group of about 15% of the audit partners with three to five clients.

Finally, in the fourth specification, we create dummy variables for the following levels of specialisation: PUBS \geq 7, PUBS = 6, PUBS = 5, PUBS = 4, PUBS = 3, and PUBS = 2. These are compared to the base group PUBS \leq 1. These variables enable us to test whether there exists a point at which the audit quality benefits associated with public-client specialisation are mitigated due to over-commitment.

Based on prior audit quality literature, we include numerous control variables in Model (2). Kaplan, O'Donnell, and Arel (2008), for example, find that high audit experience reduces the tendency to rely on information from management. To control for the audit partner's general audit experience, we use the natural logarithm of the number of years since the partner's CPA examination (LOGEXP) and expect this variable to be negatively associated with |ABWCA|.⁷ Prior research suggests that industry specialisation is associated with higher quality auditing and financial reporting outcomes (Balsam et al., 2003; Chi & Chin, 2011; Ittonen, Vähämaa, & Vähämaa, 2013; Krishnan, 2005; Zerni, 2012). Adapting Zerni (2012), an audit partner is an industry specialist if the partner audited at least two public clients

⁶The audit report must be signed by at least one auditor. In 16% of the 420 observations, the audit report is signed by two engagement partners from the same audit firm. We use the value for the partner with greater specialisation when measuring public-client specialisation, and we use the value for the partner with greater experience when measuring partner experience. As a sensitivity test, however, we use the average of the partners' public-client specialisation and inferences from these tests are essentially the same as those from the main analyses, although the significances are weaker. Based on interest from one reviewer, we also conducted analyses using the lowest number. We find that the significant results for our test variable become mainly insignificant using this specification. This makes sense because it seems most logical that the partner with greater public-client specialisation should be most important from a leadership perspective for the engagement; it also seems implausible that the effect of the partner with less public-client specialisation/experience would outweigh the effect of the partner with greater public-client specialisation.

⁷We use the database provided by the Central Chamber of Commerce to identify the number of years of professional experience for each auditor. We acknowledge that it would be desirable to control for number of years auditing public clients, however, this information is not available.

belonging to industry k in fiscal year t.⁸ We set a dichotomous variable equal to one if the partner is an industry specialist, and zero otherwise (INDSPEC). We anticipate a negative association between INDSPEC and |ABWCA|.

We measure client size using the natural logarithm of total assets, LOGASSETS, and expect that this variable will be negatively associated with ABWCA because large companies tend to have more stable operations (Zerni et al., 2012). We include growth in sales from the prior year (GROWTH) and price-to-book (PB) ratio in the model to capture the effect of growth on accruals behaviour, and expect these variables to be positively associated with |ABWCA| because growth companies have more incentives to try to meet their earnings benchmarks (Zerni et al., 2012). LEVERAGE equals total debt to total assets. We do not make a prediction on this control since there is evidence of a positive association with |ABWCA| (Francis & Wang, 2008), as well as a negative association (Zerni et al., 2012). INV equals inventory to total assets and is used to control for company complexity. Following previous studies (Ashbaugh-Skaife, Collins, Kinney, & LaFond, 2008; Ittonen et al., 2013), we expect that INV is positively associated with ABWCA, because in companies with larger proportions of inventory managerial discretion and estimates have a greater impact on reported earnings. We control for company performance by including LOSS and OCF in the model. LOSS is a dichotomous variable measuring whether the company has negative net income, and OCF is operating cash flow to total assets. We expect that LOSS (OCF) will be positively (negatively) associated with [ABWCA], because poor financial performance may induce to manipulate earnings (Francis & Wang, 2008). AC is a dichotomous variable measuring the existence of an audit committee.⁹ We expect that this variable will be negatively associated with |ABWCA| because the existence of an audit committee is likely to improve corporate governance and reduce aggressive earnings management (Klein, 2002). We also include one year lagged working capital accruals (LAGWCA) to control for the reversal of accruals, and following (Zerni et al., 2012) we expect a negative association with |ABWCA|). In addition, we include the standard deviation of sales over the years t-3 to t0 to control for the effect of operating variability on abnormal accruals (STDSALES), anticipating a positive association between this variable and ABWCA because larger variations in operations may provide greater opportunities for earnings management (Hribar & Nichols, 2007; Zerni et al., 2012).

We include audit firm fixed effects in Model (2) to control for audit firm-level effects on audit quality. We also include year fixed effects to control for temporal variation in audit quality and industry fixed effects to control for industry differences.

3.3. Descriptive Statistics

Table 3 Panel A presents the descriptive statistics. The mean (median) absolute ABWCA, |ABWCA|, equals 0.103 (0.066). In terms of individual audit partner characteristics, we find that the company-year observations in our sample have on average (median) a partner with 3.607 (3.000) publicly traded clients, while the mean (median) partner experience (EXPERI-ENCE) is 19 (20) years, and about 13% of the companies are audited by industry specialist auditors. The mean (median) total assets is 1785 (202) million euros. Mean (median) GROWTH is

⁸The industry classifications are based on two-digit SIC codes. Zerni (2012) defines industry specialists as those with five clients in the same industry. Our sample is characterised by smaller industry groups and we observe that only about 13% of the engagement partners are classified as industry specialists. Our sample includes only 15 company-year observations audited by engagement partners that have more than two public clients in the same industry.

⁹'The Finnish Corporate Governance Code' recommends that the public companies establish an audit committee. Noncompliance with the recommendation must be identified and explained in the annual report.

5.3 (5.2) per cent, and mean (median) PB ratio is 2.262 (1.776). Mean (median) LEVERAGE is 0.246 (0.249), and mean (median) INV is 0.144 (0.133). About 25% of companies have a current-year LOSS. The mean (median) operating cash flow to total assets (OCF) is 0.075 (0.081). About 49% of companies have an audit committee (AC). Finally, mean (median) WCA in the previous fiscal year (LAGWCA) is -0.005 (0.001), and mean (median) standard deviation of sales over the years t-3 to t0 (STDSALES) is 0.186 (0.132).

Table 3 Panel B displays Pearson correlation coefficients among selected variables used in the multivariate analyses. The number of public clients (PUBS), and experience (LOGEXP) are negatively correlated with |ABWCA|. The highest correlation among the independent variables is between LOGASSETS and AC (0.523). The variance inflation factors (VIF) do not indicate problems with multicollinearity among these variables.¹⁰

In Table 3 Panel C, we report descriptive statistics based on the number of public clients in the previous year. Table 3 Panel C shows that mean |ABWCA| is generally decreasing as the number of public clients audited increases. Audit partner experience is increasing as the number of public clients audited increases. Finally, industry specialisation is highest for audit partners auditing five public clients.

4. Results

4.1. Univariate Results

Table 4 reports the univariate results when the sample is partitioned into three groups based on the extent of public-client specialisation. The results in Panel A show that companies with audit partners that have three to five public clients have significantly lower |ABWCA| (p = 0.040), LAGWCA (p = 0.016), and STDSALES (p = 0.003), whereas LOGASSETS (p = 0.016) and the probability of having an audit committee (p = 0.015) are higher as compared to companies with audit partners that have two or fewer public clients. In terms of audit partner characteristics, partners that have three to five public clients have significantly more overall experience (LOGEXP) (p < 0.001) and industry specialisation (INDSPEC) (p = 0.001) than auditors that have two or fewer public clients. Panel B shows that companies with audit partners that have six or more public clients have significantly lower |ABWCA| (p = 0.016) and marginally lower LEVERAGE (p = 0.086), but higher LOGASSETS (p = 0.001) and a higher probability of having an audit committee ($p \le 0.001$) compared to companies audited by auditors with two or fewer public clients. Panel B also reports that audit partners with six or more public clients have more overall experience (LOGEXP) ($p \le 0.001$) and are more likely to be industry specialists (INDSPEC) (p = 0.011) than auditors with two or fewer public clients. Finally, Panel C indicates that companies with audit partners that have six or more public clients are more likely to have audit committees (AC) ($p \le 0.001$) and higher LAGWCA (p = 0.004) than partners with three to five clients.

4.2. Multivariate Results

Table 5 Panel A reports results of estimating Model (2) for the total sample. Statistical significances are calculated by clustering the standard errors within companies (Petersen, 2009). The results using the number of public clients (PUBS) are reported in column (1), the dichotomous variable PUBS \geq 3 in column (2), dichotomous variables 3 \leq PUBS < 6 and PUBS \geq 6 in

¹⁰For example, in one of our main estimations, column (1) of Table 5 Panel A, the highest VIF among these variables is 3.833.

Table 3.	Overall dea	scriptive s	tatistics,	correlati	ons, and	descriptive	statistics b	y public-	client sp	scialisatio	N = N	420)		
Variable	Mean	Std.	Min	25% tile	Median	75% tile	Мах	Skew.	Kurt.					
Panel A: Distributional stati ABWCA	istics 0.103	0.120	0.000	0.029	0.066	0.130	066.0	2.859	14.541					
PUBS	3.607	2.627	0.000	2.000	3.000	5.000	12.000	1.153	4.459					
$PUBS \ge 3$	0.564	0.496	0.000	0.000	1.000	1.000	1.000	-0.259	1.067					
Audit partner experience in	19.298	5.226	6.000	15.500	20.000	23.000	29.000	-1.047	3.605					
years INDSPEC	0.133	0.340	0.000	0.000	0.000	0.000	1.000	2.157	5.654					
Total assets (m EUR)	1784.897	4697.450	9.422	63.230	202.035	1403.199 3	7.619.000	0.402	2.254					
GROWTH	0.053	0.255	-1.000	-0.060	0.052	0.167	1.296	0.867	8.363					
PB	2.262	1.681	0.170	1.108	1.776	2.904	15.050	2.149	11.510					
LEVERAGE	0.246	0.155	0.000	0.123	0.249	0.355	0.705	0.211	2.458					
INV	0.144	0.127	0.000	0.029	0.133	0.216	0.714	1.205	5.298					
LOSS	0.248	0.432	0.000	0.000	0.000	0.000	1.000	1.169	2.368					
OCF	0.075	0.107	-0.514	0.031	0.081	0.137	0.337	-1.477	8.578					
AC	0.488	0.500	0.000	0.000	0.000	1.000	1.000	0.048	1.00					
LAGWCA	-0.005	0.100	-0.430	-0.049	0.001	0.042	0.660	0.180	9.456					
STDSALES	0.186	0.196	0.004	0.078	0.132	0.223	1.715	3.797	23.428					
Variable	2	3	4	5	9	7	8	6	10	11	12	13	14	15
Panel B: Pearson's correlati	ion coefficie	nts												
ABWCA (1)	-0.115	-0.127	-0.127	0.012	-0.192	0.040	-0.071	-0.170	0.017	0.200 -	-0.245	0.001	-0.031	0.285
PUBS (2)		0.734	0.373	0.120	0.109	0.003	-0.013	-0.055	0.034	-0.030	0.040	0.273	0.016	0.018
$PUBS \ge 3$ (3)			0.359	0.161	0.162	0.033	0.024	-0.088	0.014	-0.052	0.055	0.243	-0.076	-0.102
LOGEXP (4)				-0.069	0.273	-0.012	-0.082	-0.041	-0.024	-0.100	0.071	0.169	0.062	-0.103
INDSPEC (5)					-0.027	-0.008	-0.030	0.001	-0.075	0.083 -	-0.072	0.037	0.004	-0.010
LOGASSETS (6)						-0.006	-0.082	0.105	0.100	-0.269	0.187	0.523	0.099	-0.259
GROWTH (7)							0.106	-0.112	-0.011	-0.261	0.139	0.054 -	-0.014	0.090
PB (8)								-0.312	-0.064	-0.152	0.175 -	-0.052 -	-0.081	-0.070
LEV (9)									0.082	0.154 -	- 0.176 -	-0.041	0.020	-0.040
INV (10)										-0.103 -	-0.117	0.030	0.126	0.026
LOSS (11)										'	- 0.457 -	- 0.008 -	-0.092	0.130
OCF (12)											I	-0.024	0.133	-0.199
AC (13)													-0.030	-0.046
LAGWCA (14)														0.026
STDSALES (15)														

Panel C: Descriptive statistics Numbe	s by numb r of audit r	er of audi	it partne blic clien	r public its in the	clients previous	vear		
Mean	0	1	7	ю	4	, S	9	≥7
ABWCA	0.149	0.113	0.118	0.090	0.109	0.073	0.074	0.097
Audit partner experience in	12.480	18.684	17.733	20.780	20.306	20.357	19.137	23.400
years	0.160	0000	0.000	0110	3000	90L U	0.137	
INDOFEC	0.100	0.000	0.00	0.117	0.440	0.200	101.0	0.200
Ν	25	57	101	59	49	28	51	50
Notes: Variables as defined in Tabl	e 2. Bold tex	tt in Panel	B indicate	es signific:	ance at 0.0	5 level.		

Variable	$PUBS \le 2$	$3 \leq PUBS < 6$	Difference <i>t</i> -stat. or X^2 and <i>p</i> -value	
Panel A: Compari	sons between low and mo	derate levels of public-clie	ent specialisation	
ABWCA	0.121	0.093	2.07	0.040
LOGEXP	2.788	2.997	-6.18	<.001
INDSPEC	0.071	0.191	10.49	0.001
LOGASSETS	19.191	19.705	-2.42	0.016
GROWTH	0.043	0.060	-0.62	0.534
PB	2.217	2.335	-0.59	0.556
LEVERAGE	0.261	0.236	1 35	0.179
INV	0.142	0.140	0.16	0.874
2201	0.142	0.140	0.38	0.538
OCE	0.069	0.077	-0.64	0.536
	0.009	0.077	5.04	0.525
	0.004	0.483	2.42	0.015
	0.004	-0.023	2.45	0.010
SIDSALES	0.209	0.155	5.02	0.005
IN	183	136		
			Difference	e <i>t</i> -stat. or
	PUBS ≤ 2	$PUBS \ge 6$	X^2 and X^2	<i>p</i> -value
Panel B: Comparis	sons between low and hig	h levels of public-client sp	ecialisation	
ABWCA	0.121	0.085	2.43	0.016
LOGEXP	2.788	3.039	-7.54	<.001
INDSPEC	0.071	0.168	6.52	0.011
LOGASSETS	19.191	19.948	-3.24	0.001
GROWTH	0.043	0.060	-0.48	0.630
PB	2.217	2.245	-0.15	0.883
LEVERAGE	0.261	0.230	1 72	0.086
INV	0.142	0.153	-0.72	0.000
2201	0.273	0.208	1.48	0.171
OCE	0.069	0.085	-1.40	0.224
	0.350	0.005	40.10	< 001
	0.004	0.006	40.19	<.001 0.828
	0.004	0.000	-0.20	0.858
SIDSALES	0.209	0.180	0.75	0.450
IN	183	101		
		/	Difference	e <i>t</i> -stat. or
	$3 \le PUBS < 6$	$PUBS \ge 6$	X^2 and	<i>p</i> -value
Panel C: Compari	sons between moderate an	nd high levels of public-cli	ent specialisation	
ABWCA	0.093	0.085	-0.62	0.539
LOGEXP	2.997	3.039	1.56	0.119
INDSPEC	0.191	0.168	0.20	0.652
LOGASSETS	19.705	19.948	1.01	0.313
GROWTH	0.060	0.060	-0.02	0.982
PB	2.335	2.245	-0.43	0.665
LEVERAGE	0.236	0.230	-0.31	0.758
INV	0.140	0.153	0.94	0.347
LOSS	0.243	0.208	0.40	0.529
OCF	0.077	0.085	0.70	0.485
AC	0.485	0.743	15.92	< .001
LAGWCA	-0.025	0.006	2.88	0.004
STDSALES	0.155	0.186	1.12	0.264
N	136	101		0.201
	100	101		

Table 4.Univariate results

Notes: The significances are based on the *t*-test for continuous variables and the chi-square test for dichotomous variables. All variables are defined in Table 2.

Panel A: Total sample (N	N = 420				
Dependent variable	Exp. sign	(1) ABWCA Coef. (<i>t</i> -value)	(2) ABWCA Coef. (<i>t</i> -value)	(3) ABWCA Coef. (<i>t</i> -value)	(4) ABWCA Coef. (<i>t</i> -value)
PUBS	_	-0.005			
$PUBS \ge 3$	_	(-1.47)	-0.040		
PUBS: PUBS ≥ 6	_		(-1.85)	-0.024	
$3 \le PUBS < 6$	_			(-0.97) -0.050 $(-2.21)^{**}$	
$PUBS \ge 7$	_				-0.031
PUBS = 6	_				(-0.94) -0.037 (-1.22)*
PUBS = 5	_				(-1.52) -0.077 (-2.66)***
PUBS = 4	_				$(-2.00)^{**}$ -0.041
PUBS = 3	_				$(-1.39)^{\circ}$ -0.076
PUBS = 2	_				$(-2.61)^{-0.022}$
LOGEXP	—	-0.030	-0.026	-0.025	(-1.11) -0.020
INDSPEC	—	$(-1.34)^{2}$ -0.010	(-1.19) -0.004	(-1.16) -0.005	(-0.96) -0.006
LOGASSETS	—	(-0.41) -0.015	(-0.15) -0.014	(-0.18) -0.014	(-0.25) -0.014
GROWTH	+	$(-2.55)^{-11}$ 0.032	0.033	$(-2.53)^{-11}$ 0.035	$(-2.65)^{+++}$ 0.037
PB	+	(0.73) -0.008	(0.77) -0.008	(0.82) -0.008	(0.82) -0.008
LEVERAGE	?	(-1.95) -0.112	(-1.96) -0.118	(-1.94) -0.113	(-2.04) -0.110
INV	+	$(-1.90)^{2}$ -0.236	$(-2.02)^{++}$ -0.244	$(-1.97)^{2}$ -0.260	$(-1.92)^{*}$ -0.267
LOSS	+	0.015	(-2.04) 0.013	(-2.11) 0.012	(-2.19) 0.013
OCF	_	(0.70) -0.129	(0.63) -0.137	(0.60) -0.139	(0.64) -0.139
AC	_	$(-1.94)^{**}$ 0.021	$(-2.08)^{**}$ 0.020	$(-2.10)^{**}$ 0.018	$(-2.10)^{**}$ 0.019
LAGWCA	_	(0.93) 0.021	(0.92) 0.007	(0.83) -0.001	(0.88) 0.008
STDSALES	+	(0.15) 0.129 $(2.73)^{***}$	(0.05) 0.122 $(2.66)^{***}$	(-0.01) 0.123 $(2.70)^{***}$	(0.05) 0.125 $(2.73)^{***}$
Audit firm fixed effects? Year fixed effects? Industry fixed effects? Intercept		Yes Yes 0.510 (3.34)***	Yes Yes 0.501 (3.46)***	Yes Yes 0.494 (3.44)***	Yes Yes 0.505 (3.53)***
Adjusted R^2		0.28	0.29	0.30	0.30

 Table 5.
 Primary hypothesis testing models

(Continued)

Table 5. Continued Panel B: Sample of income-increasing ABWCA(N = 223) (1)(2)(3) (4) |ABWCA| |ABWCA| |ABWCA| |ABWCA| Coef. Coef. Coef. Coef. Exp. sign Dependent variable (t-value) (t-value) (t-value) (t-value) PUBS 0.000 _ (0.08) $PUBS \ge 3$ -0.025(-1.29)PUBS: $PUBS \geq 6$ -0.007(-0.32) $3 \le PUBS < 6$ -0.037(-1.84)** $PUBS \ge 7$ -0.009(-0.25)PUBS = 6-0.011(-0.38)PUBS = 5-0.036 $(-1.34)^{*}$ PUBS = 4-0.024(-0.80)PUBS = 3-0.055 $(-1.78)^{*}$ PUBS = 2-0.006(-0.25)LOGEXP -0.0050.004 0.006 0.006 (-0.25)(0.23)(0.31)(0.30)INDSPEC -0.001**0.011** 0.009 **0.009** _ (-0.04)(0.46)(0.40)(0.35)LOGASSETS -0.007-0.006-0.006-0.006(-1.05)(-1.19) $(-1.38)^*$ (-1.16)GROWTH +0.095 0.090 0.092 0.096 (1.86)** (1.82)** (1.86)** (1.81)** PB +-0.009 -0.009 -0.009 -0.009 (-1.98)(-1.98)(-1.95)(-2.02)? LEVERAGE -0.025-0.038-0.031-0.034(-0.41)(-0.63)(-0.51)(-0.55)INV -0.083-0.086-0.105-0.106+(-0.78)(-0.93)(0.72)(-0.92)LOSS -0.041-0.041-0.045 -0.042+(-1.84)(-1.74)(-1.67)(-1.68)OCF -0.059-0.076-0.084-0.091(-0.55)(-0.43)(-0.62)(-0.66)AC -0.007-0.007-0.009-0.008_ (-0.39)(-0.38)(-0.49)(-0.42)-0.793LAGWCA -0.802-0.814-0.799(-5.58)*** $(-5.65)^{***}$ (-5.77)*** $(-5.72)^{***}$ **STDSALES** 0.150 0.148 0.146 +0.150 $(1.63)^*$ $(1.60)^*$ $(1.61)^*$ $(1.48)^*$ Audit firm fixed effects? Yes Yes Yes Yes Year fixed effects? Yes Yes Yes Yes Industry fixed effects? Yes Yes Yes Yes 0.212 0.199 0.191 0.203 Intercept (1.91)* $(1.78)^*$ (1.96)* (1.82)* Adjusted R^2 0.49 0.49 0.50 0.49

(Continued)

		Tuble et	Continued		
Panel C: Sample of inco	me-decreas	ing ABWCA (<i>N</i> = 197)		
Dependent variable	Exp. sign	(1) ABWCA Coef. (<i>t</i> -value)	(2) ABWCA Coef. (<i>t</i> -value)	(3) ABWCA Coef. (<i>t</i> -value)	(4) ABWCA Coef. (<i>t</i> -value)
		-0.006	(******)	(******)	(*******)
1015		$(-1.45)^*$			
$PUBS \ge 3$	_		(-0.042) $(-1.57)^*$		
PUBS: PUBS ≥ 6	_			-0.037	
$3 \le PUBS < 6$	_			(-1.17) -0.046 $(-1.77)^{**}$	
$PUBS \geq 7$	-			(1.77)	-0.042
PUBS = 6	_				(-1.08) -0.067 $(-1.78)^{**}$
PUBS = 5	_				(-1.78) -0.064 $(-1.52)^*$
PUBS = 4	_				(-1.55) -0.065 (-1.04)**
PUBS = 3	_				(-1.94) -0.072 (-2.12)**
PUBS = 2	_				(-2.12) -0.035 $(-1.42)^*$
LOGEXP	_	-0.006	-0.002	-0.002	(-0.004
INDSPEC	_	(-0.19) 0.009	(-0.07) 0.016	(-0.05) 0.016	(-0.10) 0.014 (0.55)
LOGASSETS	_	(0.57) -0.011 $(-1.27)^*$	(0.09) -0.011 (-1.27)*	(0.09) -0.011 (-1.27)*	(0.53) -0.010
GROWTH	+	(-1.57) -0.029 (-0.30)	(-1.57) -0.027 (-0.38)	(-1.57) -0.026 (-0.36)	(-1.19) -0.026 (-0.25)
PB	+	0.000	0.000	0.000	0.001
LEVERAGE	?	(0.02) -0.129 $(-2.46)^{**}$	(0.05) -0.130 $(-2.64)^{***}$	(0.05) -0.129 (-2.63)***	(0.19) -0.131 (-2.65)***
INV	+	(-0.314)	(-0.319)	(-2.03) -0.323 (-2.76)	(-2.03) -0.333 (-2.01)
LOSS	+	(-2.08) 0.043 (1.14)	(-2.72) 0.040 (1.07)	(-2.76) 0.040 (1.07)	(-2.91) 0.041 (1.05)
OCF	_	(1.14) -0.069 (-0.61)	(1.07) -0.068 (-0.59)	(1.07) -0.070 (-0.61)	-0.075
AC	_	0.022	0.021	0.020	0.017
LAGWCA	—	(0.93) 0.734 (4.47)	0.722	(0.92) 0.716 (4.44)	(0.725) (4.24)
STDSALES	+	0.091	0.082	0.083	0.089
Audit firm fixed effects? Year fixed effects? Industry fixed effects? Intercept Adjusted R^2		Yes Yes 0.321 (2.08)** 0.57	Yes Yes Yes 0.319 (2.06)** 0.58	Yes Yes 0.316 (2.05)** 0.58	Yes Yes 0.325 (2.15)** 0.58

Table 5. Continued

Notes: All variables are defined in Table 2. The following symbols indicate significant effects: * = <0.10; ** = <0.05; *** = <0.01, with probability levels one-tailed for directional expectations. The reported *t*-values are based on clustered standard errors within companies (Petersen, 2009). For simplicity, results for the fixed effects are not reported.

column (3), and dichotomous variables for the different levels of public clients in column (4). All probability levels are one-tailed for directional expectations.

The results in column (1) indicate that the number of public clients (PUBS) is marginally significantly and negatively associated with |ABWCA| (p = 0.073), implying that companies audited by partners with greater public-client specialisation have somewhat higher audit quality. The results in column (2) further indicate that public-client specialisation as proxied by PUBS ≥ 3 is significantly negatively associated with |ABWCA| (p = 0.035). This finding suggests that auditors who have three or more public-clients perform higher quality audits than auditors with less than three public clients. Column (3) shows that $3 \leq PUBS \leq 6$ is significantly associated with |ABWCA| (p = 0.015), whereas the coefficient on PUBS ≥ 6 is insignificant. This finding implies that the association is robust for audit partners with three to five public clients. Overall, these results suggest that there is a positive relationship between public-client specialisation and audit quality, supporting H1.

Column (4) provides more detailed information about the different levels of specialisation. The results indicate significantly less abnormal accruals for companies audited by partners with three to six public clients. Regarding Research Question 1, these results imply that there is, indeed, a point at which a partner becomes so 'busy' auditing public clients that the audit quality benefits of public-client specialisation are mitigated. That point seems to be when a partner has more than six public clients.

In terms of the economic significance of our results, the results in column (2) imply that companies with partners that audit greater than or equal to three public clients have (on average) about 39% lower ABWCA than companies with auditors having less than three public clients. The results for our control variables in column (1) indicate that STDSALES (p = 0.004) is positively associated with |ABWCA|. LOGEXP is marginally negatively associated with |ABWCA| (p = 0.093), suggesting that companies audited by partners with more general audit experience have higher audit quality. LOGASSETS (p = 0.006), LEVERAGE (p = 0.060), and OCF (p =0.028) are negatively associated with |ABWCA|. In general, our control variables perform similarly across the different estimations.

In Table 5 Panels B and C, we split the sample into two sub-samples based on whether the abnormal accruals are greater than zero (income-increasing abnormal accruals, ABWCA > 0) or less than zero (income-decreasing abnormal accruals, ABWCA < 0), respectively. In Panel B columns (1) and (2), the association between public-client specialisation and income-increasing abnormal accruals is insignificant. In column (3) the negative and significant (p = 0.035) coefficient on the dichotomous variable $3 \le PUBS < 6$ indicates that the lower income-increasing ABWCA is related to audit partners that had three to five public clients. In column (4), PUBS = 3 is negative and significant (p = 0.039), and PUBS = 5 is marginally negative and significant (p = 0.091).

Panel C columns (1) and (2) indicate that the extent of public-client specialisation is marginally negatively (p = 0.076 and p = 0.060) associated with income-decreasing abnormal accruals. The results reported in column (3) indicate that companies audited by partners with three to five public clients have significantly smaller income-decreasing abnormal accruals (p = 0.040). Moreover, column (4) shows that the association is attributable to companies audited by partners with two to six public-clients. Taking Table 5 Panels B and C together, these results suggest that the negative association between public-client specialisation and |ABWCA| is primarily driven by income-decreasing abnormal accruals.

4.3. Robustness Checks

As a first robustness check, we employ the propensity score matching technique to control for differences in client characteristics between those that have an audit partner with more versus

Dependent variable: PUBS ≥ 3	Coef.	Wald χ^2
Total sample ($N = 420$)		
LOGASSETS	0.015	0.03
LEVERAGE	-0.965	1.73
LOSS	0.057	0.04
OCF	0.742	0.42
STATEOWN	0.680	2.39
AC	0.933	12.05***
Year fixed effects?	Yes	
Industry fixed effects?	Yes	
Intercept	-0.159	0.01
Likelihood ratio		
χ^2		57.02
$\Pr{Pr} > \chi^2$		<.0001

 Table 6.
 Logit regression for modelling the decision to hire an audit partner with greater public-client specialisation

Note: Industry dummies are based on one digit SIC codes.

*** = 0.01, with Probability level one-tailed for directional expectations.

less public-client specialisation (Lawrence, Minutti-Meza, & Zhang, 2011; Lennox, Francis, & Wang, 2012; Rosenbaum & Rubin, 1983). We match, without replacement, each company with an audit partner with more public-client specialisation (based on PUBS \geq 3), with a partner with less public-client specialisation (two or fewer public clients) that has the closest predicted value based on the estimated logit model.¹¹ The estimates from that model are reported in Table 6. All matches are within a maximum distance of 1%. The results show that the decision to hire an audit partner with greater public-client specialisation is strongly associated with the existence of an audit committee.

Next, we the use the matched sample of 306 observations and re-estimate model (2). The results in Table 7 show that PUBS ≥ 3 in column (2), and $3 \leq PUBS \leq 5$ in column (3) are negative and significant (p = 0.081 and p = 0.036). Moreover, PUBS = 3, PUBS = 5, and PUBS = 6 are significantly and negatively associated with |ABWCA| in column (4) (p = 0.005 and p = 0.011, and p = 0.098, respectively). The regression results for the income-increasing and income-decreasing sub-samples (untabulated) indicate that the negative association between public-client specialisation and client abnormal accruals is primarily driven by income-decreasing abnormal accruals. Consequently, the estimations on the matched samples add credibility to our main findings in Table 5. While matching cannot entirely solve the problem of endogeneity, it does to some extent alleviate concerns about the issue.

As a second robustness check, we repeat the analyses using alternative proxies for audit partners' public-client specialisation. First, we measure public-client specialisation based on the average number of public clients in years t-1 and t-2 to provide a longer-term average measure of public-client specialisation. The sample used in these regressions covers the fiscal years 2007-2010, since we need the information about the number of public clients from the previous two years. This gives us a sample size of 347 observations. The results

¹¹Adapted from Zerni et al. (2012) the following logit model is used to estimate the probability of employing a publicclient specialist audit partner: Prob(PUBS \geq 3) = $a + \beta_1 \text{LOGASSETS} + \beta_2 \text{LEVERAGE} + \beta_3 \text{LOSS} + \beta_4 \text{OCF} + \beta_5 \text{STATEOWN} + \beta_6 \text{AC} + \text{annual fixed effects} + \text{industry fixed effects} + \varepsilon$. STATEOWN equals one if the Finnish government has ownership in the company and zero otherwise. Industry fixed effects are based on one-digit SIC codes. Other variables are described in Table 2.

		1 1 5		I 、	-
Dependent variable	Exp. sign	(1) ABWCA Coef. (<i>t</i> -value)	(2) ABWCA Coef. (<i>t</i> -value)	(3) ABWCA Coef. (t-value)	(4) ABWCA Coef. (<i>t</i> -value)
PUBS	_	-0.004			
PUBS ≥ 3	_	(-0.89)	-0.034		
PUBS: PUBS ≥ 6	_		(-1.41)*	-0.016	
$3 \le PUBS < 6$	_			(-0.50) -0.044	
$PUBS \geq 7$	_			(-1.82)**	-0.008
PUBS = 6	_				(-0.18) -0.043
PUBS = 5	_				$(-1.30)^{*}$ -0.074
PUBS = 4	_				$(-2.34)^{**}$ -0.031
PUBS = 3	_				(-0.92) -0.085
PUBS = 2	_				$(-2.64)^{-0.030}$
LOGEXP	_	-0.041	-0.033	-0.032	(-1.27) -0.025
INDSPEC	_	$(-1.47)^{*}$ 0.005 (0.16)	(-1.24) 0.014 (0.45)	(-1.22) 0.014 (0.46)	(-0.93) 0.008
LOGASSETS	-	(0.10) -0.020 (-2.67)***	(0.43) -0.021 (-2.77)***	(0.40) -0.022 (-2.84)***	(0.20) -0.022 (-2.05)***
GROWTH	+	(-2.07) 0.095 (1.82)**	(-2.77) 0.091 (1.82)**	(-2.84) 0.091 (1.81)**	(-2.93) 0.096 (1.82)**
PB	+	$(1.83)^{-0.004}$	$(1.82)^{-0.004}$	$(1.81)^{-0.004}$	$(1.82)^{-0.005}$
LEVERAGE	?	(-0.71) -0.177 (-2.72)***	(-0.08) -0.176 (-2.82)***	(-0.04) -0.167 (-2.71)***	(-0.76) -0.166 (-2.78)***
INV	+	$(-2.72)^{-0.299}$	$(-2.82)^{-0.304}$	$(-2.71)^{-0.319}$	$(-2.78)^{-0.320}$
LOSS	+	(-2.08) 0.034 $(1.40)^*$	(-2.13) 0.034 $(1.42)^*$	(-2.18) 0.033 $(1.41)^*$	(-2.19) 0.037 $(1.57)^*$
OCF	_	(1.40) -0.181 (-2.12)**	(1.42) -0.176	(1.41) -0.173 (-2.01)**	(1.57) -0.170 $(-1.07)^{**}$
AC	_	(-2.12) 0.040 (1.57)	(-2.03) 0.040 (1.62)	(-2.01) 0.039 (1.64)	(-1.97) 0.040 (1.71)
LAGWCA	_	0.036	(1.02) 0.024 (0.14)	0.017	0.032
STDSALES	+	0.167	0.163	0.164	0.175
Audit firm fixed effects? Year fixed effects? Industry fixed effects? Intercept Adjusted B^2		(2.67) Yes Yes 0.563 (2.97)***	(2.74) Yes Yes 0.564 (3.05)***	(2.60) Yes Yes 0.576 (3.13)***	(3.06) Yes Yes 0.591 (3.25)***
Aujusieu A		0.31	0.52	0.32	0.55

Table 7. Robustness tests: propensity score matched sample (N = 306)

Notes: All variables are defined in Table 2. The following symbols indicate significant effects: * = <0.10; ** = <0.05; *** = <0.01, with probability levels one-tailed for directional expectations.

The reported *t*-values are based on clustered standard errors within companies (Petersen, 2009). For simplicity, results for the fixed effects are not reported.

(untabulated) support the inferences derived from the main results in Table 5, although the significances are weaker. We find that companies audited by partners with greater public-client specialisation have higher audit quality. In particular, companies audited by partners with on average three to five public clients have lower |ABWCA| (p = 0.076) than companies audited by partners with less than three public clients, whereas companies audited by partners with on average at least six clients do not have significantly lower abnormal accruals than the control group.

Next, we measure public-client specialisation based on the aggregated number of public clients in years t-1 and t-2 (results untabulated). First, the coefficient of the continuous variable PUBS is negative and marginally significant (p = 0.061). Second, a dummy variable PUBS $\ge 5^{12}$ is negative and significant at the 0.05 level (p = 0.026). Third, $5 \le PUBS < 10$ is negative and significant at the 0.05 level (p = 0.013), whereas PUBS $\ge 10^{13}$ is negative and marginally significant at the 0.10 level (p = 0.082).

In addition, we conducted a sensitivity test (results not tabled) in which we measured publicclient specialisation as the square root of the sum of partners' clients' total assets in the previous year scaled by largest partner's total assets. The results from this analysis are, however, statistically insignificant. This result is perhaps not surprising given that prior research generally shows that task-specific auditing experience is associated with superior performance. Thus, it is not the size of the assets audited, but rather the number of client engagements that the partner has been involved in that appears to be important.

Finally, since we are using panel data, we estimate the main regressions with company fixed effects instead of industry fixed effects to control for any unobserved heterogeneity across sample companies. The inferences derived from this analysis are mainly similar to the results in Table 5, but with lower significance levels. In contrast to Table 5, we find a negative and marginally significant coefficient on PUBS ≥ 6 in the full sample and a negative and marginally significant coefficient on PUBS ≥ 7 in the income-decreasing sub-sample. While the results of the regressions with company fixed effects support our main findings, we note that company dummies might cause problems in examining the research question if the public-client specialisation variables do not vary enough within companies across years and therefore interpretations must be made with caution.

4.4. Supplemental Tests

Based on previous findings suggesting that auditor experience may affect audit quality (Kaplan et al., 2008), it could be expected that the importance of public-client specialisation is different for auditors with lower general audit experience compared to auditors with higher experience. Therefore, we examine whether public-client specialisation affects the expected negative association between shorter audit experience and earnings quality. To address this, we estimate Model (2) after including an interaction term of public-client specialisation and experience (PUBS*LOGEXP). The results are reported in column (1) of Table 8. The results indicate that while public-client specialisation and experience alone are significantly and negatively associated with abnormal accruals (p = 0.022 and p = 0.020, respectively), the coefficient of their interaction term is significant and positive (p = 0.059 based on a two-tailed test). This implies that public-client specialisation compensates for lower audit experience.

¹²Consistent with our main variable definition, we observe that only the top 20% of partners have five public clients or more during t-1 and t-2. Consequently, PUBS \geq 5 represents the public-client specialisation in this analysis.

¹³The variable PUBS \geq 10 represents the top 5% of partners based on the number of public clients during t-1 and t-2, and in this analysis they represent the partners with the highest level of public-client specialisation.

Gen	eral au	diting experience	, public-client	specialisation, a	nd ABWCA	
		The interaction between PUBS and LOGEXP	The sample experi	partitioned base ience (Mean =	ed on audit par 19.298, Media	ther's general $n = 20$)
		(N = 420)	Audit partne less than 20 y	r experience is years $(N = 187)$	Audit partner least 20 yea	experience is at ars $(N = 233)$
		(1)	(2) ABWCA	(3) ABWCA	(4) ABWCA	(5) ABWCA
Dependent variable	Exp. sign	Coef. (<i>t</i> -value)	Coef. (<i>t</i> -value)	Coef. (<i>t</i> -value)	Coef. (<i>t</i> -value)	Coef. (<i>t</i> -value)
PUBS	_	-0.057 $(-2.05)^{**}$	-0.018 $(-2.18)^{**}$		-0.001 (-0.19)	
$PUBS \ge 3$	_			-0.079 $(-2.29)^{**}$		0.000 (0.00)
LOGEXP	-	-0.064 $(-2.08)^{**}$	-0.047 (-0.83)	-0.043 (-0.79)	0.064 (0.60)	0.060 (0.55)
PUBS*LOGEXP	?	0.017 (1.91)*				
INDSPEC	-	-0.013 (-0.52)	0.030 (0.71)	0.035 (0.85)	-0.040 $(-2.50)^{***}$	-0.041 $(-2.51)^{***}$
LOGASSETS	-	-0.014 $(-2.55)^{***}$	-0.022 $(-2.71)^{***}$	-0.024 $(-2.86)^{***}$	-0.004 (-0.50)	-0.004 (-0.47)
GROWTH	+	0.031 (0.70)	0.034 (0.53)	0.039 (0.64)	-0.016 (-0.23)	-0.015 (-0.22)
PB	+	-0.009 (-1.98)	-0.016 (-3.21)	-0.015 (-3.14)	0.007 (1.13)	0.007 (1.13)
LEVERAGE	?	-0.113 $(-1.98)^*$	-0.254 $(-2.81)^{***}$	-0.257 $(-3.01)^{***}$	-0.033 (-0.47)	-0.032 (-0.42)
INV	+	-0.240 (-2.03)	-0.217 (-1.37)	-0.245 (-1.51)	-0.213 (-1.73)	-0.213 (-1.71)
LOSS	+	0.014 (0.69)	0.057 (1.53)*	0.052 (1.49)*	-0.023 (-0.84)	-0.023 (-0.83)
OCF	-	-0.126 $(-1.88)^{**}$	0.002 (0.02)	-0.012 (-0.13)	-0.294 $(-2.95)^{***}$	-0.293 $(-2.86)^{***}$
AC	-	0.023 (1.02)	0.038 (1.13)	0.034 (1.02)	-0.018 (-0.90)	-0.019 (-0.95)
LAGWCA	_	0.023 (0.17)	-0.290 $(-2.66)^{***}$	-0.318 $(-2.65)^{***}$	0.337 (1.75)	0.338 (1.79)
STDSALES	+	0.126 (2.68)***	0.061 (0.81)	0.047 (0.63)	0.163 (4.62)***	0.162 (4.76)***
Audit firm fixed effects?		Yes	Yes	Yes	Yes	Yes
Year fixed effects? Industry fixed effects?		Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes
Intercept		0.616 (3.44)***	0.804 (3.23)***	0.826 (3.24)***	-0.036 (-0.09)	-0.033 (-0.08)
Adjusted R^2		0.29	0.39	0.40	0.36	0.35

Table 8.Supplemental tests

Notes: All variables are defined in Table 2. The following symbols indicate significant effects: * = <0.10; ** = <0.05; *** = <0.01, with probability levels one-tailed for directional expectations. The reported *t*-values are based on clustered standard errors within companies (Petersen, 2009). For simplicity, results for the fixed effects are not reported.
Audit Partner Public-Client Specialisation 23

Moreover, we split the sample based on the median of auditor experience (20 years), and separately estimate Model (2) for the sub-sample of observations where auditor experience is less than 20 years, and for the sub-sample of observations where auditor experience is greater than or equal to 20 years. The results in columns (2)-(5) of Table 8 indicate that public-client specialisation is significantly and negatively associated with abnormal accruals in the below-average experience sub-sample, but insignificant in the above average experience sub-sample. In sum, it appears that public-client specialisation has its most important impact on audit quality when general auditing experience is less than 20 years, but does not have a significant impact on audit quality when audit experience is higher.¹⁴

4.5. Limitations

Prior to discussing our conclusions, we acknowledge certain limitations of the study. First, due to data limitations, the public-client specialisation variable does not take into account partners' entire career-long past experience in auditing public clients, public client experience gained when working on public client audits without signing the audit report, or the time available for public clients given the number of other (private) clients they have in their portfolio. We attempt to address the first issue by additionally using the averages and cumulative number of public clients from the previous two years. In addition, we acknowledge that as a robustness check it would be interesting to construct a measure of specialisation that takes into account the individual partner's entire portfolio of clients, both public and private. For example, two audit partners with the same number of public clients may still differ in the quality of the audits they provide because they may differ in terms of their private client portfolio. Measuring audit partner portfolios to include both public and private clients would be an extremely onerous task, since e.g. the Orbis database reveals that there are about 180,000 active private companies in Finland and we would have to manually track the name of the audit partner because it is not available in the database. Therefore, we opt to use a measure based on public clients only. It should be noted that the variable measuring audit partner industry specialisation is affected by data limitations in that we are able to determine industry specialisation based only on the data of listed companies.

Second, our sample is limited to one geographic region of the world. Replication in a broader European setting would also be appropriate. For example, it would be interesting to replicate our study in a European country that has low alignment between financial reporting and tax accounting to determine how our results might change regarding income-decreasing abnormal accruals as the driver of the association between public-client specialisation and audit quality. Third, abnormal accruals are a noisy measure of audit quality. While prior related research also uses this measure (Francis et al., 2009; Zerni et al., 2012), future research using more direct audit quality metrics would be a useful extension. Fourth, it is possible that the regression model used in this study suffers from self-selection caused by some client characteristics simultaneously affecting both the test variables and the dependent variables. We considered controlling for self-selection, but were unable to find an exogenous variable that strongly affects the company's choice of public-specialist auditors but has no direct effect on abnormal accruals. Following the conclusions of Larcker and Rusticus (2010) and Lennox et al. (2012), who emphasise that implementing selection models without valid instrumental variables can cause extremely fragile inferences, we are unable to control for self-selection. The matching procedures reported in the robustness checks should alleviate these concerns to some extent.

¹⁴Results using cutoffs of 19, 18, 17, and 16 years yield the same inference.

24 K. Ittonen et al.

5. Conclusions

In this paper we empirically examine the association between abnormal accruals and Big 4 audit engagement partner specialisation in public clients. Based on the literature showing that firm, office and partner-level specialisation improves audit quality, we hypothesise that audit engagement partners with greater specialisation in public clients have superior knowledge concerning financial reporting and auditing requirements, and a better understanding of the litigation risks that such clients present as compared to partners with less public-client specialisation. Consequently, these partners should be able to address the complex financial reporting, regulatory, and litigation-related issues related to public clients, and thus, outperform non-specialist audit partners in terms of audit quality. Furthermore, partners with greater public-client specialisation are likely more willing to resist client pressure because dependence on any one client diminishes with a larger public-client portfolio.

Our analyses consider the association between audit partner public-client specialisation and audit quality, proxied by the absolute value of client ABWCA. Extending research on audit quality at the individual audit partner level, our results reveal a statistically significant negative association between abnormal accruals and individual audit partner public-client specialisation. Specifically, companies with partners having greater public-client specialisation have lower abnormal accruals, and thus higher actual audit quality, compared to companies with partners having less public-client specialisation. It appears that partners with greater public-client specialisation seem better able to talk clients away from relatively more aggressive financial reporting compared to partners with less public-client specialisation. Extending Zerni's (2012) findings that clients appear to recognise and value the knowledge that specialised partners bring to the audit task, thus, implying *perceived* audit quality, our results suggest that publicclient specialisation that may be optimum, the results show that the association between specialisation and audit quality is most pronounced for partners with three to six public clients; thus, there may be a point at which 'busyness' becomes an issue.

Moreover, our results suggest that companies audited by partners with greater public-client specialisation have significantly smaller income-*decreasing* abnormal accruals. In the Finnish setting that is characterised by high-tax and high alignment between taxation and financial reporting, the tax authorities are likely to monitor corporations' tax minimisation strategies and thus auditors with greater public-client specialisation appear to recognise the downside reputational implications and achieve audit quality by discouraging tax avoidance. Finally, the results of supplemental tests suggest that public-client specialisation compensates for potential lower audit quality of less experienced auditors; public-client specialisation appears to be most important for less experienced audit partners.

Acknowledgements

Johnstone acknowledges support from the Andersen Center for Financial Reporting at the University of Wisconsin School of Business, and the EY Foundation for support of her EY Professorship. Ittonen acknowledges financial support from the NASDAQ OMX Nordic Foundation, the Marcus Wallenberg Foundation, Southern Ostrobothnia Fund, and the Foundation for Promoting Equity Markets in Finland. Myllymäki gratefully acknowledges the financial support received from the Finnish Foundation for Economic Education, Jenny and Antti Wihuri Foundation, Nordea Bank Foundation, Oskar Öflund's Foundation, and Evald and Hilda Nissi Foundation. We appreciate the thoughtful comments of two anonymous reviewers, the editor Ann Vanstraelen, Jean C. Bedard, Marsha Keune, Nicole Ratzinger-Sakel, Stefan Sundgren, Sami

Vähämaa, Mikko Zerni, and the seminar participants at the 3rd Workshop on Audit Quality (Bellagio), NHH (Bergen), and the Aarhus University Business School.

References

- Ashbaugh-Skaife, H., Collins, D. W., Kinney, W. R., & LaFond, R. (2008). The effect of SOX internal control deficiencies and their remediation on accrual quality. *The Accounting Review*, 83(1), 217–250.
- Balsam, S., Krishnan, J., & Yang, J. (2003). Auditor industry specialization and earnings quality. Auditing: A Journal of Practice & Theory, 22(2), 71–97.
- Beasley, M. (1996). An empirical analysis of the relation between the board of director composition and financial statement fraud. *The Accounting Review*, 71(4), 443–465.
- Becker, C., DeFond, M., Jiambalvo, J., & Subramanyam, K. R. (1998). The effect of audit quality on earnings management. Contemporary Accounting Research, 15(1), 1–24.
- Bédard, J. (1989). Expertise in auditing: Myth or reality? Accounting, Organizations and Society, 14, 113-131.
- Bédard, J., & Chi, M. T. H. (1993). Expertise in auditing. Auditing: A Journal of Practice & Theory, 12(Supplement), 21-45.
- Bonner, S., & Walker, P. (1994). The effects of instruction and experience on the acquisition of auditing knowledge. *The Accounting Review*, 69, 157–178.
- Borgatti, S. P., & Cross, R. (2003). A relational view of information seeking and learning in social networks. *Management Science*, 49(4), 432–445.
- Brown, H., & Johnstone, K. M. (2009). Resolving disputed financial reporting issues: Effects of auditor negotiation experience and engagement risk on negotiation process and outcomes. *Auditing: A Journal of Practice & Theory*, 28(2), 65–92.
- Carey, P., & Simnett, R. (2006). Audit partner tenure and audit quality. The Accounting Review, 81(3), 653-676.
- Chi, H., & Chin, C. (2011). Firm versus partner measures of auditor industry expertise and effects on auditor quality. Auditing: A Journal of Practice & Theory, 30(2), 201–229.
- Chin, C., & Chi, H. (2009). Reducing restatements with increased industry expertise. Contemporary Accounting Research, 26(3), 729–765.
- Collins, D., & Hribar, P. (2000). Earnings-based and accrual based market anomalies: One effect or two. *Journal of* Accounting and Economics, 29(1), 101–123.
- Core, J., Holthausen, R., & Larcker, D. (1999). Corporate governance, chief executive officer compensation, and firm performance. *Journal of Financial Economics*, 51, 371–406.
- Council of the European Union. (2006). Directive 2006/43/EC on statutory audits of annual accounts and consolidated accounts, amending Council Directives 78/60/EEC and repealing Council Directive 84/253/EEC. Official Journal of the European Union L157/87, 9 June 2006.
- Craswell, A. T., Francis, J. R., & Taylor, S. (1995). Auditor brand name reputations and industry specializations. *Journal of Accounting and Economics*, 20(3), 297–322.
- DeAngelo, L. E. (1981). Auditor size and audit quality. Journal of Accounting and Economics, 3(3), 183-199.
- DeFond, M. L., & Francis, J. R. (2005). Audit research after Sarbanes-Oxley. Auditing: A Journal of Practice & Theory, 24(Supplement), 5–30.
- DeFond, M. L., & Park, C. W. (2001). The reversal of abnormal accruals and the market valuation of earnings surprises. *The Accounting Review*, 76(3), 5–30.
- Einhorn, H. J., & Hogarth, R. M. (1981). Behavioral decision theory: Processes of judgment and choice. Annual Review of Psychology, 32, 53–88.
- Ferguson, A., Francis, J. R., & Stokes, D. J. (2003). The effects of firm-wide and office-level industry expertise on audit pricing. *The Accounting Review*, 78(2), 429–448.
- Fich, E., & Shivdasani, A. (2006). Are busy boards effective monitors? Journal of Finance, 61(2), 689-724.
- Francis, J. R. (2011). A framework of understanding and researching audit quality. Auditing: A Journal of Practice & Theory, 30(2), 125–152.
- Francis, J. R., & Krishnan, J. (1999). Accounting accruals and auditor reporting conservatism. Contemporary Accounting Research, 16(1), 135–165.
- Francis, J. R., Reichelt, K., & Wang, D. (2005). The pricing of national and city-specific reputations for industry expertise in the U.S. audit market. *The Accounting Review*, 80(1), 113–136.
- Francis, J. R., Richard, C., & Vanstraelen, A. (2009). Assessing France's joint audit requirement: Are two heads better than one? Auditing: A Journal of Practice & Theory, 28(2), 35–63.
- Francis, J. R., Stokes, D. J., & Anderson, D. (1999). City markets as a unit of analysis in audit research and the re-examination of Big 6 market shares. *Abacus*, 35(2), 185–206.

26 K. Ittonen et al.

- Francis, J. R., & Wang, D. (2008). The joint effect of investor protection and Big 4 audits on earnings quality around the world. *Contemporary Accounting Research*, 25(1), 157–191.
- Francis, J. R., & Yu, M. D. (2009). Big 4 office size and audit quality. The Accounting Review, 85(4), 1521-1552.
- Gee, M., Haller, A., & Nobes, C. (2010). The influence of tax on IFRS consolidated statements: The convergence of Germany and the UK. *Accounting in Europe*, 7(1), 97–122.
- Hribar, P., & Nichols, C. (2007). The use of unsigned earnings quality measures in tests of earnings management. *Journal of Accounting Research*, 45(5), 1017–1053.
- Ittonen, K., Vähämaa, E., & Vähämaa, S. (2013). Female auditors and accruals quality. Accounting Horizons, 27(2), 205–228.
- Jamal, K., & Tan, H. T. (2001). Can auditors predict the choices made by other auditors? Journal of Accounting Research, 39(3), 583–597.
- Johnstone, K. M., Bedard, J. C., & Biggs, S. F. (2002). Aggressive client reporting: Factors affecting auditors' generation of financial reporting alternatives. Auditing: A Journal of Practice & Theory, 21(1), 47–65.
- Johnstone, K. M., Li, C., & Luo, S. (in press). Client-auditor supply chain relationships, audit quality, and audit pricing. Auditing: A Journal of Practice & Theory. doi: http://dx.doi.org/10.2308/ajpt-50783
- Johnstone, K. M., Sutton, M., & Warfield, T. (2001). Antecedents and consequences of independence risk: Framework for analysis. Accounting Horizons, 15(1), 1–18.
- Jones, J. (1991). Earnings management during import relief investigations. Journal of Accounting Research, 29(2), 193– 228.
- Kaplan, S., O'Donnell, E., & Arel, B. (2008). The influence of auditor experience on the persuasiveness of information provided by management. *Auditing: A Journal of Practice and Theory*, 27(1), 67–83.
- Klein, A. (2002). Audit committee, board of director characteristics, and earnings management. *Journal of Accounting and Economics*, 33(3), 375–400.
- Krishnan, G. V. (2003). Does Big 6 auditor industry expertise constrain earnings management? Accounting Horizons, 17(Supplement), 1–16.
- Krishnan, G. V. (2005). The association between Big 6 auditor industry expertise and the asymmetric timeliness of earnings. *Journal of Accounting Auditing & Finance*, 20(3), 209–228.
- Larcker, D. F., & Rusticus, T. O. (2010). On the use of instrumental variables in accounting research. Journal of Accounting and Economics, 49(3), 186–205.
- Lawrence, A., Minutti-Meza, M., & Zhang, P. (2011). Can big 4 versus non-big 4 differences in audit-quality proxies be attributed to client characteristics? *The Accounting Review*, 86(1), 259–286.
- Lennox, C. S., Francis, J. R., & Wang, Z. (2012). Selection models in accounting research. *The Accounting Review*, 87(2), 589-616.
- Libby, R., & Luft, J. (1993). Determinants of judgment performance in accounting settings: Ability, knowledge, motivation, and environment. *Accounting, Organizations and Society, 18*(5), 425–450.
- Libby, R., & Tan, H. T. (1994). Modeling determinants of audit expertise. Accounting, Organizations and Society, 19(8), 701–716.
- Maijoor, S., & Vanstraelen, A. (2006). Earnings management within Europe: The effects of member state audit environment, audit firm quality and international capital markets. Accounting and Business Research, 36(1), 33–52.
- Nelson, M. W., Libby, R., & Bonner, S. E. (1995). Knowledge structure and the estimation of conditional probabilities in audit planning. *The Accounting Review*, 70(1), 27–47.
- Peasnell, K., Pope, P., & Young, S. (2005). Board monitoring and earnings management: Do outside directors influence abnormal accruals. *Journal of Business Finance and Accounting*, 32(7), 1311–1346.
- Peek, E., Meuwissen, R., Moers, F., & Vanstraelen, A. (2013). An evaluation of abnormal accruals measurement models in an international context. *European Accounting Review*, 22(3), 533–572.
- Petersen, M. A. (2009). Estimating standard errors in finance panel data sets: Comparing approaches. *Review of Financial Studies*, 22(1), 435–480.
- Public Company Accounting Oversight Board. (2009). Concept release on requiring the engagement partner to sign the audit report. PCAOB Release No. 2009–005.
- Ramsay, R. J. (1994). Senior/manager differences in audit workpaper review performance. Journal of Accounting Research, 32(1), 127–145.
- Reichelt, K., & Wang, D. (2010). National and office-specific measures of auditor industry expertise and effects on audit quality. *Journal of Accounting Research*, 48(3), 647–686.
- Reynolds, J., & Francis, J. R. (2001). Does size matter? The influence of large clients on office-level auditor reporting decisions. *Journal of Accounting and Economics*, 30(3), 375–400.
- Rosenbaum, P., & Rubin, D. (1983). The central role of the propensity score in observational studies for causal effects. *Biometrika*, 70(1), 41–55.

- Shackelford, D. A., & Shevlin, T. (2001). Empirical tax research in accounting. *Journal of Accounting & Economics*, 31(1-3), 321-387.
- Solomon, I., Shields, M., & Whittington, O. (1999). What do industry-specialist auditors know? Journal of Accounting Research, 37(1), 191–208.
- Thibodeau, J. C. (2003). The development and transferability of task knowledge. *Auditing: A Journal of Practice and Theory*, 22(1), 47–67.

Tversky, A. (1977). Features of similarity. Psychological Review, 84, 327-352.

- Van Tendeloo, B., & Vanstraelen, A. (2008). Earnings management and audit quality in Europe: Evidence from the private client segment market. *European Accounting Review*, 17(3), 447–469.
- Zerni, M. (2012). Audit partner specialization and audit fees: Some evidence from Sweden. *Contemporary Accounting Research*, 29(1), 312–340.
- Zerni, M., Haapamäki, E., Järvinen, T., & Niemi, L. (2012). Do joint audits improve audit quality? Evidence from voluntary joint audits. *European Accounting Review*, 21(4), 731–765.

Appendix

Descriptive data on the number of public-clients of audit partners

Year	Number of signing partners of listed companies	Average number of public clients per partner	≥2 Public clients (%)	≥3 Public clients (%)	≥4 Public clients (%)	≥5 Public clients (%)	≥6 Public clients (%)
2005	90	1.89	38 (42)	18 (20)	10(11)	6 (7)	3 (3)
2006	89	1.91	37 (42)	17 (19)	10(11)	7 (8)	4 (4)
2007	86	1.91	38 (44)	18 (21)	8 (9)	5 (6)	3 (3)
2008	80	1.94	35 (44)	15 (19)	10 (13)	6 (8)	5 (6)
2009	76	1.95	35 (46)	16 (21)	9 (12)	5 (7)	4 (5)
2010	74	1.95	32 (43)	16 (22)	9 (12)	6 (8)	4 (5)
Mean	82.5	1.92	36 (44)	17 (20)	9 (11)	6 (7)	4 (5)